

Contents lists available at ScienceDirect

# Journal of Functional Analysis

www.elsevier.com/locate/jfa

Full Length Article

Solution properties of the incompressible Euler system with rough path advection



Dan Crisan<sup>a</sup>, Darryl D. Holm<sup>b</sup>, James-Michael Leahy<sup>b,\*</sup>, Torstein Nilssen<sup>c</sup>

<sup>a</sup> Department of Mathematics, Imperial College London, SW7 2AZ, UK

<sup>b</sup> Department of Applied Mathematics, University of Twente, 7522 NB Enschede,

Netherlands

<sup>c</sup> Institute of Mathematics, University of Agder, 4604 Kristiansand, Norway

#### ARTICLE INFO

Article history: Received 23 May 2021 Accepted 9 July 2022 Available online 20 July 2022 Communicated by Maria Colombo

Keywords: Rough path theory Euler equations Rough PDEs A priori estimates

## ABSTRACT

The present paper aims to establish the local well-posedness of Euler's fluid equations on geometric rough paths. In particular, we consider the Euler equations for the incompressible flow of an ideal fluid whose Lagrangian transport velocity possesses an additional rough-in-time, divergence-free vector field. In recent work, we have demonstrated that this system can be derived from Clebsch and Hamilton-Pontryagin variational principles that possess a perturbative geometric rough path Lie-advection constraint. In this paper, we prove the local well-posedness of the system in  $L^2$ -Sobolev spaces  $H^m$  with integer regularity m > |d/2| + 2 and establish a Beale-Kato-Majda (BKM) blow-up criterion in terms of the  $L_t^1 L_x^\infty$ -norm of the vorticity. In dimension two, we show that the  $L^p$ -norms of the vorticity are conserved, which yields global well-posedness and a Wong-Zakai approximation theorem for the stochastic version of the equation.

© 2022 The Authors. Published by Elsevier Inc. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

\* Corresponding author.

*E-mail addresses:* d.crisan@imperial.ac.uk (D. Crisan), d.holm@imperial.ac.uk (D.D. Holm), j.leahy@imperial.ac.uk (J.-M. Leahy), torstein.nilssen@uia.no (T. Nilssen).

https://doi.org/10.1016/j.jfa.2022.109632

<sup>0022-1236/</sup>© 2022 The Authors. Published by Elsevier Inc. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

#### Contents

1.	Introd	uction	2
2.	Prelim	inaries	7
	2.1.	Basic notation, function spaces, and inequalities	7
		2.1.1. Hodge and Helmholtz decomposition and the Biot-Savart operator	9
		2.1.2. The Lie derivative	11
	2.2.	Geometric rough paths and the sewing lemma	12
3.	Main r	results	
	3.1.	Formulations of the rough incompressible Euler system	14
	3.2.	Statement of the main results	18
		3.2.1. Applications to stochastic partial differential equations	20
		3.2.2. Critical points of the Clebsch and Hamilton-Pontryagin variational principles	22
4.	A prio	ri estimates	24
	4.1.	Remainder estimates	24
	4.2.	Solution estimates	27
	4.3.	Difference estimates and uniqueness	32
5.	Local	well-posedness (Proof of Theorem 3.7)	36
6.	Proof of	of the remaining results	40
	6.1.	Recovery of the pressure and harmonic part (Proof of Proposition 3.3)	40
	6.2.	Maximally extended solution (Proof of Corollary 3.8)	
	6.3.	BKM blow-up criterion (Proof of Theorem 3.9)	42
	6.4.	Global well-posedness in two-dimensions (Proof of Theorem 3.10)	
	6.5.	Continuous dependence on data (Proof of Corollary 3.11)	
Apper	ndix A.	Unbounded rough drivers	
Apper	ndix B.	Rough Gronwall's lemma	48
Refere	ences .	•••••••••••••••••••••••••••••••••••••••	49

## 1. Introduction

Mathematical models of the interaction dynamics of disparate space and time scales in fluid flows remains one of the most active research areas, emboldened not just by theoretical considerations but also by practical applications such as the need to understand the dynamics of Earth's oceans and atmosphere in the context of global climate change. Incorporating fine-scale perturbations into the fluid motion equations and then analyzing their effects on the coarse-scale flow has become one of the primary objectives of mainstream fluid models, particularly in the last two decades. Approaches toward this objective include deterministic perturbations [66, 37, 50, 58, 67] and stochastic perturbations [57, 53, 63, 56, 59, 52, 64, 6, 54, 47, 51, 46, 3, 27, 65, 62]. Such perturbations can be exogenously introduced into the fluid model to account for (possibly unknown) external forces. They can also be introduced endogenously, for example, to model the effects of unresolved fast sub-grid scale physics or other uncertain processes. In geophysical fluid dynamics, this trend has led to many numerical developments, including the introduction of parameterization schemes used to represent model uncertainties in the interaction of disparate space and time scales to improve the probabilistic skill of the ensemble weather forecasts [64, 3, 65, 14, 15, 61, 60, 16].

Many fluid equations can be characterized as critical points of action functionals [1,42] that incorporate fluid physics via a Lagrangian whose kinetic energy is defined in terms of velocity vector fields which are right-invariant under the diffeomorphisms and whose potential energy is defined in terms of advected quantities which evolve under pushforward by the flows generated by these vector fields. Perhaps the most well-known fluid model that arises in this manner is the perfect incompressible Euler system [1], which describes geodesic flow on the manifold of diffeomorphisms endowed with the  $H^s$ -topology,  $s > \frac{d}{2} + 1$ , with respect to the weak  $L^2$ -metric defined by the fluid kinetic energy, [28].

A natural framework for introducing parameterization schemes into fluid models that arise from variational principles is through the introduction of a parameterized perturbation at the level of the Lagrangian in the action functional. Critical paths of the parameterized action functional then satisfy modified fluid motion equations that preserve the fundamental properties of the unperturbed model inherited from the variational principle. In particular, variational parameterization schemes for fluid models possess a Kelvin-Noether theorem which governs their circulation dynamics, as well as any other conservation laws arising from unbroken Lie symmetries of the original model [42]. The stochastic setting for this variational approach was introduced in [46], and many further developments of it have been made subsequently, [32,26,25,62,14]. In [18], the present authors extended [46] to obtain a class of variational principles for fluid dynamics on geometric rough paths [49,31,29]. This extension to geometric rough paths was achieved by constraining the advective transport equation to incorporate a temporally rough vector field. Critical points of the corresponding action functionals satisfy a system of rough partial differential equations (RPDEs) whose dynamics incorporate both the resolvedscale fluid velocity and the modeled effects of the unresolved fluctuations. The paper [18] provides a bridge between fluid dynamics and rough path theory. It draws upon knowledge from both areas, and we hope that it will impact both areas.

In this work, we consider Euler's equations for perfect incompressible fluid flow on geometric rough paths. It was shown in [18, Section 4.1 and 4.2] that this system is a critical path of the Clebsch or Hamilton-Pontryagin action functionals and satisfies a Kelvin circulation balance law (see, also, Section 3.2.2). On the *d*-dimensional torus  $\mathbb{T}^d$ , the rough Euler system is given by

$$\begin{cases} \mathrm{d}u + u \cdot \nabla u \, \mathrm{d}t - \pounds_{\xi_k}^* u \, \mathrm{d}\mathbf{Z}_t^k = -\mathrm{d}\nabla q_t - \mathrm{d}h_t & \text{on } (0,T] \times \mathbb{T}^d, \\ \mathrm{div} \, u = 0 & \text{on } [0,T] \times \mathbb{T}^d, \\ \int_{\mathbb{T}^d} u \, \mathrm{d}V = 0, \quad \int_{\mathbb{T}^d} q \, \mathrm{d}V = 0 & \text{on } [0,T], \\ u = u_0, \quad q = 0, \quad h = 0 & \text{on } \{0\} \times \mathbb{T}^d, \end{cases}$$

where  $\mathbf{Z} = (Z, \mathbb{Z}) \in C_g^{p-\text{var}}(\mathbb{R}_+; \mathbb{R}^K)$  is an  $\mathbb{R}^K$ -valued geometric rough path with variation  $p \in [2,3), \xi : \mathbb{T}^d \to {\mathbb{R}^d}^K$  is a collection of sufficiently smooth divergence-free vector fields, and  $\pounds_{\xi_k}^* u = -(\xi_k^j \partial_{x^j} u^i + u^j \partial_{x^i} \xi_k^j) e_i$ . Equation (3.1) is to be solved for an

unknown divergence and mean-free vector field (velocity)  $u : [0, T] \times \mathbb{T}^d \to \mathbb{R}^d$ , meanfree scalar field ('time-integrated' pressure)  $q : [0, T] \times \mathbb{T}^d \to \mathbb{R}$ , and harmonic constant (time-integrated)  $h : [0, T] \to \mathbb{R}^d$ . The pressure q and harmonic constant h should be understood as Lagrange multipliers associated with the divergence-free and mean-free constraints, respectively. The present paper aims to establish the local well-posedness of equations (3.1) in  $L^2$ -Sobolev spaces  $H^m$  with integer regularity  $m \ge |d/2| + 2$ .

The *p*-variation of the rough path provides a measure of roughness: the higher the value of p, the rougher the path. See Definition 2.1 for details. The analysis of the Euler equation perturbed by a bounded variation path (i.e., p = 1) follows the same steps as that of the (unperturbed) Euler equation (see, e.g., [5, Chapter 3] and [4, Chapter 7]). We refer to [5, Notes for Chapter 3] and [4, References and remarks for Chapter 7] for a historical account of the solution theory for the deterministic incompressible Euler equations.

For  $p \ge 2$ , the classical integration methodology is no longer applicable and we enter the realm of rough path theory (see e.g., [29]). In this paper, we treat rough path perturbations with  $p \in [2, 3)$ , in the first non-trivial regime. A similar treatment is possible for paths with variation  $p \in (1, 2)$  (i.e., the Young integration case) and  $p \ge 3$ .

The case p = 2 includes the incompressible stochastic Euler system driven by Brownian motion. See, for example, [55,36,8,17,7,48] for a non-exhaustive selection of papers covering various types of incompressible stochastic Euler systems. Having eliminated the need for stochastic integration, the RPDEs retain a pathwise interpretation. Unlike in the stochastic setting, though, no Burkholder-Davis-Gundy inequality is available to estimate the rough integral. Consequently, we will apply the method of unbounded rough drivers [9] (explained below). This method will enable us to establish a Wong-Zakai approximation for the solution. In addition, we will give an interpretation of the solution as a bona fide random dynamical system [13]. It is worth mentioning that it is possible to prove Wong-Zakai approximation results for SPDEs driven by Brownian motion without the use of the rough paths (see, e.g., [38,11]).

Our work also includes a solution theory for fractional Brownian motion driven equations, which enables memory effects to be introduced through our formulation. In our previous work [18], we explained how fluid models on geometric rough paths can be used in the context of stochastic parameterization schemes and uncertainty quantification (see, e.g., [14–16]. Our results also set the stage for investigating numerical schemes and developing geometric rough path parameterization schemes for fluid models to account for additional properties such as unknown Lagrangian trajectory roughness and system memory.

An intrinsic theory of transport-type RPDEs was developed in the papers [9,22]. In [9], the authors use a priori estimates based on Davies' type expansions [21], doubling of variables, and commutator estimates in the spirit of DiPerna-Lions to establish well-posedness and the analog of the renormalization property. In contrast, the authors of [22] use a generalized Feynman-Kac formula and forward-backward duality type arguments. The method of unbounded rough drivers was extended in [23] to RPDEs with non-

linear drift terms, which influenced a series of papers [39,43,40,41,44,34,33,45,19,30]. The most relevant to our current study are the papers [40,41,30] on the rough Navier-Stokes equations.

In [40], a rough Navier-Stokes equation with a pure linear rough transport perturbation  $\xi_k \cdot \nabla u \mathbf{Z}_t^k$  was considered, which at the formal level, conserves energy. At the present moment, the only method the authors know to obtain estimates (after passing to the limit) and the uniqueness of solutions of rough transport type PDEs in the highest spatial norm for which one has a uniform bound in time is through the method of doubling variables and commutator estimates. Unfortunately, the incompressibility constraint present in the velocity formulation makes it challenging to obtain solution estimates using this method. In [40], the existence of "weak" solutions (i.e.,  $u \in L^2([0,T]; H^1) \cap L^{\infty}([0,T]; L^2)$ was proved in 2d and 3d by establishing an energy equality along a smooth-noise approximating sequence, and thereby deriving an energy inequality for the solution by passing to the limit. In dimension two, uniqueness, continuity in  $L^2$ , and an energy equality were only proved in the special case of constant vector fields  $\xi$  because of the difficulty that arises from the incompressibility constraint. It is worth noting that for constant  $\xi$ , scalar linear transport agrees with Lie transport.

In [41], a Navier-Stokes equation with Lie-advection in the noise, as in the present paper, was considered. In contrast with [40], the vorticity formulation of the Lie-advected equation does not include projections, which facilitated the use of the doubling of variables technique for "strong" solutions (i.e.,  $u \in L^2([0,T]; H^2) \cap L^{\infty}([0,T]; H^1)$  (see Remark 3.6). In [41], the local existence of strong solutions was proved in dimension three, and global existence and uniqueness was proved for strong solutions in dimension two. Additional regularity was not investigated, and a rough version of an Aubin-Lions type compactness result was used in the proof of existence.

The present paper expands the scope of the theory of unbounded rough drivers. The main tool from this theory is Theorem 4.1, which extends the usual Davies' remainder estimates for RDEs to RPDEs with linear rough transport structure. As in [41], we make use the vorticity formulation to obtain solution estimates. In the absence of the smoothing effect of viscosity (cf., [41]), we need to obtain bounds on the higher-order derivatives of the solution. Specifically, in Section 4, we derive a system of equations for the vorticity and its derivatives up to order m with  $m \ge |d/2| + 2$ , which has a linear rough symmetric transport structure. We then derive an equation for the square of this system and apply Theorem 4.1. Next, we obtain solution estimates by applying a rough version of Grönwall's lemma (see Appendix B). The additional regularity of the solutions allows us to avoid using doubling-of-variables (used in e.g., [9,23,41]) at the expense of having to assume  $\xi$  is slightly more regular whenever continuity in time in the highest norm is needed. This simplifies many of the techniques needed for obtaining a priori estimates (see Remark 4.3). As a result of the additional regularity, we use Arzela-Ascoli to obtain compactness, rather than Aubin-Lions as in [40,41]. In Section 6.2, we show how to construct a maximally extended solution in the rough path setting, which as far as we are aware is a new result. Further extensions of this work to bounded domains

with boundary conditions is the subject of future research. However, the extension to bounded domains will encounter technical difficulties involving, for example, the method of doubling of variables, or applying the equation for the square of the vorticity when boundary conditions are present (provided additional regularity is known).

**Description of results.** The goal of the present paper is to establish solution properties of the rough incompressible Euler system given in equation (3.1). Theorem 3.7 states that for an initial condition of Sobolev regularity  $H^m$  with  $m \geq \lfloor \frac{d}{2} \rfloor + 2$  there exists a unique  $H^m$  solution of the *d*-dimensional rough Euler system on the interval  $[0, T_*]$ , where the time  $T_* > 0$  depends on the initial condition of the equation, the driving rough path, and the coefficients of the rough driver. This solution can be extended to a unique maximal solution (Corollary 3.8). Theorem 3.9 gives a blow-up criterion in terms of the  $L_t^1 L_x^{\infty}$ norm of the vorticity, which extends the well-known Beale-Kato-Majda criterion [10]. In dimension two, we show that the  $L^p$ -norms of the vorticity are conserved for all  $p \in [2, \infty]$ and thereby establish global well-posedness (see Theorem 3.10). Corollary 3.11 states that the solution, the pressure term, and harmonic constant are jointly continuous as functions of the initial condition, the driving rough path, and the coefficients of the rough driver. In Section 3.2.1, we discuss applications of our results to stochastic partial differential equations (Sades), including Theorem 3.13, which gives a Wong-Zakai approximation result for the corresponding Stratonovich driven SPDE. In Section 3.2.2, we explain how our main results yield critical points of the Clebsch and Hamilton-Pontryagin action functionals introduced in [18]. To do this, we derive a generalized Weber representation formula (3.19) of smooth (in space) solutions.

#### Structure of the paper:

- Section 2 introduces the basic notation and summarizes the standard results that will be required throughout the paper.
- In Section 3, we formulate the rough incompressible Euler equation in velocity form and as well as vorticity form and state the main results of the paper: local wellposedness in any dimension, the Beale-Kato-Majda blow-up criteria, global wellposedness in 2d, and continuous dependence on data. We discuss various applications to stochastic equations, including a Wong-Zakai approximation result. Finally, we explain the correspondence of solutions with critical points of the action functionals presented in [18].
- Section 4 contains a priori estimates for remainders, the solution, and differences of solutions.
- Section 5 contains the proof of local well-posedness.
- Section 6 contains the proof of the remaining results.
- The appendices A and B contain technical results that are used in establishing the a priori estimates.

## Acknowledgments

All of the authors are grateful to our friends and colleagues who have generously offered their time, thoughts, and encouragement in the course of this work during the time of COVID-19. We are particularly grateful to T. D. Drivas for thoughtful discussions, especially an interesting discussion regarding Section 3.2.2. We also thank the anonymous referee for the helpful comments and constructive remarks on this manuscript. DC and DH are grateful for partial support from ERC Synergy Grant 856408 - STUOD (Stochastic Transport in Upper Ocean Dynamics). JML is grateful for support from US AFOSR Grant FA9550-19-1-7043 - FDGRP (Fluid Dynamics of Geometric Rough Paths) and AFOSR Grant FA8655-21-1-7034 - RPFD (Rough Path Fluid Dynamics). TN is grateful for partial support from the DFG via the Research Unit FOR 2402.

# 2. Preliminaries

## 2.1. Basic notation, function spaces, and inequalities

Let  $d \in \{2, 3, ...\}$  and  $\mathbb{T}^d = \mathbb{R}^d / (2\pi\mathbb{Z})^d \cong (S^1)^d$  denote the *d*-dimensional flat torus. The Riemannian covering map  $\pi : \mathbb{R}^d \to \mathbb{T}^d$  induces global orthonormal frames  $\{\partial_{x^i}\}_{i=1}^d$ and  $\{dx^i\}_{i=1}^d$  of the tangent  $T\mathbb{T}^d \cong \mathbb{T}^d \times \mathbb{R}^d$  and cotangent bundle  $T^*\mathbb{T}^d \cong \mathbb{T}^d \times (\mathbb{R}^d)^*$ , respectively, and a normalized Haar measure  $dV = \operatorname{Vol}(\mathbb{T}^d)^{-1}dx^1 \wedge \cdots \wedge dx^d$ .

Let V denote an arbitrarily given finite dimensional real vector space with inner product  $(\cdot, \cdot)_V$  and norm  $|\cdot|_V$ . Denote by  $C^{\infty}(\mathbb{T}^d; V)$  the Fréchet space of smooth Vvalued functions on  $\mathbb{T}^d$ . For given  $m \in \mathbb{N}$ , let  $C^m(\mathbb{T}^d; V)$  denote the Banach space of *m*times continuously differentiable V-valued functions on  $\mathbb{T}^d$ . We will blur the distinction between  $2\pi$ -periodic functions and functions on  $\mathbb{T}^d$ . Let  $\mathcal{L}(\mathbb{R}^d; V)$  denote the space of linear maps from vector space  $\mathbb{R}^d$  to V and  $D: C^{\infty}(\mathbb{T}^d; V) \to C^{\infty}(\mathbb{T}^d; \mathcal{L}(\mathbb{R}^d; V))$  denote the derivative operator. Let  $\Delta: C^{\infty}(\mathbb{T}^d; V) \to C^{\infty}(\mathbb{T}^d; V)$  denote the Laplacian, which is defined by  $\Delta f = \partial_{x^i}^2 f$ . Here and below we use the convention of summing repeated indices over their range of values.

For given  $p \in [1, \infty]$ , denote by  $L^p(\mathbb{T}^d; V) = L^p((0, 2\pi)^d; V)$  the Banach space of equivalence classes of V-valued of  $L^p$ -integrable functions on  $\mathbb{T}^d$  with norm  $|\cdot|_{L^p}$ . We denote by  $(\cdot, \cdot)_{L^2}$  the inner product on the Hilbert space  $L^2(\mathbb{T}^d; V)$ . Since it will always be clear from the context where a function takes its values, we drop the dependence of norms and inner products on V. It is well-known that the sequence  $\{\psi_n\}_{n\in\mathbb{Z}^d} \subset C^\infty(\mathbb{T}^d;\mathbb{C})$ defined by for all  $n \in \mathbb{Z}^d$  and  $x \in \mathbb{T}^d$  by  $\psi_n(x) = e^{in \cdot x}$  forms an orthonormal basis of  $L^2(\mathbb{T}^d;\mathbb{C})$ .

Denote the Fourier transform  $\mathcal{F} : L^1(\mathbb{T}^d; V) \to \ell^{\infty}(\mathbb{Z}^d; V)$  by  $\mathcal{F}f(n) = \hat{f}(n) = \int_{\mathbb{T}^d} f\psi_n dV$  and its inverse  $\mathcal{F}^{-1} : l^{\infty}(\mathbb{Z}^d; V) \to L^1(\mathbb{T}^d; V)$  by  $\mathcal{F}^* \hat{f} = \sum_{n \in \mathbb{Z}^d} \hat{f}(n)\psi_n$ . Let  $S(\mathbb{Z}^d; V) = \{\{c_n\}_{n \in \mathbb{Z}^d} \subset V : \sup_{n \in \mathbb{Z}^d} (1 + |n|)^N |c_n| < \infty, \forall N \in \mathbb{N}\}$  denote the Fréchet space of rapidly decreasing multi-sequences. It follows that  $\mathcal{F} : C^{\infty}(\mathbb{T}^d; V) \to$   $S(\mathbb{Z}^d; V), \mathcal{F}^{-1} : S(\mathbb{Z}^d; V) \to C^{\infty}(\mathbb{T}^d; V)$ , and moreover that  $\mathcal{F}$  extends to an isometric isomorphism  $\mathcal{F} : L^2(\mathbb{T}^d; V) \to \ell^2(\mathbb{Z}^d; V)$ .

Denote by  $\mathcal{D}'(\mathbb{T}^d; V) = \{f : C^{\infty}(\mathbb{T}^d; \mathbb{C}) \to V; f \text{islinearandcontinuous}\}$  the Fréchet space of V-valued distributions. Equivalently,  $\mathcal{D}'(\mathbb{T}^d; V)$  may be characterized as  $2\pi$ -periodic distributions  $\mathcal{D}'(\mathbb{R}^d; V)$  or as distributional Fourier series. Indeed, the Fourier transform extends via duality to an isomorphism  $\mathcal{F} : \mathcal{D}'(\mathbb{T}^d; V) \to S'(\mathbb{Z}^d; V)$ , where  $S'(\mathbb{Z}^d; V) = \{\{c_n\}_{n \in \mathbb{Z}^d} \subset V : \exists N \in \mathbb{N} \text{s.t.} \sup_{n \in \mathbb{Z}^d} (1 + |n|)^{-N} |c_n| < \infty\}$ denotes the space of slowly increasing multi-sequences. Clearly, for all  $p \in [1, \infty)$ ,  $C^{\infty}(\mathbb{T}^d; V), L^p(\mathbb{T}^d; V) \subset \mathcal{D}'(\mathbb{T}^d; V)$ . The differential operators  $\partial^{\alpha}$ , D, and  $\Delta$  extend to distributions via duality.

Denote by  $\mathring{C}^{\infty}(\mathbb{T}^d; V)$ ,  $\mathring{L}^p(\mathbb{T}^d; V)$ , and  $\mathring{\mathcal{D}}'(\mathbb{T}^d; V)$  the corresponding subspace of distributions f with  $\widehat{f}(0) = 0$ . If  $f \in \mathcal{D}'(\mathbb{T}^d; V)$  is such that  $\Delta f = -\sum_{n \in \mathbb{Z}^d} |n|^2 \widehat{f}(n) \psi_n = 0_V$  (i.e., harmonic), then  $\widehat{f}(n) = 0$  for all  $n \in \mathbb{Z}^d - \{0\}$ , which implies  $\operatorname{Ker} \Delta = V$ .

For given  $s \in \mathbb{R}$ , define  $(I - \Delta)^{-\frac{s}{2}} : \mathcal{D}'(\mathbb{T}^d; V) \to \mathcal{D}'(\mathbb{T}^d; V)$  by

$$(I - \Delta)^{-\frac{s}{2}} f = \sum_{n \in \mathbb{Z}^d} \langle n \rangle^{-s} \hat{f}(n) \psi_n \,,$$

where  $\langle n \rangle = (1 + |n|^2)^{\frac{1}{2}}$ . Let  $s \in \mathbb{R}$  and  $H^s(\mathbb{T}^d; V) = (I - \Delta)^{-\frac{s}{2}} L^2(\mathbb{T}^d; V)$  denote the Bessel potential spaces, which are Hilbert spaces with inner products and norms given by

$$(f,g)_{H^s} := \left(\sum_{n \in \mathbb{Z}^d} \langle n \rangle^s (\hat{f}(n), \hat{g}(n))_V \right)^{\frac{1}{2}} \quad \text{and} \quad |f|_{H^{s,p}} := \left|\sum_{n \in \mathbb{Z}^d} \langle n \rangle^s \hat{f}(n) \psi_n \right|_{L^p}.$$

The duality pairing  $\langle \cdot, \cdot \rangle_{H^s} : H^{-s}(\mathbb{T}^d; V) \times H^s(\mathbb{T}^d; V) \to \mathbb{R}$  given by

$$\langle f, g \rangle_{H^s} = ((I - \Delta)^{-\frac{s}{2}} f, (I - \Delta)^{\frac{s}{2}} g)_{L^2}$$

induces an isomorphism  $H^{-s}(\mathbb{T}^d; V) \cong H^s(\mathbb{T}^d; V)'$ . It follows that for all  $m \in \mathbb{N}_0$ ,  $H^m(\mathbb{T}^d; V) = \{f \in L^2(\mathbb{T}^d; V) : \sum_{0 \le n \le m} |D^n f|_{L^2}^2 < \infty\}$ . For  $m \in \mathbb{N}$ , denote by  $W^{m,\infty}(\mathbb{T}^d; V)$  the Banach space of functions  $f \in L^\infty(\mathbb{T}^d; V)$  such that  $|f|_{W^{m,\infty}} := \max_{0 \le |\alpha| \le m} |\partial^{\alpha} f|_{L^{\infty}} < \infty$ .

For given  $s \in \mathbb{R}$ , we let  $\mathring{H}^s(\mathbb{T}^d; V) = \{f \in H^s(\mathbb{T}^d; V) : \widehat{f}(0) = 0_V\}$  and for  $m \in \mathbb{N}_0$ , let  $\mathring{W}^{m,\infty}(\mathbb{T}^d; V) = \{f \in \mathring{W}^{m,\infty}(\mathbb{T}^d; V) : \widehat{f}(0) = \int_{\mathbb{T}^d} f dV = 0_V\}.$ 

Henceforth, we use the notation  $\leq_{c_1,\ldots,c_d}$  to denote less than equal to up to a constant  $C = C(c_1,\ldots,c_d)$  depending only on parameters  $c_1,\ldots,c_d$ . Throughout the paper, we will make regular use of the following facts:

• For given  $N \in \mathbb{N}$ , define  $P_{\leq N} : \mathcal{D}'(\mathbb{T}^d; V) \to C^{\infty}(\mathbb{T}^d; V)$  by  $P_{\leq N}f = \sum_{|n|\leq N} \hat{f}(n)\psi_n$ . If  $s_0 < s_1$ , then for all  $f \in H^{s_0}(\mathbb{T}^d; V)$ ,  $|P_{\leq N}f|_{H^{s_1}} \leq N^{s_1-s_0}|f|_{H^{s_0}}$  and for all  $f \in H^{s_1}(\mathbb{T}^d; V)$ ,  $|(I - P_{\leq N})f|_{H^{s_0}} \leq N^{s_0-s_1}|f|_{H^{s_1}}$ .

- If s > d/2 + m for some m ∈ N<sub>0</sub>, then C<sup>m</sup>(T<sup>d</sup>; V) ⊂ H<sup>s</sup>(T<sup>d</sup>; V).
  (Poincaré) For all f ∈ H<sup>s+1</sup>(T<sup>d</sup>; V), |f|<sub>H<sup>s</sup></sub> ≤<sub>d,s</sub> |Df|<sub>H<sup>s</sup></sub>. For all f ∈ W<sup>m+1,∞</sup>(T<sup>d</sup>; V),  $|f|_{W^{m,\infty}} \lesssim_{d,m} |Df|_{W^{m,\infty}}.$
- For all  $m_1, m_2 \in \mathbb{N}_0$  and  $f, g \in L^{\infty}(\mathbb{T}^d; V) \cap H^{m_1+m_2}(\mathbb{T}; V)$ ,

$$||D^{m_1}f||D^{m_2}g||_{L^2} \lesssim_{d,m_1,m_2} |f|_{L^{\infty}} |g|_{H^{m_1+m_2}} + |f|_{H^{m_1+m_2}} |g|_{L^{\infty}}.$$
 (2.1)

• For all  $m \in \mathbb{N}_0$ ,  $f \in L^{\infty}(\mathbb{T}^d; V) \cap H^{m+1}(\mathbb{T}; V)$  and  $g \in L^{\infty}(\mathbb{T}^d; V) \cap H^m(\mathbb{T}; V)$ 

$$\sum_{0 \le |\alpha| \le m} |\partial^{\alpha}(f\nabla g) - f\partial^{\alpha}\nabla g|_{L^2} \lesssim_{d,m} \left( |\nabla f|_{L^{\infty}} |g|_{H^m} + |f|_{H^{m+1}} |g|_{L^{\infty}} \right).$$
(2.2)

Let  $(E, |\cdot|_E)$  be an arbitrarily given Banach space. For an interval  $I \subset \mathbb{R}_+$ , denote by C(I; E) (resp.  $C_w(I; E)$ ) the space of continuous (resp. weakly-continuous) E-valued functions. For  $p \in [1, \infty]$ , let  $L^p(I; E)$  the Banach space of equivalence classes of E-valued  $L^{p}$ -integrable strongly measurable *E*-valued functions on *I*.

# 2.1.1. Hodge and Helmholtz decomposition and the Biot-Savart operator

Let  $\{e_i\}_{i=1}^d$  denote the standard basis of  $\mathbb{R}^d$  and let  $\{e^i\}_{i=1}^d$  denote the dual basis. We identify vector fields  $u = u^i \partial_{x^i} \in \Gamma(\mathbb{T}^d; T\mathbb{T}^d)$  with  $\mathbb{R}^d$ -valued maps  $u = u^i e_i : \mathbb{T}^d \to \mathbb{R}^d$ and k-forms  $\alpha = \sum_{i_1 < \cdots < i_d} \alpha_{i_1 \dots i_k} dx^{i_1} \wedge \cdots \wedge dx^{i_k} \in \Gamma(\mathbb{T}^d; \Lambda^k T^* \mathbb{T}^d)$  with  $\Lambda^k(\mathbb{R}^d)^*$ valued maps  $\alpha = \sum_{i_1 < \cdots < i_k}^{i_1 < \cdots < i_d} \alpha_{i_1 \dots i_k} e^{i_1} \wedge \cdots \wedge e^{i_k} : \mathbb{T}^d \to \Lambda^k(\mathbb{R}^d)^*$ . Denote by  $\flat$  the map that sends vector fields to one-forms given by  $u^{b} = u_{i}e^{i}$ , where  $u_{i} = u^{i}$  since the metric is locally Euclidean. Let  $\sharp$  denote the inverse of  $\flat$ .

For a given diffeomorphism  $\phi : \mathbb{T}^d \to \mathbb{T}^d$  and  $\alpha \in \Gamma(\mathbb{T}^d; \Lambda^k T^* \mathbb{T}^d)$ , denote the pullback and pushforward by

$$\phi^* \alpha = k! \sum_{\substack{i_1 < \ldots < i_k \\ j_1 < \cdots < j_k}} \alpha_{j_1 \ldots j_k} \circ \phi \partial_{x^{i_1}} \phi^{j_1} \cdots \partial_{x^{i_k}} \phi^{j_k} e^{i_1} \wedge \cdots \wedge e^{i_k}$$

and  $\phi_* \alpha = \phi^{-1;*} \alpha$ , respectively.

Denote by **d**,  $\delta$ , and  $-\Delta = \mathbf{d}\delta + \delta\mathbf{d}$  the exterior derivative, co-differential (i.e., adjoint of  $\mathbf{d}$ ), and Hodge Laplacian operators, respectively. In particular, using the above identification, for  $f \in \mathcal{D}(\mathbb{T}^d; \mathbb{R}^d)$ ,  $\alpha \in \mathcal{D}'(\mathbb{T}^d; (\mathbb{R}^d)^*)$ ,  $\omega \in \mathcal{D}'(\mathbb{T}^d; \Lambda^2(\mathbb{R}^d)^*)$ , and  $\gamma \in \mathcal{D}'(\mathbb{T}^d; \Lambda^3(\mathbb{R}^d)^*),$ 

$$\begin{split} \mathbf{d}f &= \partial_{x^i} f e^i, \quad \mathbf{d}\alpha = \sum_{i < j} (\partial_{x^i} \alpha_j - \partial_{x^j} \alpha_i) e^i \wedge e^j, \\ \mathbf{d}\omega &= \sum_{i < j < k} (\partial_{x^i} \omega_{jk} - \partial_{x^j} \omega_{ik} + \partial_{x^k} \omega_{ij}) e^i \wedge e^j \wedge e^k, \\ \delta f &= 0, \quad \delta \alpha = -\partial_{x^i} \alpha_i, \quad \delta \omega = \partial_{x^j} \omega_{ij} e^i, \quad \delta \gamma = -\sum_{i < j} \partial_{x^k} \gamma_{ijl} e^i \wedge e^j, \end{split}$$

$$-\Delta f = -\partial_{x^j}^2 f, \quad -\Delta \alpha = -\partial_{x^j}^2 \alpha_i e^i, \quad -\Delta \omega = -\partial_{x^k}^2 \omega_{ij} e^i \wedge e^j.$$

We recall that  $\mathbf{d}\phi_* = \phi_* \mathbf{d}$ ,  $\mathbf{d}^2 = 0$ , and  $\delta^2 = 0$ .

Let  $s \in \mathbb{R}$  be arbitrarily given. For arbitrarily given  $\alpha \in H^s(\mathbb{T}^d; \Lambda^k(\mathbb{R}^d)^*)$ ,  $k \in \{0, 1, 2\}$ , there exists a unique  $\beta \in \mathring{H}^{s+2}(\mathbb{T}^d; \Lambda^k(\mathbb{R}^d)^*)$  such that  $-\Delta\beta = \alpha - \hat{\alpha}(0)$  given by  $\beta = (-\Delta)^{-1}(\alpha - \hat{\alpha}(0)) = \sum_{n \in \mathbb{Z}^d} |n|^{-2} \hat{\alpha}(n) \psi_n$ , which yields the Hodge decomposition

$$\alpha = \mathbf{d}\delta\beta + \delta\mathbf{d}\beta + \hat{\alpha}(0) = \mathbf{d}\delta(-\Delta)^{-1}(\alpha - \hat{\alpha}(0)) + \delta\mathbf{d}(-\Delta)^{-1}(\alpha - \hat{\alpha}(0)) + \hat{\alpha}(0)$$

Let

$$\mathring{H}^{s}_{\mathbf{d}}(\mathbb{T}^{d};\Lambda^{k}(\mathbb{R}^{d})^{*}) = \mathbf{d}\mathring{H}^{s+1}(\mathbb{T}^{d};\Lambda^{k-1}(\mathbb{R}^{d})^{*}) = \{\alpha \in \mathring{H}^{s}(\mathbb{T}^{d};\Lambda^{k}(\mathbb{R}^{d})^{*}) : \mathbf{d}\alpha = 0\},\$$

which is understood to be  $\emptyset$  if k = 0,

$$\mathring{H}^{s}_{\delta}(\mathbb{T}^{d};\Lambda^{k}(\mathbb{R}^{d})^{*}) = \delta\mathring{H}^{s+1}(\mathbb{T}^{d};\Lambda^{k+1}(\mathbb{R}^{d})^{*}) = \{\alpha \in \mathring{H}^{s}(\mathbb{T}^{d};\Lambda^{k}(\mathbb{R}^{d})^{*}) : \delta\alpha = 0\},$$

and  $\mathcal{H}^k = \text{Ker}\,\Delta = \Lambda^k(\mathbb{R}^d)^*$ . Thus, we obtain the following orthogonal decomposition for  $k \in \{0, 1, 2\}$ :

$$H^{s}(\mathbb{T}^{d};\Lambda^{k}(\mathbb{R}^{d})^{*}) = \mathring{H}^{s}_{\mathbf{d}}(\mathbb{T}^{d};\Lambda^{k}(\mathbb{R}^{d})^{*}) \oplus \mathring{H}^{s}_{\delta}(\mathbb{T}^{d};\Lambda^{k}(\mathbb{R}^{d})^{*}) \oplus \mathcal{H}^{k}.$$
(2.3)

For given  $u \in \mathcal{D}'(\mathbb{T}^d; \mathbb{R}^d)$ , denote div  $u = \partial_{x^i} u^i = \partial_{x^i} u_i = -\delta u^{\flat}$ . Moreover, let

$$\overset{}{H}^{s}_{\sigma}(\mathbb{T}^{d};\mathbb{R}^{d}) = \{ u \in \overset{}{H}^{s}(\mathbb{T}^{d};\mathbb{R}^{d}) : \operatorname{div} u = 0 \},$$

$$W^{m,\infty}_{\sigma}(\mathbb{T}^{d};\mathbb{R}^{d}) = \{ u \in W^{m,\infty}(\mathbb{T}^{d};\mathbb{R}^{d}) : \operatorname{div} u = 0 \}$$

$$\overset{}{C}^{\infty}_{\sigma}(\mathbb{T}^{d};\mathbb{R}^{d}) = \{ u \in C^{\infty}(\mathbb{T}^{d};\mathbb{R}^{d}) : \operatorname{div} u = 0 \},$$

$$\overset{}{D}'_{\sigma}(\mathbb{T}^{d};\mathbb{R}^{d}) = \{ u \in \mathcal{D}'(\mathbb{T}^{d};\mathbb{R}^{d}) : \operatorname{div} u = 0 \}.$$

Henceforth, we will simply write  $\mathring{H}^{s}_{\sigma}, W^{m,\infty}_{\sigma}, \mathring{C}^{\infty}_{\sigma}$ , and  $\mathring{D}'_{\sigma}$ . Applying the sharp operator  $\sharp$  to (2.3) in the case k = 1, we derive the *d*-dimensional Helmholtz decomposition of vector fields:

$$H^{s}(\mathbb{T}^{d};\mathbb{R}^{d}) = \nabla \mathring{H}^{s+1}(\mathbb{T}^{d};\mathbb{R}) \oplus \mathring{H}^{s}_{\sigma}(\mathbb{T}^{d};\mathbb{R}^{d}) \oplus \mathbb{R}^{d}.$$

Let  $Q \in \mathcal{L}(H^s(\mathbb{T}^d; \mathbb{R}^d); \nabla \mathring{H}^{s+1}(\mathbb{T}^d; \mathbb{R})), P \in \mathcal{L}(H^s(\mathbb{T}^d; \mathbb{R}^d); \nabla \mathring{H}^s_{\sigma})$ , and  $H \in \mathcal{L}(H^s(\mathbb{T}^d; \mathbb{R}^d); \mathbb{R}^d)$  denote the projections associated with the decomposition so that I = P + Q + H. Let

$$BS = \sharp(-\Delta)^{-1}\delta = \sharp\delta(-\Delta)^{-1}$$

10

denote the inverse of  $\mathbf{d}\flat$ :  $H^{s+1}_{\sigma}(\mathbb{T}^d; \mathbb{R}^d) \to H^s_{\mathbf{d}}(\mathbb{T}^d; \Lambda^2(\mathbb{R}^d)^*)$ . The operator BS is called the Biot-Savart operator. Owing to Propositions 7.5 and 7.7 of [4], for all  $\omega \in \mathring{H}^{s-1}(\mathbb{T}^d; \Lambda^2(\mathbb{R}^d)^*)$  such that  $s > \frac{d}{2} + 1$ , we have

$$|\operatorname{BS}\omega|_{H^s} \lesssim_d |\nabla\operatorname{BS}\omega|_{H^{s-1}} \lesssim_{d,s} |\omega|_{H^{s-1}} \lesssim_{d,s} |\operatorname{BS}\omega|_{H^s},$$
(2.4)

$$|\nabla \operatorname{BS} \omega|_{L^{\infty}} \lesssim_{d,s} \ln\left(e + |\omega|_{H^{s-1}}\right) |\omega|_{L^{\infty}}, \ \forall \omega \in \check{H}^{s-1}(\mathbb{T}^d; \Lambda^2(\mathbb{R}^d)^*).$$
(2.5)

## 2.1.2. The Lie derivative

Let  $s > \frac{d}{2} + 1$  so that  $|\nabla f|_{L^{\infty}} < \infty$  for all  $f \in H^{s}(\mathbb{T}^{d}; V)$ . For  $v \in H^{s}(\mathbb{T}^{d}; \mathbb{R}^{d})$  and  $k \in \{0, 1, 2\}$ , the Lie derivative  $\pounds_{v} \in \mathcal{L}(H^{s}(\mathbb{T}^{d}; \Lambda^{k}(\mathbb{R}^{d})^{*}); H^{s-1}(\mathbb{T}^{d}; \Lambda^{k}(\mathbb{R}^{d})^{*})$  acts on  $f \in H^{s}(\mathbb{T}^{d}; \mathbb{R}), \ \alpha \in H^{s}(\mathbb{T}^{d}; (\mathbb{R}^{d})^{*}), \ \omega \in H^{s}(\mathbb{T}^{d}; \Lambda^{2}(\mathbb{R}^{d})^{*})$  by

$$\pounds_v f = v \cdot \nabla f = v^j \partial_{x^j} f, \qquad \pounds_v \alpha = (v^j \partial_{x^j} \alpha_i + \alpha_j \partial_{x^i} v^j) e^i,$$
$$\pounds_v \omega = \sum_{i < j} (v^q \partial_{x^q} \omega_{ij} + \omega_{qj} \partial_{x^i} v^q + \omega_{iq} \partial_{x^j} v^q) e^i \wedge e^j.$$

Indeed, it is a direct consequence of (2.1) that the range of the Lie derivative lies in  $H^{s-1}$ . A simple computation shows that  $\mathbf{d} \pounds_v = \pounds_v \mathbf{d}$ . Moreover, if  $\mathbf{d} \omega = 0$ , then

$$\pounds_{v}\omega = \sum_{i < j} \left( \partial_{x^{i}}(v^{q}\omega_{qj}) - \partial_{x^{j}}(v^{q}\omega_{qi}) \right) e^{i} \wedge e^{j} = \mathbf{d}[\omega(v, \cdot)],$$

which is just a special case of Cartan's formula.

The Lie derivative and covariant derivative  $\pounds_v, \nabla_v = v \cdot \nabla \in \mathcal{L}(H^s(\mathbb{T}^d; \Lambda^k(\mathbb{R}^d)^*); H^{s-1}(\mathbb{T}^d; \mathbb{R}^d)$  act on  $u \in H^s(\mathbb{T}^d; \mathbb{R}^d)$  by

$$\pounds_v u = (v^j \partial_{x^j} u^i - u^j \partial_{x^j} v^i) e_i, \quad \nabla_v u = v \cdot \nabla u = v^j \partial_{x^j} u^i e_i.$$

Stoke's theorem implies that for all  $w, u, v \in H^s(\mathbb{T}^d; \mathbb{R}^d)$ ,

$$(\pounds_v w, u)_{L^2} = -\int_{\mathbb{T}^d} w^i (v^j \partial_{x^j} u^i + u^i \operatorname{div} v + u^j \partial_{x^i} v^j) dV =: (w, \pounds_v^* u)_{L^2}.$$

In particular, if  $\operatorname{div} v = 0$ , then

$$\pounds_v^* u = -(v^j \partial_{x^j} u^i + u^j \partial_{x^i} v^j) e_i$$

and

$$-\pounds_v^* u = \sharp \pounds_v u^\flat \quad \Rightarrow \quad -\mathbf{d} \flat \pounds_u^* w = \mathbf{d} \pounds_v u^\flat = -\pounds_v \mathbf{d} u^\flat, \tag{2.6}$$

$$P\pounds_v^* u = P\pounds_v^* P u. \tag{2.7}$$

Moreover,

$$u \cdot \nabla u = -\pounds_u^* u + 2^{-1} \nabla |u|^2. \tag{2.8}$$

If d = 2, then the Hodge Star (see, e.g., Section 6.2 of [2]) map  $\star : H^s(\mathbb{T}^d; \Lambda^2(\mathbb{R}^2)^*) \to H^s(\mathbb{T}^2; \mathbb{R})$  defined by  $\tilde{\omega} = \omega_{12}$  is an isomorphism and for all  $v \in H^s_{\sigma}(\mathbb{T}^d; \mathbb{R}^d)$ , we have

$$\star \pounds_v \omega = \pounds_v \star \omega = v \cdot \nabla \omega. \tag{2.9}$$

Moreover, if d = 3, then the Hodge Star map  $\sharp \star H^s(\mathbb{T}^d; \Lambda^2(\mathbb{R}^3)^*) \to H^s(\mathbb{T}^3; \mathbb{R}^3)$  given by  $\sharp \star \omega = \omega_{23}e_1 + \omega_{31}e_2 + \omega_{12}e_3$  is an isomorphism and for all  $v \in H^s_{\sigma}(\mathbb{T}^d; \mathbb{R}^d)$ , we have

$$\sharp \star \pounds_v \omega = \pounds_v \sharp \star \omega = [v, \sharp \star \omega]. \tag{2.10}$$

## 2.2. Geometric rough paths and the sewing lemma

For an arbitrarily given closed interval I = [a, b], denote

$$\Delta_I := \{(s,t) \in I^2 : s \le t\} \quad \text{and} \quad \Delta_I^2 := \{(s,\theta,t) \in I^3 : s \le \theta \le t\}$$

If the interval I = [0, T], we write  $\Delta_T = \Delta_I$ .

Let  $(E, |\cdot|_E)$  be an arbitrarily given Banach space with norm  $|\cdot|_E$ . We say a twoparameter function  $g: \Delta_I \to E$  has finite *p*-variation for some  $p \in (0, \infty)$  on *I* if

$$|g|_{p-\operatorname{var};I;E} := \sup_{\mathfrak{p}=(t_i)\in\mathcal{P}(I)} \left(\sum_{i=1}^{\#\mathfrak{p}-1} |g_{t_it_{i+1}}|_E^p\right)^{\frac{1}{p}} < \infty,$$

where  $\mathcal{P}(I)$  is the set of all finite partitions of I and  $\#\mathfrak{p}$  denotes the number of points in a given partition  $\mathfrak{p} \in \mathcal{P}(I)$ . Denote by  $C_2^{p-\operatorname{var}}(I; E)$  the set of all continuous functions with finite p-variation on I equipped with the seminorm  $|\cdot|_{p-\operatorname{var};I;E}$ . Denote by  $C^{p-\operatorname{var}}(I; E)$  the set of all paths  $z: I \to E$  such that  $\delta z \in C_2^{p-\operatorname{var}}(I; E)$ , where

$$\delta z_{st} := z_t - z_s, \quad (s, t) \in \Delta_I.$$

For an arbitrary interval  $I \subset \mathbb{R}$  that is not necessarily closed, denote by  $C_{2,\text{loc}}^{p-\text{var}}(I; E)$ the set of all continuous functions  $g: \Delta_I \to E$  such that there exists a countable sequence of closed intervals  $\{I_k\}$  such that  $\cup_k I_k = I$  and  $g \in C_2^{p-\text{var}}(I_k; E)$ .

A continuous mapping  $\varpi : \Delta_I \to [0, \infty)$  is called a control on I if  $\omega(s, s) = 0$  for all  $s \in I$  and if for all  $(s, \theta, t) \in \Delta_I^2$ ,

$$\varpi(s,\theta) + \varpi(\theta,t) \le \varpi(s,t),$$

which is referred to as superadditivity. If  $\varpi_1$  and  $\varpi_2$  are controls, then for all  $\alpha, \beta \in \mathbb{R}_+$  such that  $\alpha + \beta \geq 1$ ,  $\varpi_1^{\alpha} \varpi_2^{\beta}$  is a control (see, e.g., [31, Ex. 1.9]).

If  $g \in C_2^{p-\text{var}}(I; E)$  for a given  $p \in (0, \infty)$ , then  $\varpi_g : \Delta_I \to [0, \infty)$  defined for all  $(s, t) \in \Delta_I$  by

$$\varpi_g(s,t) = |g|_{p-\operatorname{var};[s,t];E}^p$$

is a control (see, e.g., [31, Prop. 5.8]). Moreover, it is straightforward to check that for all  $(s,t) \in \Delta_I$ 

$$\inf\{\varpi(s,t)^{\frac{1}{p}}: \varpi \text{isacontrols.t.} | g_{r\theta}|_E \le \varpi(r,\theta)^{\frac{1}{p}}, \ \forall (r,\theta) \in \Delta_{[s,t]}\},\$$

is an equivalent semi-norm on  $C_2^{p-\text{var}}(I; E)$ .

For an arbitrarily given two-index map  $g: \Delta_I \to \mathbb{R}$ , define the increment operator

$$\delta g_{s\theta t} = g_{st} - g_{\theta t} - g_{s\theta}, \quad (s, \theta, t) \in \Delta_I^2$$

We will now give the definition of a rough path and geometric rough path. We refer the reader to [49,31,29] for more thorough expositions.

**Definition 2.1.** Let  $K \in \mathbb{N}$  and  $p \in [2, 3)$ . A *p*-variation rough path is a pair

$$\mathbf{Z} = (Z, \mathbb{Z}) \in \mathcal{C}^{p-\operatorname{var}}(I; \mathbb{R}^K) := C_2^{p-\operatorname{var}}(I; \mathbb{R}^K) \times C_2^{\frac{p}{2}-\operatorname{var}}(I; \mathbb{R}^{K \times K})$$

that satisfies the Chen relations

$$\delta Z_{s\theta t} = 0 \quad \text{and} \quad \delta \mathbb{Z}_{s\theta t} = Z_{s\theta} \otimes Z_{\theta t}, \quad \forall (s, \theta, t) \in \Delta_I^2.$$
 (2.11)

For an arbitrarily given  $\mathbf{Z} \in \mathcal{C}^{p-\operatorname{var}}([0,T];\mathbb{R}^K)$ , denote

 $\varpi_{\mathbf{Z}} = \inf\{\varpi : \varpi \text{isacontrols.t.} | Z_{st} | \le \varpi(s, t) \text{ and } | \mathbb{Z}_{st} | \le \varpi(s, t), \ \forall (s, t) \in \Delta_I \}.$ (2.12)

Given a smooth path  $z : I \to \mathbb{R}^K$ , we define its canonical lift  $\mathbf{Z} = (Z, \mathbb{Z}) \in \mathcal{C}^{p-\operatorname{var}}([0,T];\mathbb{R}^K)$  by

$$Z_{st} = \delta z_{st}$$
 and  $\mathbb{Z}_{st} := \int_{s}^{t} Z_{sr} \otimes dz_{r}, (s,t) \in I.$ 

An element  $\mathbf{Z} = (Z, \mathbb{Z}) \in C^{p-\operatorname{var}}(I; \mathbb{R}^K)$  is said to be geometric if it can be obtained as the limit in the product topology of a sequence of rough paths  $\{(Z^n, \mathbb{Z}^n)\}_{n=1}^{\infty}$  that are canonical lifts of smooth paths  $z^n : I \to \mathbb{R}^K$ . We denote by  $C_g^{p-\operatorname{var}}(I; \mathbb{R}^K)$  the set of geometric *p*-variation rough paths and endow it with the product topology. Finally, we denote by  $C_g^{p-\operatorname{var}}(\mathbb{R}_+; \mathbb{R}^K)$  the corresponding Fréchet space of pairs  $\mathbf{Z} = (Z, \mathbb{Z}) :$  $\Delta_{[0,\infty)} \to \mathbb{R}^K \times \mathbb{R}^{K \times K}$  belonging to  $C_g^{p-\operatorname{var}}([0,T]; \mathbb{R}^K)$  for all positive T. The following lemma, referred to as the *sewing lemma*, lies at the very foundation of the theory of rough paths. The proof is a straightforward modification of [23, Lemma 1.2] or [29, Lemma 4.2].

**Lemma 2.2** (Sewing lemma). Let  $\varpi_1$  and  $\varpi_2$  be controls on I. Let  $L \in (0, \infty)$ ,  $\zeta \in [0, 1)$ and  $p \geq \zeta$ . Assume that  $h \in C_{2,\text{loc}}^{p-\text{var}}(I; E)$  is such that for all  $(s, u, t) \in \Delta_I^2$  with  $\varpi_1(s, t) \leq L$ ,

$$|\delta h_{sut}| \le \omega_2(s,t)^{\frac{1}{\zeta}}.$$

Then there exists a unique path  $Ih \in C^{p-\text{var}}(I; E)$  such that  $Ih_a = 0$ ,  $\Lambda h := h - \delta Ih \in C_{2,\text{loc}}^{\zeta-\text{var}}(I; E)$ , and for all  $(s, t) \in \Delta_I$  with  $\varpi_1(s, t) \leq L$ ,

$$|(\Lambda h)_{st}| \le C_{\zeta} \varpi_2(s,t)^{\frac{1}{\zeta}},\tag{2.13}$$

for a universal positive constant  $C_{\zeta}$ . Moreover, for all  $(s,t) \in \Delta_I$ ,

$$\delta Ih_{st} = \lim_{|\mathfrak{p}| \to 0} \sum_{i=1}^{\#\mathfrak{p}-1} h_{t_i t_{i+1}},$$

where the limit is understood as a limit of nets over finite partitions  $\mathfrak{p} \in \mathcal{P}([s,t])$  of the interval  $[s,t] \subset I$  partially ordered by inclusion with mesh size  $|\mathfrak{p}|$  tending to zero.

# 3. Main results

## 3.1. Formulations of the rough incompressible Euler system

Let  $d \in \{2, 3, ...\}$ ,  $K \in \mathbb{N}$ ,  $p \in [2, 3)$ , and  $m \in \mathbb{N}$  be such that  $m \ge m_* := \lfloor \frac{d}{2} \rfloor + 2$ . For an arbitrarily given initial condition  $u_0 \in \mathring{H}^m_{\sigma}$ , geometric rough path  $\mathbf{Z} = (Z, \mathbb{Z}) \in C_g^{p-\operatorname{var}}(\mathbb{R}_+; \mathbb{R}^K)$ , and collection of vector fields  $\xi \in (W^{m+2,\infty}_{\sigma})^K$ , we consider the rough incompressible Euler system given by

$$\begin{cases} du + u \cdot \nabla u \, dt - \pounds_{\xi_k}^* u \, d\mathbf{Z}_t^k = -d\nabla q_t - dh_t & \text{on } (0,T] \times \mathbb{T}^d, \\ div \, u = 0 & \text{on } [0,T] \times \mathbb{T}^d, \\ \int_{\mathbb{T}^d} u \, dV = 0, \quad \int_{\mathbb{T}^d} q \, dV = 0 & \text{on } [0,T], \\ u = u_0, \quad q = 0, \quad h = 0 & \text{on } \{0\} \times \mathbb{T}^d, \end{cases}$$
(3.1)

where (see Section 2.1.2)  $\pounds_{\xi_k}^* u = -(\xi_k^j \partial_{x^j} u^i + u^j \partial_{x^i} \xi_k^j) e_i$ . Equation (3.1) is to be solved for an unknown divergence and mean-free vector field (velocity)  $u : [0,T] \times \mathbb{T}^d \to \mathbb{R}^d$ , mean-free scalar field ('time-integrated' pressure)  $q : [0,T] \times \mathbb{T}^d \to \mathbb{R}$ , and harmonic constant (time-integrated)  $h : [0,T] \to \mathbb{R}^d$ . The pressure q and harmonic constant h should be understood as Lagrange multipliers associated with the divergence-free and mean-free constraints, respectively.

In contrast to the unperturbed system (i.e.,  $\xi \equiv 0$  or  $\mathbf{Z} \equiv 0$ ), the system (3.1) does not preserve the mean of the initial condition if the  $\xi$ 's are not constant in space due to the term  $u^j \partial_{x^i} \xi^j_k$ . Indeed, upon formally integrating (3.1) over  $\mathbb{T}^d$ , all other terms vanish due to the periodic boundary conditions. Thus, the Lagrangian multiplier h is required to enforce the constraint that the velocity u remains mean-free. It is worth noting that we impose the mean-free constraint because it simplifies our analysis. For details on how to avoid this assumption, we refer to [41], which establishes the existence of a strong solution of the associated viscous version of equation (3.1).

Applying the divergence and mean-free projection operator P to (3.1), we find

$$\begin{cases} \mathrm{d}u + P(u \cdot \nabla u) \, \mathrm{d}t - P \pounds_{\xi_k}^* u \, \mathrm{d}\mathbf{Z}_t^k = 0 \quad \text{on } (0, T] \times \mathbb{T}^d, \\ u = u_0 \quad \text{on } \{0\} \times \mathbb{T}^d, \end{cases}$$
(RE)

which is to be solved for  $u: [0, T] \to \mathring{H}^m_{\sigma}$ . We will now define the notion of solution of (RE) we will use throughout the paper, which is simply a specific case of Definition A.4. Formally, the definition can be obtained by integrating (RE) over an arbitrary interval and iterating the equation into the  $d\mathbf{Z}^k$ -integral twice. In fact, in the proof of local existence (see Section 5), we smooth out the path  $\mathbf{Z}$  and iterate in such a manner.

**Definition 3.1** ( $H^m$ -solution). Let  $m \ge m_*$  and  $\xi \in (W^{m+2,\infty}_{\sigma})^K$ . We say a bounded path  $u : [0,T] \to H^m$  is a  $H^m$ -solution of (RE) on the interval [0,T] if  $u|_{t=0} = u_0$  and

$$u_{st}^{P,\natural} \coloneqq \delta u_{st} + \int_{s}^{t} P(u_r \cdot \nabla u_r) \,\mathrm{d}r - P \pounds_{\xi_k}^* u_s Z_{st}^k - P \pounds_{\xi_k}^* P \pounds_{\xi_l}^* u_s \mathbb{Z}_{st}^{lk}, \quad (s,t) \in \Delta_T, \quad (3.2)$$

satisfies  $u^{P,\natural} \in C_{2,\text{loc}}^{\frac{p}{2}-\text{var}}([0,T]; \mathring{H}^{m-3}_{\sigma})$ . A bounded path  $u : [0,T] \to H^m$  is said to be a  $H^m$ -solution of (RE) on the interval [0,T) if u is a  $H^m$ -solution of (RE) on the interval [0,T'] for all 0 < T' < T.

**Remark 3.2.** An  $H^m$ -solution actually possesses additional time-smoothness properties, as proven in Theorem 3.7 (see, also, Theorem A.5).

In Section 6.1, given  $H^m$ -solution u of (RE) on an interval [0, T], we apply the sewing lemma (i.e., Lemma 2.2) to construct the rough integral  $\int_0^{\cdot} \pounds_{\xi_k}^* u_s d\mathbf{Z}_s^k \in C^{p-\text{var}}([0,T]; H^{m-3})$ ; that is, for all  $t \in [0,T]$ ,

$$\int_{0}^{t} \pounds_{\xi_{k}}^{*} u_{s} \, \mathrm{d}\mathbf{Z}_{s}^{k} := \lim_{\substack{|\mathfrak{p}| \to 0\\ \mathfrak{p} \in \mathcal{P}([0,t])}} \sum_{i=1}^{\#\mathfrak{p}-1} \left( \pounds_{\xi_{k}}^{*} u_{t_{i}} Z_{t_{i}t_{i+1}}^{k} + \pounds_{\xi_{k}}^{*} P \pounds_{\xi_{l}}^{*} u_{t_{i}} \mathbb{Z}_{t_{i}t_{i+1}}^{lk} \right).$$

It follows by the continuity and linearity of the divergence and mean-free projection map  $P: H^m \to \mathring{H}^m_{\sigma}$  that for all  $t \in [0, T]$ ,

$$u_t - u_0 + \int_0^t P(u_s \cdot \nabla u_s) \,\mathrm{d}s - \int_0^t P \pounds_{\xi_k}^* u_s \,\mathrm{d}\mathbf{Z}_s^k = 0,$$
(3.3)

where  $P \int_0^{\cdot} \pounds_{\xi_k}^* u_s \, \mathrm{d}\mathbf{Z}_s^k \in C^{p-\mathrm{var}}([0,T]; \mathring{H}_{\sigma}^{m-3})$  is defined from the projected expansion appearing in the right-hand-side of (3.2). The pressure q and harmonic constant h can then be recovered using the Helmholtz decomposition P = I - Q - H (see Section 2.1.1)

$$\nabla q_t := \int_0^t Q(u_s \cdot \nabla u_s) \,\mathrm{d}s - \int_0^t Q \pounds_{\xi_k}^* u_s \,\mathrm{d}\mathbf{Z}_s^k, \quad h_t := -\int_0^t H \pounds_{\xi_k}^* u_s \,\mathrm{d}\mathbf{Z}_s^k, \quad t \in [0,T].$$

**Proposition 3.3** (Recovery of pressure and harmonic constant). If u is a  $H^m$ -solution of (RE) on the interval [0,T], then there exists  $\int_0^{\cdot} \mathcal{L}_{\xi_k}^* u_s dZ_s^k \in C^{p-\text{var}}([0,T]; H^{m-3})$  and uniquely determined  $q \in C^{p-\text{var}}([0,T]; \mathring{H}^{m-2})$  and  $h \in C^{p-\text{var}}([0,T]; \mathbb{R}^d)$  initiating from zero such that for all  $t \in [0,T]$ ,

$$u_t - u_0 + \int_0^t u_s \cdot \nabla u_s \,\mathrm{d}s - \int_0^t \pounds_{\xi_k}^* u_s \,\mathrm{d}\mathbf{Z}_s^k = -\nabla q_t - h_t. \tag{3.4}$$

Using (2.8) and (2.6), we find that (3.1) can be expressed in terms of the associated co-vector  $u^{\flat}$  (see Section 2.1.2):

$$du^{\flat} + \pounds_{u} u^{\flat} dt + \pounds_{\xi_{k}} u^{\flat} d\mathbf{Z}_{t}^{k} = -\mathbf{d}(dq_{t} - 2^{-1}|u|^{2}dt) - dh_{t}^{\flat}.$$
 (3.5)

Let  $\omega = \mathbf{d}u^{\flat} = (\partial_{x^i}u_j - \partial_{x^j}u_i)e^i \wedge e^j$ . By applying the exterior derivative  $\mathbf{d}$  to (3.5) and using that  $\mathbf{d}\pounds_v = \pounds_v \mathbf{d}$  we arrive at the vorticity formulation:

$$\begin{cases} \mathrm{d}\omega + \pounds_{u}\omega\,\mathrm{d}t + \pounds_{\xi_{k}}\omega\,\mathrm{d}\mathbf{Z}_{t}^{k} = 0 \quad \mathrm{on} \quad (0,T] \times \mathbb{T}^{d}, \\ u = \mathrm{BS}\,\omega \quad \mathrm{on} \quad (0,T] \times \mathbb{T}^{d}, \\ \omega = \mathrm{d}u_{0}^{\flat} \quad \mathrm{on} \quad \{0\} \times \mathbb{T}^{d}, \end{cases}$$
(3.6)

where for a divergence-free vector field v we have (see Section 2.1.2)

$$\pounds_{v}\omega = \sum_{i < j} (v^{q}\partial_{x^{q}}\omega_{ij} + \omega_{qj}\partial_{x^{i}}v^{q} + \omega_{iq}\partial_{x^{j}}v^{q})e^{i} \wedge e^{j}$$
$$= \sum_{i < j} (\partial_{x^{i}}(v^{q}\omega_{qj}) - \partial_{x^{j}}(v^{q}\omega_{qi}))e^{i} \wedge e^{j} = \mathbf{d}[\omega(v, \cdot)].$$
(3.7)

Recall that (see Section 2.1.1) for given  $s \in \mathbb{R}$ ,  $\mathring{H}^s_{\mathbf{d}} := \{\omega \in \mathring{H}^s(\mathbb{T}^d; \Lambda^2(\mathbb{R}^d)^*) : \mathbf{d}\omega = 0\}$  and that BS :  $\mathring{H}^{m-1}_{\mathbf{d}} \to \mathring{H}^m_{\sigma}$  denotes the inverse of  $\mathbf{d}\mathbf{b}$ . The Cartan formulation (3.7) of the Lie derivative  $\pounds_v \omega = \mathbf{d}[\omega(v, \cdot)]$  implies that the dynamics preserve the property that  $\mathbf{d}\omega = 0$  and  $\hat{\omega}(0) = 0$ , and thus no Lagrange multipliers are needed to enforce these constraints. In contrast, the velocity equation requires the pressure and the harmonic constant to enforce the divergence-free constraint.

We summarize the equivalence between the velocity and vorticity formulation in the following proposition. The direct implication is a simple consequence of (2.8) and (2.6) and the converse follows from the properties of the Biot-Savart operator presented in Section 2.1.1.

**Proposition 3.4** (Vorticity formulation). If u is a  $H^m$ -solution of (RE) on the interval [0,T], then  $\omega = \mathbf{d}u^{\flat} : [0,T] \to \mathring{H}^{m-1}_{\mathbf{d}}$  is bounded and

$$\omega_{st}^{\natural} := \delta \omega_{st} + \int_{s}^{t} \pounds_{u_{r}} \omega_{r} \, \mathrm{d}r + \pounds_{\xi_{k}} \omega_{s} Z_{st}^{k} - \pounds_{\xi_{k}} \pounds_{\xi_{l}} \omega_{s} \mathbb{Z}_{st}^{lk}, \quad (s,t) \in \Delta_{T},$$
(3.8)

satisfies  $\omega^{\natural} \in C_{2,\text{loc}}^{\frac{p}{2}-\text{var}}([0,T]; \mathring{H}_{\mathbf{d}}^{m-4})$ . Moreover,  $\omega^{\natural} = \mathbf{d} \flat u^{P,\natural}$ . Conversely, if  $\omega$  and  $\omega^{\natural}$  belong to the aforementioned spaces and satisfy (3.8) with  $\omega_0 := \mathbf{d} u_0^{\flat}$  and  $u := BS \omega$ , then u is  $H^m$ -solution of (RE) and  $u^{P,\natural} = BS \omega^{\natural}$ .

**Remark 3.5.** In dimension two, the vorticity  $\omega$  can be identified with a scalar-valued function  $\tilde{\omega} = \operatorname{curl} u = \star \omega = \omega_{12}$ . Using (2.9), we find that

$$\mathrm{d}\tilde{\omega} + u \cdot \nabla\tilde{\omega}\,\mathrm{d}t + \xi_k \cdot \nabla\tilde{\omega}\,\mathrm{d}\mathbf{Z}_t^k = 0. \tag{3.9}$$

The scalar transport structure then implies that  $L^p$ -norms of  $\tilde{\omega}$  are conserved, which is used to prove global well-posedness (see (3.14) in Theorem 3.10). In dimension three, the vorticity  $\omega$  can be identified with a vector field  $\hat{\omega} = \operatorname{curl} u = \sharp \star \omega = \omega_{23}e_1 + \omega_{31}e_2 + \omega_{12}e_3$ . Applying (2.10), we get

$$\mathrm{d}\hat{\omega} + [u,\hat{\omega}]\,\mathrm{d}t + [\xi_k,\hat{\omega}]\,\mathrm{d}\mathbf{Z}_t^k = 0.$$

**Remark 3.6.** In the vorticity equation, neither Lagrange multipliers, nor projections, appear. The structure of the operators (i.e., (3.7)) imply that the mean-free and exterior-derivative-free condition is preserved by the dynamics. This fact plays an important role in the proof of Theorem 4.2 in which we obtain a priori solution estimates of (RE) using the vorticity formulation. More precisely, the linear hyperbolic symmetric structure of the operator in the rough part of (3.6) enables us to develop an 'equation' for the 'square'  $\omega \otimes \omega$ . Unlike in the stochastic setting, there is no Burkholder-Davis-Gundy inequality that can be used to estimate the rough integral, and thus we make use of the method of unbounded rough drivers [9], and more precisely Theorem A.5 to obtain remainder

estimates for the equation and the 'squared' equation. It is not clear how to obtain a priori estimates of (RE) directly due to the projection operators, or equivalently the presence of the pressure and harmonic constant. As a consequence, altering the structure of the operator appearing in the  $d\mathbf{Z}$ -term in (3.1) even in a multiplicative way directly impacts the structure of the vorticity equation and prevents us from obtaining a priori solution estimates.

## 3.2. Statement of the main results

Recall that we always work under the assumption  $m \ge m_* := \lfloor \frac{d}{2} \rfloor + 2$ . Our first main results establish local well-posedness and the existence of a maximally extended solution in  $H^m$  of (RE). Recall that  $\varpi_{\mathbf{Z}}$  is the control of  $\mathbf{Z}$ , which is defined in (2.12).

**Theorem 3.7** (Local well-posedness). Assume that  $u_0 \in \mathring{H}^m_{\sigma}$  and  $\xi \in (W^{m+2}_{\sigma})^K$ . For all  $T_* > 0$  satisfying

$$e^{C_1(1+\varpi_{\mathbf{Z}}(0,T_*))}T_* < |u_0|_{H^{m_*}}^{-1},$$
(3.10)

where  $C_1 = C_1(p, d, |\xi|_{W^{m_*+2,\infty}})$  depends in an increasing way on  $|\xi|_{W^{m_*+2,\infty}}$ , there exists a unique  $H^m$ -solution  $u \in C_w([0, T_*]; \mathring{H}^m_{\sigma}) \cap C^{p-\operatorname{var}}([0, T_*]; \mathring{H}^{m-1}_{\sigma})$  of (RE) on the interval  $[0, T_*]$  satisfying

$$\sup_{t \le T_*} |u_t|_{H^{m_*}} \le \frac{e^{C_1(1 + \varpi_{\mathbf{Z}}(0, T_*))}}{|u_0|_{H^{m_*}}^{-1} - e^{C_1(1 + \varpi_{\mathbf{Z}}(0, T_*))}T_*}.$$
(3.11)

Moreover, if  $m > m_*$ , there is a constant  $C_2 = C_2(p, d, m, |\xi|_{W^{m+2,\infty}})$  which increases with  $|\xi|_{W^{m_*+2,\infty}}$ , such that

$$\sup_{t \le T_*} |u_t|_{H^m} \le \sqrt{2} \exp\left(C_2\left(\int_0^{T_*} |\nabla u_s|_{L^{\infty}} \,\mathrm{d}s + \varpi_{\mathbf{Z}}(0, T_*)\right)\right) |u_0|_{H^m}.$$
(3.12)

If  $\xi \in (W^{m+4,\infty}_{\sigma})^K$ , then  $u \in C([0,T_*]; \mathring{H}^m_{\sigma})$ .

**Corollary 3.8** (Maximally extended solution). Assume that  $u_0 \in \mathring{H}_{\sigma}^m$  and  $\xi \in (W_{\sigma}^{m+2})^K$ . Then there exists a unique maximally extended  $H^m$ -solution  $u \in C_w([0, T_{\max}); \mathring{H}_{\sigma}^m) \cap C^{p-\text{var}}([0, T_{\max}); \mathring{H}_{\sigma}^{m-1})$  of (RE) on the interval  $[0, T_{\max})$ . The time  $T_{\max}$  is uniquely specified by the property if  $T_{\max} < \infty$ , then  $\limsup_{t \uparrow T_{\max}} |u_t|_{H^{m_*}} = \infty$ . If  $\xi \in (W_{\sigma}^{m+4,\infty})^K$ , then  $u \in C([0, T_{\max}); \mathring{H}_{\sigma}^m)$ .

The next theorem extends the Beale-Kato-Majda (BKM) blow-up criterion [10].

**Theorem 3.9** (BKM blow-up criterion). Assume that  $u_0 \in \mathring{H}_{\sigma}^m$  and  $\xi \in (W_{\sigma}^{m_*+4,\infty})^K$ if  $m = m_*$  and  $\xi \in (W_{\sigma}^{m+2})^K$  if  $m > m_*$ . Let u denote the maximally extended  $H^m$ solution of (RE) and  $\omega = \mathbf{d}u^{\flat}$  denote its vorticity. Then there are constants  $C_1 = C_1(d,m)$  and  $C_2 = C_2(p,d,m,|\xi|_{W^{m+2,\infty}})$  depending in an increasing way on  $|\xi|_{W^{m+2,\infty}}$ such that for all  $T < T_{\max}$ ,

$$\sup_{t \le T} |u_t|_{H^m} \le C_1 (1 + |u_0|_{H^m}) \exp\left(C_2 (1 + \varpi_{\mathbf{Z}}(0, T)) \exp\left(C_2 \int_0^T |\omega_s|_{L^{\infty}} \,\mathrm{d}s\right)\right). \quad (3.13)$$

Moreover,  $T_{\max} < \infty$  if and only if  $\int_0^{T_{\max}} |\omega_t|_{L^{\infty}} dt = \infty$ .

In dimension two, we obtain global well-posedness.

**Theorem 3.10** (Global well-posedness in 2d). Let d = 2. Assume that  $u_0 \in \mathring{H}^m_{\sigma}$  and  $\xi \in (W^{m+2}_{\sigma})^K$ . Let u denote the maximally extended  $H^m$ -solution and  $\tilde{\omega} = \operatorname{curl} u$  denote its scalar vorticity. Then  $T_{\max} = \infty$  and for all  $t \geq 0$  and  $p \in [2, \infty]$ ,

$$|\tilde{\omega}_t|_{L^p} = |\tilde{\omega}_0|_{L^p}.\tag{3.14}$$

Moreover, there are constants  $C_1 = C_1(m)$  and  $C_2 = C_2(p, m, |\xi|_{W^{m+2,\infty}})$  that increase with  $|\xi|_{W^{m+2,\infty}}$  such that for all  $t \ge 0$ 

$$|u_t|_{H^m} \le C_1(1+|u_0|_{H^m}) \exp\left(C_2(1+\varpi_{\mathbf{Z}}(0,t))\exp\left(C_2|\tilde{\omega}_0|_{L^{\infty}}t\right)\right).$$
(3.15)

The following corollary establishes the continuity of the solution map with respect to the data.

**Corollary 3.11** (Continuous dependence on data). Assume that  $u_0 \in \mathring{H}_{\sigma}^m$  and  $\xi \in (W_{\sigma}^{m+2})^K$ . Further, assume that  $\{\mathbf{Z}^n\}_{n=1}^{\infty}$  converges to  $\mathbf{Z}$  in  $C_g^{p-\text{var}}$  and that  $\{(u_0^n, \xi^n)\}_{n=1}^{\infty}$  is bounded in  $\mathring{H}_{\sigma}^m \times W_{\sigma}^{m+2,\infty}$  and converges to  $(u_0,\xi)$  in  $\dot{L}_{\sigma}^2 \times (W_{\sigma}^{2,\infty})^K$ . Denote by u and  $\{u^n\}_{n=1}^{\infty}$  the maximally extended solutions corresponding to the data  $(u_0,\xi,\mathbf{Z})$  and  $\{(u_0^n,\xi^n,\mathbf{Z}^n)\}_{n=1}^{\infty}$ , and let (q,h) and  $\{(q^n,h^n)\}_{n=1}^{\infty}$  denote the associated pressures and harmonic constants.

- If d = 2, then {u<sup>n</sup>}<sub>n=1</sub><sup>∞</sup> converges to u in C([0,∞); H<sup>m-ϵ</sup><sub>σ</sub>) for any ϵ > 0 and in the weak-star topology of L<sup>∞</sup>([0,∞); H<sup>m</sup><sub>σ</sub>). Moreover, {(q<sup>n</sup>, h<sup>n</sup>)}<sub>n=1</sub><sup>∞</sup> converge to (q, h) in C([0,∞); H<sup>m-2-ϵ</sup>) × C([0,∞); ℝ<sup>d</sup>) for any ϵ > 0.
- If  $m > m_*$ , then for all  $T < T_{\max}$  there exists an  $N(T) \in \mathbb{N}$  such that  $\{u^n\}_{n=N(T)}^{\infty}$  converges to u in  $C([0,T]; \mathring{H}_{\sigma}^{m-\epsilon})$  for any  $\epsilon > 0$  and in the weak-star topology of  $L^{\infty}([0,T]; \mathring{H}_{\sigma}^m)$ . Moreover,  $\{(q^n, h^n)\}_{n=N(T)}^{\infty}$  converge to (q,h) in  $C([0,T]; \mathring{H}^{m-2-\epsilon}) \times C([0,T]; \mathbb{R}^d)$  for any  $\epsilon > 0$ .

**Remark 3.12** (Rough Navier-Stokes and inviscid limit). Let  $\nu > 0$ . In [41], two of the authors considered the rough Navier-Stokes system given by

$$\begin{cases} \mathrm{d}u + u \cdot \nabla u \, \mathrm{d}t - \pounds_{\xi_k}^* u \, \mathrm{d}Z_t^k = \nu \Delta u - \mathrm{d}\nabla q_t - \mathrm{d}h_t, & \text{on } [0,T] \times \mathbb{T}^d \\ \mathrm{div} \, u = 0 & \text{on } [0,T] \times \mathbb{T}^d, \\ \int_{\mathbb{T}^d} u \, \mathrm{d}V = 0, \quad \int_{\mathbb{T}^d} q \, \mathrm{d}V = 0 & \text{on } [0,T], \\ u = u_0, \quad q = 0, \quad h = 0 & \text{on } \{0\} \times \mathbb{T}^d, \end{cases}$$

for  $d \in \{2,3\}$ . We showed that for arbitrarily given  $u_0 \in \mathring{H}^1_{\sigma}$ , there exists a time  $T_* = T_*(d, \varpi_{\mathbf{Z}}, |\xi|_{W^{3,\infty}})$  and a solution  $u \in L^2([0, T_*]; \mathring{H}^2_{\sigma}) \cap L^\infty([0, T_*]; \mathring{H}^1_{\sigma})$ . In fact, the solution was not constrained to be mean-free in [41], but the proof goes through in a simpler manner. In dimension two, we also proved that there is a unique global solution of the Navier-Stokes system. With minor changes in the present paper, we can derive the existence and uniqueness of a maximally extended solution  $u \in L^2([0, T_{\max}); \mathring{H}^{m+1}_{\sigma}) \cap C([0, T_{\max}); \mathring{H}^m_{\sigma})$  in any dimension  $d \geq 2$ . Moreover, we can show that if d = 2 or  $m > m_*$ , then for an arbitrarily given  $\{\nu^n\}_{n=1}^{\infty}$  converging to zero, the corresponding sequence of Navier-Stokes solutions  $\{u^n\}_{n=1}^{\infty}$  converges to the Euler solution in  $C([0, T_{\max}); \mathring{H}^{m-\epsilon}_{\sigma})$  for any  $\epsilon > 0$  and in the weak-star topology of  $L^{\infty}([0, T_{\max}); \mathring{H}^m_{\sigma})$ . The inviscid limit stochastic Naiver-Stokes equations with additive and multiplicative noise has been studied in [35,7].

## 3.2.1. Applications to stochastic partial differential equations

In what follows, we will discuss the Wong-Zakai approximation of the Euler stochastic partial differential equation (SPDE) driven by Brownian motion in dimension two. Let  $B = \{B^k\}_{k=1}^K$  denote a collection of K-independent Brownian motions adapted to a filtered probability space  $(\mathcal{Z}, \mathcal{F}, \mathbb{F} = \{\mathcal{F}_t\}_{t\geq 0}, \mathbb{P})$  satisfying the usual conditions. In [48], it was shown that for every  $\mathcal{F}_0$ -adapted initial velocity  $u_0 \in \mathring{H}^{m-1}_{\sigma}$ , there exists a unique  $\mathbb{F}$ -adapted process  $\bar{u} \in C([0,\infty); \mathring{H}^m_{\sigma})$  such that  $\mathbb{P}$ -a.s. for all  $t \in [0,\infty)$ ,

$$\bar{u}_t - u_0 + \int_0^t P(\bar{u}_s \cdot \nabla \bar{u}_s) \,\mathrm{d}s - \int_0^t P \pounds_{\xi_k}^* \bar{u}_s \circ \mathrm{d}B_s^k = 0, \qquad (3.16)$$

where equality is understood in  $L^2$  and the stochastic integral is understood in the Stratonovich sense.

By Proposition 3.5 of [29],  $\mathbb{P}$ -a.s.,  $\mathbf{B} = (\delta B, \mathbb{B}^{\text{strat}}) \in C_g^{p-\text{var}}(\mathbb{R}_+; \mathbb{R}^K)$ , where for each  $l, k \in \{1, \ldots, K\}$  and  $(s, t) \in \Delta_{[0,\infty)}$ :

$$\mathbb{B}_{st}^{\text{strat};lk} := \int_{s}^{t} (B_{r}^{l} - B_{s}^{l}) \circ \mathrm{d}B_{r}^{k}.$$

By Proposition 3.6 of [29], the canonical lift of a dyadic piecewise-linear approximation  $\{B^n\}_{n=1}^{\infty}$  of the Brownian motion B, denoted by  $\{\mathbf{B}^n\}_{n=1}^{\infty} = \{(B^n, \mathbb{B}^n)\}_{n=1}^{\infty} \in C_g^{1-\operatorname{var}}(\mathbb{R}_+; \mathbb{R}^K)$ , converges  $\mathbb{P}$ -a.s. to  $\mathbf{Z}$  in  $C_g^{p-\operatorname{var}}(\mathbb{R}_+; \mathbb{R}^K)$ .

By Corollary 3.10,  $\mathbb{P}$ -a.s., corresponding to the data  $(u_0, \mathbf{B}, \xi)$  and  $\{(u_0, \mathbf{B}^n, \xi)\}_{n=1}^{\infty}$ , there exist unique solutions  $u, \{u^n\}_{n=1}^{\infty} \in C([0, \infty), \mathring{H}_{\sigma}^m)$  of (3.9). Owing to Corollary 3.11, the sequence  $\{u^n\}_{n=1}^{\infty}$  converges to u in  $C([0, T_{\max}); \mathring{H}_{\sigma}^{m-\epsilon})$  for any  $\epsilon > 0$ and in the weak-star topology of  $L^{\infty}([0, \infty); \mathring{H}_{\sigma}^m)$ .

For every  $t \in \mathbb{R}_+$ , the map  $\mathbf{Z}|_{[0,t]} \in C_g^{p-\text{var}}([0,t];\mathbb{R}^K) \mapsto u \in C([0,t], \mathring{H}_{\sigma}^{m-})$  is continuous and the map  $\zeta \in \mathcal{Z} \mapsto \mathbf{B}(\zeta)|_{[0,t]} \in C_g^{p-\text{var}}([0,t];\mathbb{R}^K)$  is measurable. Thus, we conclude that the composition of the two maps is measurable, and hence that the solution u is **F**-adapted. As explained above (see (3.3)),  $\mathbb{P}$ -a.s., for all  $t \in [0,T]$ , we have

$$u_t - u_0 + \int_0^t P(u_r \cdot \nabla u_r) \,\mathrm{d}r - \int_0^t P \pounds_{\xi_k}^* u_s \,\mathrm{d}\mathbf{B}_s^k = 0.$$

It can be shown (see, e.g., [29, Corollary 5.2]) that  $\mathbb{P}$ -a.s. for all  $t \in [0, T]$ ,

$$\int_{0}^{t} P \pounds_{\xi_{k}}^{*} u_{s} \, \mathrm{d}\mathbf{B}_{s}^{k} = \int_{0}^{t} P \pounds_{\xi_{k}}^{*} u_{s} \circ \, \mathrm{d}B_{s}^{k}.$$

Therefore, we obtain the following Wong-Zakai approximation result.

**Theorem 3.13** (Wong-Zakai approximation). The stochastic process u is indistinguishable from  $\bar{u}$ , and  $\mathbb{P}$ -a.s.,  $\{u^n\}_{n=1}^{\infty}$  converges to u in  $C([0,\infty); \mathring{H}_{\sigma}^{m-\epsilon})$  for any  $\epsilon > 0$  and in the weak-star topology of  $L^{\infty}([0,\infty); \mathring{H}_{\sigma}^m)$ .

**Remark 3.14** (Strook-Varadhan support theorem, large deviations principle, and random dynamical system). As in [29, Section 9.3], using the continuity of the solution map, one can characterize the support of the law of the SPDE (3.16) in  $C([0, T_{\max}); \mathring{H}_{\sigma}^{m-\epsilon})$  for any  $\epsilon > 0$  in terms of Cameron-Martin space and prove a large deviations principle (making use of contraction principle) for

$$\mathrm{d}u + P(u \cdot \nabla u) \,\mathrm{d}t - \epsilon P \pounds_{\xi_k}^* u \circ \mathrm{d}B_t^k = 0,$$

which concerns small-noise deviations about solutions of the Euler system

$$\partial_t u + P(u \cdot \nabla u) = 0.$$

Moreover, if the driving rough path  $\mathbf{Z} : \mathcal{Z} \to C_g^{p-\text{var}}$  is a continuous *p*-rough path cocycle,<sup>1</sup> viz.

 $<sup>^{1}</sup>$  This is the appropriate notion of random shifts in the rough path which enables the construction of a random dynamical system.

$$Z_{0,s+t}(\zeta) = Z_{0,s}(\zeta) + Z_{0,t}(\theta_s \zeta),$$
$$\mathbb{Z}_{0,s+t}(\zeta) = \mathbb{Z}_{0,s}(\zeta) + \mathbb{Z}_{0,t}(\theta_s \zeta) + Z_{0,s}(\zeta) \otimes Z_{0,t}(\theta_s \zeta), \quad \forall (s,t) \in \Delta_T,$$

where  $\theta_s$  is the time-shift  $\theta_s \zeta_t = \zeta_{t+s} - \zeta_s$ , then the 2D-system (RE) generates a continuous random dynamical system on  $C([0, T_{\max}); \mathring{H}_{\sigma}^{m-\epsilon})$  for any  $\epsilon > 0$ . See [41].

**Remark 3.15** (Gaussian rough paths). Our main results also yield a solution theory in any dimension  $d \in \{2, 3, ..., \}$  for a class of Euler SPDEs driven by fractional Brownian or more general Gaussian processes transport noise. We refer the reader to [29] for more details about the lifts of Gaussian processes to the space of geometric rough paths. In the introduction, we have discussed the potential applications of such models to stochastic parameterizations of sub-grid scales of ideal fluids.

## 3.2.2. Critical points of the Clebsch and Hamilton-Pontryagin variational principles

For an arbitrarily given Fréchet space E and time T > 0, denote by  $\mathcal{D}_Z([0,T]; E)$  the space of E-valued **Z**-controlled rough paths on the interval [0,T]. We refer to [29] for precise definitions. In [18], we introduced the Clebsch action functional

$$S^{\text{Clb}_{\mathbf{Z}}}(u, (\lambda^q)_{q=1}^d, (a^q)_{q=1}^d) = \int_0^T \frac{1}{2} |u_t|_{L^2}^2 \, \mathrm{d}t + \sum_{q=1}^d \left(\lambda_t^q, \mathrm{d}a_t^q + \pounds_{u_t} a_t^q \, \mathrm{d}t + \pounds_{\xi_k} a_t^q \, \mathrm{d}\mathbf{Z}_t^k\right)_{L^2},$$

defined for  $u \in C^{p-\text{var}}([0,T]; \mathring{C}^{\infty}_{\sigma})$  and  $(\lambda^q)_{q=1}^d, (a^q)_{q=1}^d \in \mathcal{D}_{\mathbf{Z}}([0,T]; C^{\infty})$ . We showed that critical points are characterized by the following system of RPDEs:

$$\begin{cases} \mathrm{d}u - P \pounds_u^* u \mathrm{d}t - P \pounds_{\xi_k}^* u \mathrm{d}\mathbf{Z}_t^k = 0, & u = P(\sum_{q=1}^d a^q \nabla \lambda^q), \\ \mathrm{d}\lambda + \pounds_u u \mathrm{d}t + \pounds_{\xi_k} \lambda \mathrm{d}\mathbf{Z}_t^k = 0, \\ \mathrm{d}a + \pounds_u a \mathrm{d}t + \pounds_{\xi_k} a \mathrm{d}\mathbf{Z}_t^k = 0. \end{cases}$$

We also introduced the Hamilton-Pontryagin action functional

$$S^{\mathrm{HP}\mathbf{z}}(u,\phi,\lambda) = \int_{0}^{T} \frac{1}{2} |u_t|_{L^2}^2 \,\mathrm{d}t + \left(\lambda_t, \mathrm{d}\phi_t \circ \phi_t^{-1} - u_t \,\mathrm{d}t - \xi_k \,\mathrm{d}\mathbf{Z}_t^k\right)_{L^2},\tag{3.17}$$

defined for

$$(u,\lambda,\phi) \in C^{p-\operatorname{var}}([0,T]; \mathring{C}^{\infty}_{\sigma}) \times \mathcal{D}_{\mathbf{Z}}([0,T]; \mathring{C}^{\infty}_{\sigma}) \times \operatorname{Diff}_{\mathbf{Z}}([0,T]; \mathbb{T}^{d}).$$

The space  $\text{Diff}_{\mathbf{Z}}([0,T]; \mathbb{T}^d)$  consists of all rough flows (i.e., diffeomorphisms on  $\mathbb{T}^d$ )  $\{\phi_t\}_{t\in[0,T]}$  (see [18]) of the form

D. Crisan et al. / Journal of Functional Analysis 283 (2022) 109632

$$\begin{cases} \mathrm{d}\phi = v \circ \phi \,\mathrm{d}t + \sigma_k \circ \phi \,\mathrm{d}\mathbf{Z}_t^k, \ t \in (0,T], \\ \phi_0 = \mathrm{id}, \end{cases}$$

for some  $(v, \sigma) \in C^{p-\text{var}}([0, T]; \mathring{C}^{\infty}_{\sigma}) \times C^{\infty}([0, T]; (C^{\infty}_{\sigma})^{K})$ . Moreover, for  $\phi \in \text{Diff}_{\mathbf{Z}}([0, T]; \mathbb{T}^{d})$ , the integral in (3.17) is defined by

$$\int_{0}^{T} \langle \lambda_t, \mathrm{d}\phi_t \circ \phi_t^{-1} \rangle := \int_{0}^{T} (\lambda_t, v_t)_{L^2} \, \mathrm{d}t + (\lambda_t, \sigma_t)_{L^2} \, \mathrm{d}\mathbf{Z}_t^k.$$

We refer the reader to [18] for the specifics of how variations of  $\text{Diff}_{\mathbf{Z}}([0,T]; \mathbb{T}^d)$  are defined. In [18], we showed that if  $\mathbf{Z}$  is truly rough (see [29, Definition 6.3]), then  $(u, \phi, \lambda)$  is a critical point of  $S^{\text{HP}_{\mathbf{Z}}}$  if and only if

$$\begin{cases} \mathrm{d}u - P \pounds_u^* u \mathrm{d}t - P \pounds_{\xi_k}^* u \mathrm{d}\mathbf{Z}_t^k = 0, & u = \lambda, \\ \mathrm{d}\phi = u \circ \phi \, \mathrm{d}t + \xi_k \circ \phi \, \mathrm{d}\mathbf{Z}_t^k. \end{cases}$$

In [18], we showed that critical points satisfy a Kelvin circulation balance law: for an arbitrarily given smooth closed curve  $\gamma \subset \mathbb{T}^d$  and all  $t \in [0, T]$ ,

$$\mathrm{d} \oint_{\phi_t(\gamma)} u_t^{\flat} = \oint_{\phi_t(\gamma)} \left( \mathrm{d} u_t^{\flat} + \pounds_{u_t} u_t^{\flat} \, \mathrm{d} t + \pounds_{\xi_k} u_t^{\flat} \, \mathrm{d} \mathbf{Z}_t^k \right) = 0.$$

Assume that  $u_0 \in \mathring{C}^{\infty}_{\sigma}$  and  $\xi \in (\mathring{C}^{\infty}_{\sigma})^K$ . Denote by  $u \in C^{p-\operatorname{var}}([0, T_{\max}); \mathring{C}^{\infty}_{\sigma})$  the unique maximally extended  $H^{\infty}$ -solution of (RE) and let  $T < T_{\max}$ . Let  $\{\phi_t\}_{t \in [0,T]} \in \operatorname{Diff}_{\mathbf{Z}}([0,T]; \mathbb{T}^d)$  denote the rough flow (see, e.g., [13,29,18]) satisfying

$$\begin{cases} \mathrm{d}\phi = u \circ \phi \,\mathrm{d}t + \xi_k \circ \phi \,\mathrm{d}\mathbf{Z}_t^k, \ t \in (0,T], \\ \phi_0 = \mathrm{id}. \end{cases}$$

It follows that  $(u, \phi, u)$  is a critical point of  $S^{\text{HP}\mathbf{z}}$  provided **Z** is truly rough.

To show that we can construct a critical point of  $S^{\text{Clb}\mathbf{z}}$ , we must find  $\lambda$  and a such that the so-called Clebsch representation  $u^{\flat} = P(\sum_{q=1}^{d} a^q \mathbf{d}\lambda^q)$  holds. Let  $\lambda = \phi^{-1}$  and  $a = u_0 \circ \phi^{-1}$ . By Theorem 3.3 in [18], we have

$$\begin{cases} \mathrm{d}\lambda + \pounds_u u \mathrm{d}t + \pounds_{\xi_k} \lambda \mathrm{d}\mathbf{Z}_t^k = 0, \\ \mathrm{d}a + \pounds_u a \mathrm{d}t + \pounds_{\xi_k} a \mathrm{d}\mathbf{Z}_t^k = 0. \end{cases}$$

Proceeding as in [24, Lemma 3] and applying the product rule for geometric rough paths (see, e.g., [29,18]), it follows that  $v = \sum_{q=1}^{d} a^{q} \mathbf{d} \lambda^{q}$  satisfies the linear RPDE

$$\mathrm{d}v - \pounds_u^* v \,\mathrm{d}t - \pounds_{\xi_k}^* v \,\mathrm{d}\mathbf{Z}_t^k = 0.$$

By virtue of identity (2.7), we find that  $\bar{v} := Pv \in C^{p-\text{var}}([0,T]; \mathring{H}^m_{\sigma})$  is a solution of the constrained (to the space of divergence and mean-free vector-fields) linear RPDE given by

$$\mathrm{d}\bar{v} - P \pounds_v^* \bar{v} \,\mathrm{d}t - P \pounds_{\mathcal{E}_v}^* \bar{v} \,\mathrm{d}Z_t^k = 0. \tag{3.18}$$

As in Proposition 3.4, using the operators db and BS we can establish an equivalence between solutions of (3.18) and solutions of the unconstrained linear RPDE

$$\mathrm{d}\bar{\omega} + \pounds_u \bar{\omega} \,\mathrm{d}t + \pounds_{\xi_k} \bar{\omega} \,\mathrm{d}\mathbf{Z}_t^k = 0.$$

Thus, by the uniqueness of solutions of unconstrained linear RPDEs (see, e.g., (4.12) in Theorem 4.4 or [9]), we can deduce uniqueness of solutions of (3.18). Since u also solves (2.7), and hence (3.18), we deduce that

$$u = P\left(\sum_{q=1}^{d} a^{q} \nabla \lambda^{q}\right) = P(\nabla \phi^{-1} u_{0} \circ \phi^{-1}) \quad \Leftrightarrow \quad u^{\flat} = P\phi_{*}u_{0}^{\flat}, \tag{3.19}$$

where with abuse of notation we denote by  $P \in \mathcal{L}(H^s(\mathbb{T}^d; (\mathbb{R}^d)^*); H^s_{\delta}(\mathbb{T}^d; (\mathbb{R}^d))$  the corresponding projection associated with the one-form Hodge decomposition (see Section 2.1.1). It follows that  $(u, \lambda, a)$  is a critical point of  $S^{\text{Clb}_{\mathbf{Z}}}$ . The above representation extends the well-known Weber formula (see, e.g., [20]). Since the exterior derivative  $\mathbf{d}$  is natural with the push-forward, we also obtain the following representation of the vorticity two-form  $\omega = \phi_* \omega_0$ .

# 4. A priori estimates

The goal of this section is to establish a priori estimates of remainders and solutions of (RE).

## 4.1. Remainder estimates

**Theorem 4.1** (Remainder estimates). Let u be an  $H^m$ -solution of (RE) on the interval [0,T] and  $\varpi_{\mu}(s,t) := \int_{s}^{t} |\nabla u_r|_{L^{\infty}} |u_r|_{H^m} dr$ ,  $(s,t) \in \Delta_T$ . Then there exists a constant  $C = C(p,d,m,|\xi|_{W^{m+2,\infty}})$  which increases with  $|\xi|_{W^{m+2,\infty}}$  such that for all  $(s,t) \in \Delta_T$  with  $C\varpi_{\mathbf{Z}}(s,t) \leq 1$ , it holds that

$$|u^{P,\natural}|_{\frac{p}{2}-\operatorname{var};[s,t];H^{m-3}}^{\frac{p}{2}} \leq C\left(\sup_{s\leq r\leq t}|u_{r}|_{H^{m}}\varpi_{\mathbf{Z}}(s,t)^{\frac{3}{p}} + \varpi_{\mu}(s,t)\varpi_{\mathbf{Z}}(s,t)^{\frac{1}{p}}\right).$$

Moreover, for all  $(s,t) \in \Delta_T$  with  $C\varpi_{\mathbf{Z}}(s,t) + C\varpi_{\mu}(s,t) \, \mathrm{d}r \leq 1$ , it holds that

$$|u^{P,\sharp}|_{\frac{p}{2}-\operatorname{var};[s,t];H^{m-2}}^{\frac{p}{2}} \leq C\left(\varpi_{\mu}(s,t) + \sup_{s \leq r \leq t} |u_{r}|_{H^{m}} \varpi_{\mathbf{Z}}(s,t)^{\frac{2}{p}}\right)$$
$$|u|_{p-\operatorname{var};[s,t];H^{m-1}}^{p} \leq C\left(\varpi_{\mu}(s,t) + \sup_{s \leq r \leq t} |u_{r}|_{H^{m}} \left(\varpi_{\mu}(s,t)^{\frac{1}{p}} + \varpi_{\mathbf{Z}}(s,t)^{\frac{1}{p}}\right)\right),$$

where

$$u_{st}^{P,\sharp} := \delta u_{st} - P\pounds_{\xi_k}^* u_s Z_{st}^k = -\int_s^t P(u_r \cdot \nabla u_r) \,\mathrm{d}r + P\pounds_{\xi_k}^* P\pounds_{\xi_l}^* u_s \mathbb{Z}_{st}^{lk} + u_{st}^{P,\natural}$$

**Proof.** By Proposition 3.4,  $\omega = \mathbf{d}u^{\flat} : [0,T] \to \mathring{H}^{m-1}_{\mathbf{d}}$  is bounded and

$$\omega_{st}^{\natural} := \mathbf{d} \flat u^{P,\natural} = \delta \omega_{st} + \int_{s}^{t} \pounds_{u_{r}} \omega_{r} \, \mathrm{d}r + \pounds_{\xi_{k}} \omega_{s} Z_{st}^{k} - \pounds_{\xi_{k}} \pounds_{\xi_{l}} \omega_{s} \mathbb{Z}_{st}^{lk}, \quad (s,t) \in \Delta_{T}, \quad (4.1)$$

satisfies  $\omega^{\natural} \in C_{2,\text{loc}}^{\frac{p}{3}-\text{var}}([0,T]; \mathring{H}_{\mathbf{d}}^{m-4})$ . We recall that for a vector-field  $v \in H^m$ ,

$$\pounds_v \omega = \sum_{i < j} (v^q \partial_{x^q} \omega_{ij} + \omega_{qj} \partial_{x^i} v^q + \omega_{iq} \partial_{x^j} v^q) e^i \wedge e^j.$$

Moreover,

$$\omega_{st}^{\sharp} = \mathbf{d} \flat u^{P,\sharp} = \delta \omega_{st} + \pounds_{\xi_k} \omega_s Z_{st}^k = -\int_s^t \pounds_{u_r} \omega_r \, \mathrm{d}r + \pounds_{\xi_k} \pounds_{\xi_l} \omega_s \mathbb{Z}_{st}^{lk} + \omega_{st}^{\sharp}.$$

The strategy of the proof is as follows. We form a system of equations for  $\omega$  and its derivatives up to order m-1 of the form Definition A.4 and then apply Theorem A.5 and (2.4). Let  $\mathcal{I}_{m-1} = \{\emptyset\} \cup \bigcup_{n=1}^{m-1} \{1, \ldots, d\}^n$ . For  $I = \emptyset$ , let |I| = 0 and for given  $I = (i_1, \ldots, i_n) \in \mathcal{I}_{m-1}$  define |I| = n. For given  $I = (i_1, \ldots, i_n) \in \mathcal{I}_{m-1}$ , define  $\partial^I = \partial_{x^{i_1}} \circ \cdots \circ \partial_{x^{i_n}}$ . Let  $A_{d,m-1} = \bigoplus_{n=0}^{m-1} (\Lambda^2(\mathbb{R}^d)^*)^{\otimes n}$ .

For given  $n \in \{0, 1, 2, 3\}$ , let  $E_n = H^n(\mathbb{T}^d; A_{d,m-1})$ . It follows that  $(E_n)_{n=0}^3$  is a scale with a smoothing (see Section 2.1)  $J^{\eta} = P_{\leq \lfloor \eta^{-1} \rfloor}, \eta \in (0, 1)$ , in the sense of Definitions A.1 and A.3. For a given function  $\Phi \in E_n, i, j \in \{1, \ldots, d\}$  with i < j, and  $I \in \mathcal{I}_{m-1}$ , denote by  $\Phi_{ij}^I$  its (I, ij)-th component.

Define  $\Omega \in L^{\infty}([0,T]; E_{-0}) \cap C([0,T]; E_{-3})$  by  $\Omega^{I} = \partial^{I} \omega$  and  $\Omega^{\natural} \in C_{2,\text{loc}}^{\frac{p}{2}-\text{var}}([0,T]; E_{-3})$  by  $\Omega^{I} = \partial^{I} \omega^{\natural}$  for  $I \in I_{m-1}$ .

Applying the weak derivative operators  $\partial^I$  to (4.1), we find that for all  $(s,t) \in \Delta_T$ and  $\Phi \in E_3$ ,

$$\langle \Omega_{st}^{\natural}, \Phi \rangle = \langle \delta \Omega_{st}, \Phi \rangle + \langle \mu_{st}, \Phi \rangle + \langle \Omega_s, [A_{st}^{1,*} + A_{st}^{2,*}] \Phi \rangle,$$

where

- $\begin{aligned} \bullet & \langle \mu_{st}, \Phi \rangle := \sum_{I \in I_{m-1}} \int_s^t \left( (\partial^I [\pounds_{u_r} \omega_r] u_r^q \partial_{x^q} \partial^I \omega_r, \Phi^I)_{L^2} + (u_r^q \partial^I \omega_r, \partial_{x^q} \Phi^I)_{L^2} \right) \, \mathrm{d}r, \\ \bullet & A_{st}^1 \Phi := \left( \xi_k^q \partial_{x^q} \Phi + \nu_k^m \Phi \right) Z_{st}^k, \\ \bullet & A_{st}^2 \Phi := \left( \left( \xi_k^q \partial_{x^q} + \nu_k^m \right) \left( \xi_l^q \partial_{x^q} + \nu_l^m \right) \Phi \right) \mathbb{Z}_{st}^{lk}, \end{aligned}$

and where for each  $k \in \{1, \ldots, K\}, \nu_k^m : \mathbb{T}^d \to \mathcal{L}(A_{d,m-1}; A_{d,m-1})$  is implicitly and inductively defined as follows: for  $i, j \in \{1, \ldots, d\}$ ,

$$(\nu_k^m \Phi)_{ij}^{\emptyset} = \Phi_{qj}^{\emptyset} \partial_{x^i} \xi_k^q + \Phi_{iq}^{\emptyset} \partial_{x^j} \xi_k^q$$

and for  $I \in \mathcal{I}_{m-1}$  with |I| < m-1,

$$\begin{aligned} \partial_{x^{l}}[\xi_{k}^{q}\partial_{x^{q}}\Phi + \nu_{k}^{m}\Phi]_{ij}^{I} &= \xi_{k}^{q}\partial_{x^{q}}\Phi_{ij}^{(I,l)} + (\partial_{x^{l}}\xi_{k}^{j})\Phi_{ij}^{(I,j)} + ((\partial_{x^{l}}\nu_{k}^{m})\Phi)_{ij}^{I} + (\nu_{k}^{m}\partial_{x^{l}}\Phi)_{ij}^{I} \\ &= \xi_{k}^{q}\partial_{x^{q}}\Phi_{ij}^{(I,l)} + (\nu_{k}^{m}\Phi)_{ij}^{(I,l)}. \end{aligned}$$

Consequently, the components of the  $\nu_k^m$  depend only at most m derivatives of  $\xi_k$ . In order to apply Theorem A.5, we need to show that the pair  $\mathbf{A} = (A^1, A^2)$  is an unbounded rough driver (see Definition A.2) in the scale  $(E_n)$  and that we have control of the nonlinearity  $\mu$  in  $E_{-1}$  in the sense of (A.4).

Since Z satisfies Chen's relation (i.e., (2.11)), (A.2) holds for A. Moreover, for all  $\Phi \in E_3$  and  $(s,t) \in \Delta_T$ , it holds that

$$A_{st}^{1,*} \Phi = (-\xi_k \cdot \nabla \Phi + \nu_k^{m,\top} \Phi) Z_{st}^k,$$
$$|A_{st}^{1,*} \Phi|_{E_n} \lesssim_{d,m} |\xi|_{W^{m+n,\infty}} |\Phi|_{E_{n+1}} \varpi_{\mathbf{Z}}(s,t)^{\frac{1}{p}}, \quad \forall n \in \{0,2\},$$
$$|A_{st}^{2,*} \Phi|_{E_n} \lesssim_{d,m} |\xi|_{W^{m+1+n,\infty}}^2 |\Phi|_{E_{n+2}} \varpi_{\mathbf{Z}}(s,t)^{\frac{2}{p}}, \quad \forall n \in \{0,1\}.$$

Thus, **A** is an unbounded rough driver in the scale  $(E_n)$ .

We will now show that (A.4) holds for  $\mu$ . For all  $I \in \mathcal{I}_{m-1}$ , we have

$$\partial^{I}[\pounds_{u}\omega] - u^{q}\partial_{x^{q}}\partial^{I}\omega = \partial^{I}[u^{q}\partial_{x^{q}}\omega] - u^{q}\partial_{x^{q}}\partial^{I}\omega + \partial^{I}[\omega_{qj}\partial_{x^{i}}u^{q} + \omega_{iq}\partial_{x^{j}}u^{q}].$$

Applying (2.1), (2.2), and (2.4), we get

$$\sum_{I \in I_{m-1}} |\partial^{I}[\pounds_{u}\omega] - u^{q}\partial_{x^{q}}\partial^{I}\omega|_{L^{2}} \lesssim_{d,m} |\nabla u|_{L^{\infty}}|\omega|_{H^{m-1}} + |u|_{H^{m}}|\omega|_{L^{\infty}}$$
$$\lesssim_{d,m} |\nabla u|_{L^{\infty}}|\omega|_{H^{m-1}}.$$
(4.2)

Using (4.2) and Poincare's inequality, we find that there exists C = C(d, m) such that for all  $(s,t) \in \Delta_T$ ,

$$|\mu_{st}|_{E_{-1}} \le C \int_{s}^{t} |\nabla u_r|_{L^{\infty}} |\omega_r|_{H^{m-1}} \,\mathrm{d}r.$$

Therefore, in the sense of Definition A.4,  $\Omega$  is a solution of

$$\mathrm{d}\Omega + \mu(\mathrm{d}t) + \mathbf{A}(\mathrm{d}t)\Omega = 0.$$

We complete the proof by invoking Theorem A.5 and (2.4).  $\Box$ 

## 4.2. Solution estimates

We will now present the main solution estimates that we will use to prove local existence and establish the BKM blow-up criteria.

**Theorem 4.2** (Solution estimates). Let u be an  $H^m$ -solution of (RE) on the interval [0, T]. Then for all  $m' \in \mathbb{N}_0$  such that  $1 \le m' \le m-2$ , there are constants  $C_1 = C_1(d, m')$  and  $C = C(p, d, m', |\xi|_{W^{m'+2,\infty}})$  depending in an increasing way on  $|\xi|_{W^{m'+2,\infty}}$  such that

$$\sup_{t \le T} |u_t|_{H^{m'}} \le \sqrt{2} \exp\left(C\left(\int_0^T |\nabla u_r|_{L^{\infty}} \,\mathrm{d}r + \varpi_{\mathbf{Z}}(0,T)\right)\right) |u_0|_{H^{m'}}$$
(4.3)

$$\sup_{t \le T} |u_t|_{H^{m'}} \le C_1 (1 + |u_0|_{H^{m'}}) \exp\left(C(1 + \varpi_{\mathbf{Z}}(0, T)) \exp\left(C\int_0^T |\omega_r|_{L^{\infty}} \,\mathrm{d}r\right)\right).$$
(4.4)

If  $m_* \leq m' \leq m-2$ , then for all  $T_* \in [0,T]$  satisfying

$$e^{C(1+\varpi_{\mathbf{Z}}(0,T_*))}T_* < |u_0|_{H^{m'}}^{-1},$$

it holds that

$$\sup_{t \le T_*} |u_t|_{H^{m'}} \le \frac{e^{C(1+\varpi_{\mathbf{Z}}(0,T_*))}}{|u_0|_{H^{m'}}^{-1} - e^{C(1+\varpi_{\mathbf{Z}}(0,T_*))}T_*}.$$
(4.5)

**Proof.** Using the notation of the proof of Theorem 4.1, we let

$$\Omega = (\partial^{I}\omega)_{I \in I_{m'-1}} \in L^{\infty}([0,T]; H^{2}(\mathbb{T}^{d}; A_{d,m'-1}) \cap C([0,T]; H^{-1}(\mathbb{T}^{d}; A_{d,m'-1})),$$
  
$$\Omega^{\natural} = (\partial^{I}\omega^{\natural})_{I \in I_{m'-1}} \in C_{2,\text{loc}}^{\frac{p}{3}-\text{var}}([0,T]; H^{-1}(\mathbb{T}^{d}; A_{d,m'-1})),$$

and

$$\Omega^{\sharp} = (\partial^{I} \omega^{\sharp})_{I \in I_{m'-1}} \in C^{\frac{p}{2} - \operatorname{var}}_{2, \operatorname{loc}}([0, T]; L^{2}(\mathbb{T}^{d}; A_{d, m'-1})).$$

Here, we have used the worst possible regularity at m' = m - 2.

For given  $n \in \{0, 1, 2, 3\}$ , let  $E_n = W^{n,\infty}(\mathbb{T}^d; A_{d,m'-1} \otimes A_{d,m'-1})$ . Let  $\rho : \mathbb{R}^d \to \mathbb{R}$ denote a non-negative radially symmetric function with support in the unit ball that integrates to one. For  $\eta \in (0,1)$ , let  $\rho_{\eta} = \eta^{-d}\rho(\frac{\cdot}{\eta})$  and define the smoothing operator  $J^{\eta} : E_0 \to C^{\infty}(\mathbb{T}^d; A_{d,m'-1} \otimes A_{d,m'-1})$  by  $J^{\eta}\Phi = \Phi * \rho_{\eta} = \int_{\mathbb{R}^d} \rho_{\eta}(y)\Phi(\cdot - y)dV$ . It follows that  $(E_n)_{n=0}^3$  is a scale with smoothing (see Section 2.1)  $J^{\eta} = \rho_{\eta} *, \eta \in (0,1)$ , in the sense of Definitions A.1 and A.3. For  $\Phi \in E_3$  and  $k \in \{1, \ldots, K\}$ , denote

$$L_{\xi_k}\Phi = \xi_k^q \partial_{x^q}\Phi + \nu_k^{m'}\Phi,$$

where  $\nu_k^{m'}$  is defined as in the proof of Theorem 4.1.

In order to obtain the desired estimates, we will proceed as follows. First, we will write down an unbounded rough driver equation (Definition A.4) for the bounded path  $\Omega^{\otimes 2} = \Omega \otimes \Omega : [0,T] \to E_{-0}$ . Second, we will apply Theorem A.5 to obtain a bound on the associated remainder term  $\Omega^{\otimes 2,\natural}$ . In the final step, we will test against a  $\Phi = \mathbf{I}$  such that  $\langle \Omega \otimes \Omega, \mathbf{I} \rangle = |\omega|^2_{H^{m'-1}}$  and apply rough Gronwall's lemma (i.e., Lemma B.1) to obtain two solution estimates; one of which is used for local existence and other of which is used to derive the BKM blow-up criterion.

For arbitrarily given  $M_1, M_2 \in A_{d,m'-1}$ , denote  $M_1 \hat{\otimes} M_2 = 2^{-1} (M_1 \otimes M_2 + M_2 \otimes M_1)$ . Note that since **Z** is geometric, we have that for all  $k, l \in \{1, \ldots, K\}$ ,

$$L_{\xi_k}\Omega_s \otimes L_{\xi_l}\Omega_s Z_{st}^k Z_{st}^l = L_{\xi_k}\Omega_s \otimes L_{\xi_l}\Omega_s (\mathbb{Z}_{st}^{lk} + \mathbb{Z}_{st}^{kl}) = 2L_{\xi_k}\Omega_s \hat{\otimes} L_{\xi_l}\Omega_s \mathbb{Z}_{st}^{lk}.$$
(4.6)

We will need the following fact. The pointwise tensor product of smooth functions extends canonically to a continuous bi-linear map

$$\otimes : H^1(\mathbb{T}^d; A_{d,m'-1}) \times H^{-1}(\mathbb{T}^d; A_{d,m'-1}) \to E_{-1}.$$

Moreover, for all  $f \in H^1(\mathbb{T}^d; A_{d,m'-1}), g \in H^{-1}(\mathbb{T}^d; A_{d,m'-1})$ , and  $\Phi \in E_1$ ,

$$\langle f \otimes g, \Phi \rangle := \langle g, (f, \Phi)_{A_{d,m'-1}} \rangle. \tag{4.7}$$

Applying (4.7) and (4.6), we find that for all  $\Phi \in E_1$  and  $(s,t) \in \Delta_T$ ,

$$\begin{split} \langle \delta\Omega_{st}^{\otimes 2}, \Phi \rangle &= 2 \langle \Omega_s \hat{\otimes} \delta\Omega_{st}, \Phi \rangle + \langle \delta\Omega_{st} \otimes \delta\Omega_{st}, \Phi \rangle \\ &= - \langle \mathcal{M}_{st}, \Phi \rangle - 2 \langle \Omega_s \hat{\otimes} L_{\xi_k} \Omega_s, \Phi \rangle Z_{st}^k \\ &+ 2 \langle \Omega_s \hat{\otimes} L_{\xi_k} L_{\xi_l} \Omega_s + L_{\xi_k} \omega_s \hat{\otimes} L_{\xi_l} \omega_s, \Phi \rangle Z_{st}^{lk} + \langle \Omega_{st}^{\otimes 2, \natural}, \Phi \rangle \\ &= - \langle \mathcal{M}_{st}, \Phi \rangle - \langle \Omega_s^{\otimes 2}, [\Gamma_{st}^{1,*} + \Gamma_{st}^{2,*}] \Phi \rangle + \langle \Omega_{st}^{\otimes 2, \natural}, \Phi \rangle, \end{split}$$
(4.8)

where

$$\langle \mathcal{M}_{st}, \Phi \rangle := \sum_{I,J \in \mathcal{I}_{m'-1}} \int_{s}^{t} \langle \partial^{I} [\pounds_{u_{r}} \omega_{r}] \otimes \partial^{J} \omega_{r} + \partial^{I} \omega_{r} \otimes \partial^{J} [\pounds_{u_{r}} \omega_{r}], \Phi^{IJ} \rangle \, \mathrm{d}r,$$

$$\Gamma^{1} \Phi := (\xi_{k}^{q} \partial_{x^{q}} \Phi + 2(\nu_{k}^{m'} \hat{\otimes} \operatorname{id}) \Phi) Z_{st}^{k},$$

$$\begin{split} \Gamma^2 \Phi &:= -(\xi_k^q \partial_{x^q} + 2\nu_k^{m'} \hat{\otimes} \operatorname{id})(\xi_l^q \partial_{x^q} + 2\nu_l^{m'} \hat{\otimes} \operatorname{id}) \Phi \mathbb{Z}_{st}^{lk}, \\ \langle \Omega_{st}^{\otimes 2, \natural}, \Phi \rangle &:= 2 \langle \Omega_s \hat{\otimes} \Omega_{st}^{\natural}, \Phi \rangle \\ &+ \sum_{I, J \in I_{m'-1}} \int_s^t \langle \partial^I [\pounds_{u_r} \omega_r] \otimes \partial^J \delta \omega_{rs} + \partial^I \delta \omega_{rs} \otimes \partial^J [\pounds_{u_r} \omega_r], \Phi^{IJ} \rangle \, \mathrm{d}r \\ &+ \langle \Omega_{st}^{\sharp} \otimes \delta \Omega_{st}, \Phi \rangle + \langle L_{\xi_k} \Omega_s Z_{st}^k \otimes \Omega_{st}^{\sharp}, \Phi \rangle. \end{split}$$

Here,  $2\nu_k^{m'}\hat{\otimes} \operatorname{id} : \mathbb{T}^d \to \mathcal{L}(A_{d,m'-1} \otimes A_{d,m'-1}; A_{d,m'-1} \otimes A_{d,m'-1})$  is the unique linear map induced by the bi-linear map  $2\nu_k^{m'}(x)\hat{\otimes} \operatorname{id} : A_{d,m'-1} \times A_{d,m'-1} \to A_{d,m'-1} \otimes A_{d,m'-1}, x \in \mathbb{T}^d$ , defined for given  $(M_1, M_2) \in A_{d,m'-1} \times A_{d,m'-1}$  by

$$(2\nu_k^{m'}(x)\hat{\otimes} \operatorname{id})(M_1, M_2) = \nu_k^{m'}(x)M_1 \otimes M_2 + M_1 \otimes \nu_k^{m'}(x)M_2.$$

We will now obtain control over the drift. Proceeding as in (4.2), we find that

$$\sum_{I \in I_{m'-1}} |\partial^I[\pounds_u \omega] - u^q \partial_{x^q} \partial^I \omega|_{L^2} \lesssim_{d,m} |\nabla u|_{L^\infty} |\omega|_{H^{m'-1}}.$$
(4.9)

Using (4.9) and Poincare's inequality, we find that there exists a constant C = C(d, m) such that for all  $\Phi \in E_1$  and  $(s, t) \in \Delta_T$ ,

$$\langle \mathcal{M}_{st}, \Phi \rangle = \sum_{I,J \in I_{m'-1}} \int_{s}^{t} \langle \left( \partial^{I} [\pounds_{u_{r}} \omega_{r}] - u^{q} \partial_{x^{q}} \partial^{I} \omega_{r} \right) \otimes \partial^{J} \omega_{r}, \Phi^{IJ} \rangle \, \mathrm{d}r$$

$$+ \sum_{I,J \in I_{m'-1}} \int_{s}^{t} \langle \partial^{I} \omega_{r} \otimes \left( \partial^{J} [\pounds_{u_{r}} \omega_{r}] - u^{q}_{r} \partial_{x^{q}} \partial^{J} \omega_{r} \right), \Phi^{IJ} \rangle \, \mathrm{d}r$$

$$- \sum_{I,J \in I_{m'-1}} \int_{s}^{t} \langle u^{q}_{r} (\partial^{I} \omega_{r} \otimes \partial^{J} \omega_{r}), \partial_{x^{q}} \Phi^{IJ} \rangle \, \mathrm{d}r$$

$$\leq C |\Phi|_{E_{1}} \int_{s}^{t} |\nabla u_{r}|_{L^{\infty}} |\omega_{r}|_{H^{m'-1}}^{2} \, \mathrm{d}r.$$

**Claim** (Verification of 'squared' remainder). We have  $\Omega^{\otimes 2,\natural} \in C_{2,\text{loc}}^{\frac{p}{3}-\text{var}}([0,T]; E_{-1}).$ 

**Proof.** We will proceed by estimating term-by-term. By the embeddings  $H^2 \otimes H^1 \hookrightarrow E_{-1}$ and  $L^1 \otimes L^1 \hookrightarrow L^2$ , we find that there is a constant  $C = C(d, m', |\xi|_{W^{m',\infty}})$  such that for all  $(s,t) \in \Delta_T$ ,

$$|\Omega_s \hat{\otimes} \Omega_{st}^{\natural}|_{E_{-1}} \leq \sup_{r \leq T} |\Omega_r^{m'}|_{H^2} |\Omega_{st}^{\natural}|_{\frac{p}{3}-\operatorname{var};[s,t];H^{-1}}^{\frac{p}{3}},$$

$$\begin{aligned} |\Omega_{st}^{\sharp} \otimes \delta\Omega_{st}|_{L^{1}} &\leq |\Omega_{st}^{\sharp}|_{\frac{p}{2}-\operatorname{var};[s,t];L^{2}} |\Omega_{st}|_{p-\operatorname{var};[s,t];L^{2}}^{p}, \\ |\pounds_{\xi_{k}}\Omega_{s}Z_{st}^{k} \otimes \Omega_{st}^{\sharp}|_{L^{1}} &\leq C \sup_{r \leq T} |\Omega_{r}|_{H^{1}} \varpi_{\mathbf{Z}}(s,t)^{\frac{1}{p}} |\Omega_{st}^{\sharp}|_{\frac{p}{2}-\operatorname{var};[s,t];L^{2}}^{\frac{p}{2}}. \end{aligned}$$

Upon using the decomposition

$$\begin{split} \langle \partial^{I}[\pounds_{u_{r}}\omega_{r}] \otimes \partial^{J}\delta\omega_{rs}, \Phi^{IJ} \rangle &= \langle \left( \partial^{I}[\pounds_{u_{r}}\omega_{r}] - u_{r}^{q}\partial_{x^{q}}\partial^{I}\omega_{r} \right) \otimes \partial^{J}\delta\omega_{rs}, \Phi^{IJ} \rangle \\ &- \langle u_{r}^{q}\partial^{I}\omega_{r} \otimes \partial_{x^{q}}\partial^{J}\delta\omega_{rs}, \Phi^{IJ} \rangle \\ &- \langle u_{r}^{q}\partial^{I}\omega_{r} \otimes \partial^{J}\delta\omega_{rs}, \partial_{x^{q}}\Phi^{IJ} \rangle \end{split}$$

in (4.9) and applying Poincare's inequality, we find that for all  $\Phi \in E_1$  and  $(s,t) \in \Delta_T$ ,

$$\sum_{I,J\in I_{m'-1}} \int_{s}^{t} \langle \partial^{I}[\pounds_{u_{r}}\omega_{r}] \otimes \partial^{J}\delta\omega_{rs} + \partial^{I}\delta\omega_{rs} \otimes \partial^{J}[\pounds_{u_{r}}\omega_{r}], \Phi^{IJ} \rangle \,\mathrm{d}r$$
$$\lesssim_{d,m} |\Phi|_{E_{1}}|\omega|_{p-\mathrm{var};[s,t];H^{m'}} \int_{s}^{t} |\nabla u_{r}|_{L^{\infty}}|\omega_{r}|_{H^{m'-1}} \,\mathrm{d}r.$$

We then complete the proof by using the property that a product of powers of regular controls whose powers sum to greater than or equal to one is a control (see, e.g., [31, Ex. 1.9].  $\Box$ 

One can easily check that the pair  $\mathbf{\Gamma} = (\Gamma^1, \Gamma^2)$  is an unbounded rough driver in the scale  $(E_n)$  and that (A.1) holds with  $\varpi_{\mathbf{\Gamma}}(s,t) := C \varpi_{\mathbf{Z}}(s,t)$ ,  $(s,t) \in \Delta_T$ , for a given constant  $C = C(d, m, |\xi|_{W^{m'+2,\infty}})$  that depends in an increasing way on  $|\xi|_{W^{m'+2}}$ . Therefore,  $\Omega^{\otimes 2}$  is a solution of

$$\mathrm{d}\Omega^{\otimes 2} + \mathcal{M}(\mathrm{d}t) + \Gamma(\mathrm{d}t)\Omega^{\otimes 2} = 0$$

in the sense of Definition A.4. Thus, by Theorem A.5, there is a constant  $C = C(p, d, m', |\xi|_{W^{m'+2,\infty}})$  depending in an increasing way on  $|\xi|_{W^{m'+2,\infty}}$  and a constant L = L(p) such that for all  $(s, t) \in \Delta_T$  with  $\varpi_{\Gamma}(s, t) \leq L$  it holds that

$$|\Omega^{\otimes 2,\natural}|_{\frac{p}{3}-\operatorname{var};[s,t];E_{-3}}^{\frac{p}{3}} \leq C \left( \sup_{s \leq r \leq t} |\omega_{r}|_{H^{m'-1}}^{2} \varpi_{\mathbf{Z}}(s,t)^{\frac{3}{p}} + \int_{s}^{t} |\nabla u_{r}|_{L^{\infty}} |\omega_{r}|_{H^{m'-1}}^{2} \operatorname{d} r \varpi_{\mathbf{Z}}(s,t)^{\frac{1}{p}} \right). \quad (4.10)$$

Let  $I \in A_{d,m'-1} \otimes A_{d,m'-1}$  be such that  $\langle \Omega \otimes \Omega, I \rangle = |\omega|^2_{H^{m'-1}}$ . Henceforth, let  $C = C(p, d, m', |\xi|_{W^{m'+2,\infty}})$  denote a constant which increases with  $|\xi|_{W^{m'+2,\infty}}$  and, thus, may change from line to line. Letting  $\Phi = I$  in (4.8), we find that for all  $(s, t) \in \Delta_T$ ,

$$\delta(|\omega|_{H^{m'-1}}^2)_{st} = -2\sum_{I\in I_{m'-1}}\int_s^t \langle \partial^I[\pounds_{u_r}\omega_r], \partial^I\omega_r\rangle \,\mathrm{d}r + \langle \Omega_s^{\otimes 2}, [\Gamma_{st}^{1,*} + \Gamma_{st}^{2,*}]\boldsymbol{I}\rangle + \langle \Omega_{st}^{\otimes 2,\natural}, \boldsymbol{I}\rangle.$$

It follows from (4.10) and the fact that  $\Gamma$  is an unbounded rough driver that for all  $(s,t) \in \Delta_T$  with  $\varpi_{\Gamma}(s,t) \leq L \wedge 1$ ,

$$\delta(|\omega|_{H^{m'-1}}^2)_{st} \le C\left(\int_{s}^{t} |\nabla u_r|_{L^{\infty}} |\omega_r|_{H^{m'-1}}^2 \,\mathrm{d}r + \sup_{s \le r \le t} |\omega_r|_{H^{m'-1}}^2 \varpi_{\mathbf{Z}}(s,t)^{\frac{1}{p}}\right).$$
(4.11)

Upon applying rough Gronwall's lemma (i.e., Lemma B.1) and (2.4), we obtain (4.3). For  $t \in [0,T]$ , define  $y_t = \ln(e + |u_t|_{H^{m'}})$ . By virtue of the inequality (2.5) and (4.3), we have that for all  $t \in [0,T]$ ,

$$y_t \le \ln(\sqrt{2}y_0) + C\varpi_{\mathbf{Z}}(0,t) + C\int_0^t |\omega_r|_{L^{\infty}} y_r \,\mathrm{d}r.$$

By using Gronwall's inequality, we find that for all  $t \in [0, T]$ ,

$$y_t \leq \left(\ln(\sqrt{2}y_0) + C\varpi_{\mathbf{Z}}(0,t)\right) \exp\left(C\int_0^t |\omega_r|_{L^{\infty}} \mathrm{d}r\right),$$

and hence (4.4) holds.

Returning to (4.11) and applying the Sobolev embedding and (2.4) with  $m' > m_*$ , we find that for all  $(s,t) \in \Delta_T$  with  $\varpi_{\Gamma}(s,t) \leq L \wedge 1$ ,

$$\delta(|\omega|_{H^{m'-1}}^2)_{st} \le C\left(\int_{s}^{t} |\omega_r|_{H^{m'-1}}^3 \,\mathrm{d}r + \sup_{s \le r \le t} |\omega_r|_{H^{m'-1}}^2 \varpi_{\mathbf{Z}}(s,t)^{\frac{1}{p}}\right).$$

Applying rough Gronwall's lemma implies that, for all  $t \in [0, T]$  and t' > t,

$$|\omega_t|^2_{H^{m'-1}} \le e^{C(1+\varpi_{\mathbf{Z}}(0,t'))} \left( |\omega_0|^2_{H^{m'-1}} + \int_0^t |\omega_r|^3_{H^{m'-1}} \,\mathrm{d}r \right) =: y_t.$$

It follows that for all  $t \in [0, T]$  and t' > t, one has

$$\frac{\mathrm{d}}{\mathrm{d}t}y_t^{-\frac{1}{2}} = -2^{-1}e^{C(1+\varpi_{\mathbf{Z}}(0,t'))} \left(y_t^{-\frac{1}{2}}|\omega_r|_{H^{m'-1}}\right)^3 \ge -2^{-1}e^{C(1+\varpi_{\mathbf{Z}}(0,t'))}$$

which implies

$$y_t^{-\frac{1}{2}} \ge \exp^{-\frac{C}{2}(1+\varpi \mathbf{z}(0,t'))} |\omega_0|_{H^{m'-1}}^{-1} - 2^{-1} e^{C(1+\varpi \mathbf{z}(0,t'))} t.$$

Therefore, upon letting  $t' \downarrow t$  and applying (2.4), we obtain (4.5).  $\Box$ 

**Remark 4.3.** In [9], the method of doubling of variables was used to derive solution estimates and prove uniqueness for linear rough transport equations. The remainder term for linear rough transport equations can only be expected to belong to  $H^{-3}$  if its initial condition belongs to  $L^2$ . Thus, more care is needed to derive solution estimates and uniqueness since it is not possible, at least directly, to test the equation against the solution. In our setting, we can avoid doubling variables and obtain solution estimates by smoothing (e.g., the path an initial condition and  $\xi$ ) at the expense of having to assume  $\xi$  is slightly more regular whenever continuity in time in the highest norm is needed. In particular, we prove uniqueness by considering differences only in the  $L^2$ -norm and not the highest norm.

# 4.3. Difference estimates and uniqueness

The following theorem establishes a priori estimates of the difference of  $H^m$ -solutions in the  $H^1$ -norm and  $H^m$ -norm.

**Theorem 4.4** (Difference estimates and uniqueness). Let  $u^1$  and  $u^2$  be  $H^m$ -solutions of (RE) on the interval [0,T] with the same data  $(\mathbf{Z},\xi) \in C_g^{p-\text{var}}(\mathbb{R}_+;\mathbb{R}^K) \times (W^{m+2,\infty})^K$  starting from  $u_0^1$  and  $u_0^2$ , respectively. Then there is a constant  $C = C(p,d,|\xi|_{W^{3,\infty}})$  such that

$$\sup_{t \le T} |u_t^1 - u_t^2|_{H^1} \le \sqrt{2} \exp\left(C\left(\int_0^T \left(|u_r^1|_{H^{m_*}} + |u_r^2|_{H^{m_*}}\right) \,\mathrm{d}r + \varpi_{\mathbf{Z}}(0,T)\right)\right) |u_0^1 - u_0^2|_{H^1}.$$
(4.12)

If  $m_* \leq m' \leq m-2$ , then there is a constant  $C = C(p, d, m', |\xi|_{W^{m'+2,\infty}})$  such that

$$\sup_{t \le T} |u_t^1 - u_t^2|_{H^{m'}} \le \sqrt{q_{m'}(T)} |u_0^1 - u_0^2|_{H^{m'}} + Tq_{m'}(T) |u_0^1 - u_0^2|_{H^{m_*-1}} |u_0^1|_{H^{m'+1}}, \quad (4.13)$$

where

$$q_{m'}(T) := \sqrt{2} \exp\left(C\left(\int_{0}^{T} \left(|u_{r}^{1}|_{H^{m_{*}}} + |u_{r}^{2}|_{H^{m'}}\right) \,\mathrm{d}r + \varpi_{\mathbf{Z}}(0,T)\right)\right).$$

**Proof.** Let  $u = u^1 - u^2$  and  $\omega = \omega^1 - \omega^2$ . We will adopt the notation of the proof of Theorem 4.2 and consider  $m' \in \{1, m_* - 1\} \cup \{m_*, m_* + 1, \ldots\}$ . For  $n \in \{0, 1, 2, 3\}$ , let  $E_n = W^{n,\infty}(\mathbb{T}^d; A_{d,m'-1} \otimes A_{d,m'-1})$ . By the analysis in the proof of Theorem 4.2, we have that

$$\Omega^{\otimes 2} := \Omega^{1, \otimes 2} - \Omega^{2, \otimes 2} \in L^{\infty}([0, T]; E_{-0}) \cap C^{p - \operatorname{var}}([0, T]; E_{-3})$$

and

$$\Omega^{\otimes 2,\natural} = \Omega^{1,\otimes 2,\natural} - \Omega^{2,\otimes 2,\natural} \in C_{2,\mathrm{loc}}^{\frac{p}{2}-\mathrm{var}}([0,T]; E_{-1})$$

satisfy for all  $\Phi \in E_1$  and  $(s,t) \in \Delta_T$ ,

$$\langle \Omega_{st}^{\otimes 2,\natural}, \Phi \rangle = \langle \delta \Omega_{st}^{\otimes 2}, \Phi \rangle + \langle \mathcal{N}_{st}, \Phi \rangle + \langle \Omega_{s}^{\otimes 2}, [\Gamma_{st}^{1,*} + \Gamma_{st}^{2,*}] \Phi \rangle,$$

where

$$\begin{split} \langle \mathcal{N}_{st}, \Phi \rangle &\coloneqq \sum_{I,J \in I_{m'-1}} \int_{s}^{t} \langle \partial^{I} [\pounds_{u_{r}} \omega_{r}^{1}] \otimes \partial^{J} \omega_{r} + \partial^{I} \omega_{r} \otimes \partial^{J} [\pounds_{u_{r}} \omega_{r}^{1}], \Phi^{IJ} \rangle \, \mathrm{d}r \\ &+ \sum_{I,J \in I_{m'-1}} \int_{s}^{t} \langle \left( \partial^{I} [\pounds_{u_{r}^{2}} \omega_{r}] - u_{r}^{2,q} \partial_{x^{q}} \partial^{I} \omega_{r} \right) \otimes \partial^{J} \omega_{r}, \Phi^{IJ} \rangle \, \mathrm{d}r \\ &+ \sum_{I,J \in I_{m'-1}} \int_{s}^{t} \langle \partial^{I} \omega_{r} \otimes \left( \partial^{J} [\pounds_{u_{r}^{2}} \omega_{r}] - u_{r}^{2,q} \partial_{x^{q}} \partial^{J} \omega_{r} \right), \Phi^{IJ} \rangle \, \mathrm{d}r \\ &- \sum_{I,J \in I_{m'-1}} \int_{s}^{t} \langle u_{r}^{2,q} \partial^{I} \omega_{r} \otimes \partial^{J} \omega_{r}, \partial_{x^{q}} \Phi^{IJ} \rangle \, \mathrm{d}r. \end{split}$$

For all  $f \in H^1$  and  $g \in H^s$  with  $s > \frac{d}{2} - 1$ , the Sobolev embedding (see, e.g., [12]) implies that

$$|fg|_{L^2} \lesssim_{d,s} |f|_{H^1} |g|_{H^s}. \tag{4.14}$$

If m' = 1, then using (4.14) and (2.1), we find

$$\begin{split} \|\pounds_{u}\omega^{1}\|_{L^{2}} \lesssim_{d} |u|_{H^{1}}|\omega^{1}|_{H^{m_{*}-1}} + |\nabla u|_{L^{2}}|\omega^{1}|_{L^{2}} \lesssim_{d} |\omega|_{L^{2}}|u^{1}|_{H^{m_{*}}} \\ \|\pounds_{u^{2}}\omega - u^{2} \cdot \nabla \omega|_{L^{2}} \lesssim_{d} |\nabla u^{2}|_{L^{\infty}}|\omega|_{L^{2}} \lesssim_{d} |\omega|_{L^{2}}|u^{2}|_{H^{m_{*}}}. \end{split}$$

Making use of (2.1) and (2.4), we obtain

$$\begin{split} \sum_{I \in I_{m'-1}} |\partial^{I}[\pounds_{u}\omega^{1}]|_{L^{2}} \lesssim_{d,m'} \sum_{l=0}^{m'-1} \left( ||D^{l}u||D^{m'-l}\omega^{1}||_{L^{2}} + ||D^{l+1}u||D^{m'-l-l}\omega^{1}||_{L^{2}} \right) \\ \lesssim_{d,m'} |u|_{L^{\infty}} |\omega^{1}|_{H^{m'}} + |\omega^{1}|_{L^{\infty}} |u|_{m'} \\ \lesssim_{d,m'} \begin{cases} |\omega|_{H^{m'-1}} |u^{1}|_{H^{m_{*}}} & \text{if } m' = m_{*} - 1, \\ |\omega|_{H^{m_{*}-2}} |u^{1}|_{H^{m'+1}} + |\omega|_{H^{m'-1}} |u^{1}|_{H^{m_{*}}} & \text{if } m' \ge m_{*}, \end{cases}$$

and

$$\begin{split} \sum_{I \in I_{m'-1}} &|\partial^{I}[\pounds_{u^{2}}\omega] - u^{2,q}\partial_{x^{q}}\partial^{I}\omega|_{L^{2}} \\ \lesssim_{d,m'} \sum_{l=1}^{m'-1} &||D^{l}u^{2}||D^{m'-l}\omega||_{L^{2}} + \sum_{l=0}^{m'-1} &||D^{l+1}u^{2}||D^{m'-l-l}\omega||_{L^{2}} \\ \lesssim_{d,m'} \begin{cases} \sum_{l=0}^{m'-1} &||D^{l}\nabla u^{2}||D^{m'-l}u||_{L^{2}} & \text{if } m' = m_{*} - 1, \\ \sum_{l=0}^{m'-1} &||D^{l}\nabla u^{2}||D^{m'-l-l}\omega||_{L^{2}} & \text{if } m' \ge m_{*}, \end{cases} \\ \lesssim_{d,m'} \begin{cases} |\nabla u^{2}|_{L^{\infty}}|u|_{H^{m'}} + |u|_{L^{\infty}}|u^{2}|_{H^{m_{*}}} & \text{if } m' = m_{*} - 1, \\ |\nabla u^{2}|_{L^{\infty}}|\omega|_{H^{m'-1}} + |\omega|_{L^{\infty}}|u^{2}|_{H^{m_{*}}} & \text{if } m' \ge m_{*}, \end{cases} \\ \lesssim_{d,m'} &|\omega|_{H^{m'-1}}|u^{2}|_{H^{m_{*}}}. \end{split}$$

Putting it all together and using  $|u^2|_{L^{\infty}} \leq_d |u^2|_{H^{m_*}}$ , we get that there exists a constant C = C(d, m') such that for all  $(s, t) \in \Delta_T$ ,

$$\begin{split} |\mathcal{N}_{st}^{m'}|_{E_{-1}} &\leq C \begin{cases} \int_{s}^{t} |\omega_{r}|_{L^{2}}^{2} \left(|u_{r}^{1}|_{H^{m*}} + |u_{r}^{2}|_{H^{m*}}\right) \mathrm{d}r & \text{if } m' = 1, \\ \int_{s}^{t} |\omega_{r}|_{H^{m'-2}}^{2} \left(|u_{r}^{1}|_{H^{m*}} + |u_{r}^{2}|_{H^{m*}}\right) \mathrm{d}r & \text{if } m' = m_{*} - 1, \\ \int_{s}^{t} |\omega_{r}|_{H^{m'-1}} |\omega_{r}|_{H^{m*-2}} |u_{r}^{1}|_{H^{m'+1}} & \text{if } m' \geq m_{*}, \\ &+ |\omega_{r}|_{H^{m'-1}}^{2} \left(|u_{r}^{1}|_{H^{m*}} + |u_{r}^{2}|_{H^{m'}}\right) \mathrm{d}r & \\ &=: \varpi_{\mathcal{N}^{m'}}(s, t). \end{split}$$

Therefore,  $\Omega^{\otimes 2}$  is a solution of

$$\mathrm{d}\Omega^{\otimes 2} + \mathcal{N}(\mathrm{d}t) + \Gamma(\mathrm{d}t)\Omega^{\otimes 2} = 0$$

in the sense of Definition A.4. Owing to Theorem A.5, there is a constant  $C = C(p, d, m, |\xi|_{W^{m'+2,\infty}})$  depending in an increasing way on  $|\xi|_{W^{m'+2,\infty}}$  and a constant L = L(p) such that for all  $(s, t) \in \Delta_T$  with  $\varpi_{\Gamma}(s, t) \leq L$  it holds that

$$|\Omega^{\otimes 2,\natural}|_{\frac{p}{3}-\operatorname{var};[s,t];E_{-3}}^{\frac{p}{3}} \leq C\left(\sup_{s\leq r\leq t}|\omega_{r}|_{H^{m'-1}}^{2}\varpi_{\mathbf{Z}}(s,t)^{\frac{3}{p}} + \varpi_{\mathcal{N}^{m'}}(s,t)\varpi_{\mathbf{Z}}(s,t)^{\frac{1}{p}}\right).$$

Henceforth, let  $C = C(p, d, m, |\xi|_{W^{m'+2,\infty}})$  denote a constant depending in an increasing way on  $|\xi|_{W^{m'+2,\infty}}$  that may change from line to line. Letting  $\Phi = \mathbf{I}$  in (4.8) as in the proof of Theorem 4.2, we get that for all  $(s, t) \in \Delta_T$ ,

$$\delta(|\omega|_{H^{m'-1}}^2)_{st} \le -2\sum_{I\in\mathcal{I}_{m'-1}}\int_s^t \langle \partial^I[\pounds_{u_r}\omega_r^1 + \pounds_{u_r^2}\omega_r], \partial^I\omega_r\rangle \,\mathrm{d}r$$

$$+ \langle \Omega_s^{\otimes 2}, [\Gamma_{st}^{1,*} + \Gamma_{st}^{2,*}] \boldsymbol{I} \rangle + \langle \Omega_{st}^{\otimes 2,\natural}, \boldsymbol{I} \rangle.$$

Thus, if  $m' \in \{1, m_* - 1\}$ , then for all  $(s, t) \in \Delta_T$  with  $\varpi_{\Gamma}(s, t) \leq L \wedge 1$ , we have

$$\delta(|\omega|_{H^{m'-1}}^2)_{st} \le C \int_{s}^{t} |\omega_r|_{H^{m'-1-2}}^2 \left( |u_r^1|_{H^{m_*}} + |u_r^2|_{H^{m_*}} \right) \, \mathrm{d}r + C \sup_{s \le r \le t} |\omega_r|_{H^{m'-1}}^2 \varpi_{\mathbf{Z}}(s,t)^{\frac{1}{p}}.$$

Applying Lemma B.1, we get

$$\sup_{t \le T} |\omega_t^1 - \omega_t^2|_{H^{m'-1}}$$
  
$$\le \sqrt{2} \exp\left(C\left(\int_0^T \left(|u_r^1|_{H^{m_*}} + |u_r^2|_{H^{m_*}}\right) \, \mathrm{d}r + \varpi_{\mathbf{Z}}(0,T)\right)\right) |\omega_0^1 - \omega_0^2|_{H^{m'-1}}, \quad (4.15)$$

and hence (4.12) follows from (2.4).

If  $m' \ge m_*$ , then for all  $(s,t) \in \Delta_T$  with  $\varpi_{\Gamma}(s,t) \le L \wedge 1$ ,

$$\delta(|\omega|_{H^{m'-1}}^2)_{st} \le C \int_{s}^{t} \left( |\omega_r|_{H^{m'-1}} |\omega_r|_{H^{m_*-2}} |u_r^1|_{H^{m'+1}} + |\omega_r|_{H^{m'-1}}^2 \left( |u_r^1|_{H^{m_*}} + |u_r^2|_{H^{m'}} \right) \right) \, \mathrm{d}r$$
$$+ C \sup_{s \le r \le t} |\omega_r|_{H^{m'-1}}^2 \varpi_{\mathbf{Z}}(s,t)^{\frac{1}{p}}.$$

Using Lemma B.1, (4.3), and (4.15), we obtain that for all  $t \in [0, T]$  and t' > t,

$$|\omega_t|_{H^{m'-1}}^2 \le q_{m'}(t') \left( |\omega_0|_{H^{m'-1}}^2 + |\omega_0|_{H^{m_*-2}} |\omega_0^1|_{H^{m_*-1}} \int_0^t |\omega_r|_{H^{m'-1}} \,\mathrm{d}r \right),$$

where

$$q_{m'}(t') := \sqrt{2} \exp\left(C\left(\int_{0}^{t'} \left(|u_{r}^{1}|_{H^{m_{*}}} + |u_{r}^{2}|_{H^{m'}}\right) \, \mathrm{d}r + \varpi_{\mathbf{Z}}(0, t')\right)\right).$$

Thus, letting  $t' \downarrow t$ , we find that for all  $t \in [0, T]$ ,

$$|\omega_t^1 - \omega_t^2|_{H^{m'}} \le \sqrt{q_{m'}(t)} |\omega_0^1 - \omega_0^2|_{H^{m'}} + tq_{m'}(t) |\omega_0^1 - \omega_0^2|_{H^{m_*-2}} |\omega_0^1|_{H^{m'+1}}$$

which yields (4.13).  $\Box$ 

# 5. Local well-posedness (Proof of Theorem 3.7)

**Proof of Theorem 3.7.** There exists a sequence  $\{(u_0^n, \mathbf{Z}^n, \xi^n)\}_{n=1}^{\infty} \in \mathring{C}_{\sigma}^{\infty} \times C_g^{1-\operatorname{var}} \times (C_{\sigma}^{\infty})^K$  such that for all  $n \in \mathbb{N}, \mathbf{Z}^n = (\delta z^n, \mathbb{Z}^n)$  is the canonical lift of a  $z \in C^{\infty}(\mathbb{R}_+; \mathbb{R}^K), \{(u_0^n, \mathbf{Z}^n, \xi^n)\}_{n=1}^{\infty}$  converges to  $(u_0, \mathbf{Z}, \xi)$  in  $H_{\sigma}^m \times C_g^{p-\operatorname{var}} \times W^{m+1,\infty}$ , and for all  $n \in \mathbb{N}$ ,

$$|u_0^n|_{H^m} \le |u_0|_{H^m}, \quad |\xi^n|_{W^{m+2,\infty}} \le |\xi|_{W^{m+2,\infty}}, \varpi_{\mathbf{Z}^n}(s,t) \le 1 + \varpi_{\mathbf{Z}}(s,t), \ \forall (s,t) \in \Delta_{[0,\infty)}.$$
(5.1)

For all  $n \in \mathbb{N}$ , there exists a maximally extended solution  $u^n \in C^1([0, T^n_{\max}); \mathring{C}^{\infty}_{\sigma})$  of

$$\begin{cases} \mathrm{d}u^n + P(u^n \cdot \nabla u^n) \mathrm{d}t - P \pounds_{\xi_k^n}^* u^n \mathrm{d}z_t^{n,k} = 0 \quad \text{on } (0, T_{\max}^n) \times \mathbb{T}^d, \\ u^n = u_0^n \quad \text{on } \{0\} \times \mathbb{T}^d \end{cases}$$
(5.2)

such that if  $T_{\max}^n < \infty$ , then  $\limsup_{t \uparrow T_{\max}^n} |u_t^n|_{H^{m_*}} = \infty$ .<sup>2</sup> Integrating (5.2) over an arbitrary interval  $[s,t] \subset [0, T_{\max}^n)$  and then substituting the equation into the  $dZ^{n,k}$ -integral twice, we obtain

$$u_{st}^{n,P,\natural} = \delta u_{st}^n + \int_{s}^{t} P(u_r^n \cdot \nabla u_r^n) \,\mathrm{d}r - P \pounds_{\xi_k^n}^* u_s^n Z_{st}^{n,k} - P[\pounds_{\xi_k^n}^* P[\pounds_{\xi_l^n}^* u_s^n]] \mathbb{Z}_{st}^{n,lk}, \qquad (5.3)$$

where

$$\begin{split} u_{st}^{n,P,\natural} &:= -\int\limits_{s < t_2 < t_1 < t} P \pounds_{\xi_k^n}^* P u_{t_2}^n \cdot \nabla u_{t_2}^n \mathrm{d}t_2 \, \mathrm{d}z_{t_1}^{n,k} \\ &- \int\limits_{s < t_3 < t_2 < t_1 < t} P \pounds_{\xi_k^n} P \pounds_{\xi_l^n}^* P (u_{t_3}^n \cdot \nabla u_{t_3}^n) \mathrm{d}t_3 \, \mathrm{d}z_{t_2}^{n,l} \, \mathrm{d}z_{t_1}^{n,k} \\ &+ \int\limits_{s < t_3 < t_2 < t_1 < t} P \pounds_{\xi_k^n}^* P \pounds_{\xi_l^n}^* P \pounds_{\xi_q^n}^* u_{t_3}^n \, \mathrm{d}z_{t_3}^{n,q} \, \mathrm{d}z_{t_2}^{n,l} \, \mathrm{d}z_{t_1}^{n,k}, \end{split}$$

$$\begin{cases} \partial_t u^{N,n} + PP_{\leq N}(P_{\leq N}u^{N,n} \cdot \nabla P_{\leq N}u^N) - PP_{\leq N}(\pounds_{\xi_k}^* P_{\leq N}u^{N,n}) \, \dot{z}_t^{n,k} = 0 \quad \text{on } (0,\infty) \times \mathbb{T}^d, \\ u^N = P_{\leq N}u_0 \quad \text{on } \{0\} \times \mathbb{T}^d. \end{cases}$$

<sup>&</sup>lt;sup>2</sup> Indeed, to see this, for given  $N \in \mathbb{N}$ , consider the equation

One may then derive solution estimates independent of N but dependent on  $\int_0^t |\dot{z}^n| dt$  as in [5, Chapter 3] or [4, Chapter 7]. We are not able to derive a priori estimates independent of N and n jointly because the presence of the projection  $P_{\leq N}$  prohibits us from deriving the unbounded rough driver equation (i.e. Definition A.4) for  $\partial_I \omega^{N,n} \otimes \partial_J \omega^{N,n}$ , and hence applying Theorem A.5 to derive solution estimates independent of n as in the proof of Theorem 4.2. For fixed  $n \in \mathbb{N}$ , one passes to the limit as  $N \to \infty$  and establishes all other solution properties (e.g., BKM blow-up criterion) as in [5, Chapter 3] or [4, Chapter 7] or as detailed below for the rough version of the equation.

and hence  $u^{n,P,\natural} \in C_{2,\text{loc}}^{\frac{p}{3}-\text{var}}([0,T_{\max}^{n});\mathring{C}_{\sigma}^{\infty})$ . In particular,  $u^{n}$  is a  $H^{m+2}$ -solution of (5.3) on the interval  $[0,T_{\max}^{n})$  in the sense of Definition 3.1.

By virtue of (5.1) and Theorem 4.2, there are constants  $C_1 = C_1(p, d, m, |\xi|_{W^{m_*+2,\infty}})$ depending in an increasing way on  $|\xi|_{W^{m_*+2,\infty}}$  and constant  $C_2 = C_2(p, d, m, |\xi|_{W^{m+2,\infty}})$ such that for all  $T_* < T_{\max}^n$  satisfying

$$e^{C_1(1+\varpi_{\mathbf{Z}}(0,T_*))}T_* < |u_0|_{H^{m_*}}^{-1},$$

it holds that for all  $n \in \mathbb{N}$ ,

$$\sup_{t \le T_*} |u_t^n|_{H^{m_*}} \le \frac{e^{C_1(1 + \varpi_{\mathbf{Z}}(0, T_*))}}{|u_0^n|_{H^{m_*}}^{-1} - e^{C_1(1 + \varpi_{\mathbf{Z}}(0, T_*))}T_*},$$
(5.4)

$$\sup_{t \le T_*} |u_t^n|_{H^m} \le \sqrt{2} \exp\left(C_2\left(\int_0^{T_*} |\nabla u_s^n|_{L^{\infty}} \, \mathrm{d}r + \varpi_{\mathbf{Z}}(0, T_*)\right)\right) |u_0|_{H^m} \qquad (5.5)$$

$$\le \sqrt{2} \exp\left(C_2\left(\int_0^{T_*} |u_s^n|_{H^{m_*}} \, \mathrm{d}r + \varpi_{\mathbf{Z}}(0, T_*)\right)\right) |u_0|_{H^m}.$$

Furthermore, by Theorem 4.1, there exists a constant  $C = C(p, d, m, |\xi|_{W^{m+2,\infty}})$  such that for all  $(s,t) \in \Delta_{T_*}$  with  $C\varpi_{\mathbf{Z}}(s,t) \leq 1$ , it holds that

$$|u^{n,P,\natural}|_{\frac{p}{3}-\operatorname{var};[s,t];H^{m-3}}^{\frac{p}{3}} \leq C\left(\sup_{s\leq r\leq t}|u_{r}^{n}|_{H^{m}}\varpi_{\mathbf{Z}}(s,t)^{\frac{3}{p}} + \int_{s}^{t}|u_{r}^{n}|_{H^{m}}^{2}\,\mathrm{d}r\varpi_{\mathbf{Z}}(s,t)^{\frac{1}{p}}\right)$$

and for all  $(s,t) \in \Delta_T$  with  $C\varpi_{\mathbf{Z}}(s,t) + C \int_s^t |u_r^n|_{H^m}^2 \, \mathrm{d}r \leq 1$ , it holds that

$$|u^{n}|_{p-\operatorname{var};[s,t];H^{m-1}}^{p} \leq C\left(\int_{s}^{t} |u_{r}^{n}|_{H^{m}}^{2} \,\mathrm{d}r + \sup_{s \leq r \leq t} |u_{r}|_{H^{m}} \left(\left(\int_{s}^{t} |u_{r}^{n}|_{H^{m}}^{2} \,\mathrm{d}r\right)^{\frac{1}{p}} + \varpi_{\mathbf{Z}}(s,t)^{\frac{1}{p}}\right)\right).$$

We deduce that  $\{u^n\}_{n=1}^{\infty}$  is bounded in  $L^{\infty}([0,T_*]; \mathring{H}_{\sigma}^m) \cap C^{p-\text{var}}([0,T_*]; \mathring{H}_{\sigma}^{m-1})$ . In particular,  $\{u^n\}_{n=1}^{\infty}$  is equicontinuous in  $\mathring{H}_{\sigma}^{m-1}$ . For all  $\epsilon > 0$ , there exists a  $\theta \in (0,1)$  such that for all  $(s,t) \in [0,T_*]^2$ ,

$$|u_t^n - u_s^n|_{H^{m-\epsilon}} \lesssim_{d,m} |u_t^n - u_s^n|_{H^{m-1}}^{\theta} \sup_{t \le T_*} |u_t^n|_{H^m}^{1-\theta},$$

which implies that the sequence  $\{u^n\}_{n=1}^{\infty}$  is equicontinuous in  $H^{m-\epsilon}$ . Moreover, by virtue of the boundedness of  $\{u^n\}_{n=1}^{\infty}$  in  $L^{\infty}([0, T_*]; \mathring{H}_{\sigma}^m)$  and the compactness of  $H^m$  in  $H^{m-\epsilon}$ ,

the generalized Arzelá-Ascoli theorem implies that there exists a subsequence  $\{u^{n_j}\}$  that converges to u in  $C([0, T_*]; \mathring{H}^{m-\epsilon}_{\sigma})$  for any  $\epsilon > 0$ . Moreover, we may extract a further subsequence, also denoted by  $\{u^{n_j}\}$  that converges to u in the weak-star topology of  $L^{\infty}([0, T_*]; \mathring{H}^m_{\sigma})$ . Furthermore,  $u \in C^{p-\text{var}}([0, T_*]; \mathring{H}^{m-1}_{\sigma}) \cap L^{\infty}([0, T_*]; \mathring{H}^m_{\sigma})$ . Using the lower semicontinuity of norms in the weak-star topology, we may pass to the limit in (5.4) and (5.5) to obtain (3.11) and (3.12).

Claim.  $u \in C_w([0, T_*]; \mathring{H}^m_{\sigma}).$ 

**Proof.** Fix an arbitrary  $\epsilon > 0$  and  $\phi \in \mathring{H}_{\sigma}^{-m}$ . We need to show that there exists a  $\delta_{\epsilon,\phi} > 0$  such that for all  $(s,t) \in [0,T_*]$  with  $|t-s| < \delta_{\epsilon,\phi}$ , it holds that

$$\langle \phi, u_t - u_s \rangle_{H^m} < \epsilon$$

Since  $\mathring{H}_{\sigma}^{-m+1}$  is dense in  $\mathring{H}_{\sigma}^{-m}$ , there exists a  $\psi_{\epsilon} \in \mathring{H}_{\sigma}^{-m+1}$  such that

$$|\phi - \psi_{\epsilon}|_{H^{-m}} < \frac{\epsilon}{4} \sup_{t \le T_*} |u_t|_{H^m}^{-1}.$$

It follows that for all  $(s,t) \in [0,T_*]^2$ ,

$$\begin{split} \langle \phi, u_t - u_s \rangle_{H^m} &= \langle \psi_{\epsilon}, u_t - u_s \rangle_{H^{m-1}} + \langle \phi - \psi_{\epsilon}, u_t - u_s \rangle_{H^m} \\ &\leq \langle \psi_{\epsilon}, u_t - u_s \rangle_{H^{m-1}} + 2 \sup_{t \leq T_*} |u|_{H^m} |\phi - \psi_{\epsilon}|_{H^{-m}} \\ &< \langle \psi_{\epsilon}, u_t - u_s \rangle_{H^{m-1}} + \frac{\epsilon}{2}. \end{split}$$

Since  $u \in C([0,T_*]; \mathring{H}^{m-1,2}_{\sigma}) \subset C_w([0,T_*]; \mathring{H}^{m-1}_{\sigma})$  we can find  $\delta_{\epsilon,\phi} > 0$  such that for all  $(s,t) \in [0,T_*]^2$  with  $|t-s| < \delta_{\epsilon,\phi}, \langle \psi_{\epsilon}, u_t - u_s \rangle_{H^{m-1}} < \frac{\epsilon}{2}$ , which completes the proof of the claim.  $\Box$ 

We will now show that u is a  $H^m$ -solution of (RE) on the interval  $[0, T_*]$  in the sense of Definition 3.1. By (5.3), for all  $\phi \in \mathring{C}^{\infty}_{\sigma}$  and  $(s, t) \in \Delta_{[0, T_*]}$ , we have

$$\langle u_{st}^{n_j, P, \natural}, \phi \rangle = (\delta u_{st}^{n_j}, \phi)_{L^2} - \int_s^t (u_r^{n_j} \otimes u_r^{n_j}, \nabla \phi)_{L^2} \, \mathrm{d}r + (u_s^{n_j}, \pounds_{\xi_k^n} \phi)_{L^2} Z_{st}^{n_j, k} - (u_s^{n_j}, \pounds_{\xi_l^n} P \pounds_{\xi_k^n} \phi)_{L^2} Z_{st}^{n_j, lk}.$$

Since  $u^{n_j} \to u$  in  $C([0,T_*]; \mathring{L}^2_{\sigma}), \xi^{n_j} \to \xi$  in  $(W^{2,\infty}_{\sigma})^K$ , and  $\mathbf{Z}^{n_j} \to \mathbf{Z}$  in  $C^{p-\text{var}}_g$ , the right-hand-side of the above converges for every  $\phi \in \mathring{C}^{\infty}_{\sigma}$ . Thus,  $\langle u^{n,P,\natural}_{st}, \phi \rangle$  converges to  $\langle u^{\natural,P}_{st}, \phi \rangle$ , which implies that  $u^{P,\natural} \in C^{\frac{p}{2}-\text{var}}([0,T]; \mathring{D}'_{\sigma})$ . Since  $\{u^{n_j,P,\natural}\}$  is bounded

uniformly in  $C_{2,\text{loc}}^{\frac{p}{3}-\text{var}}([0,T_*]; \mathring{H}^{m-3}_{\sigma})$ , we have  $u^{P,\natural} \in C_{2,\text{loc}}^{\frac{p}{3}-\text{var}}([0,T_*]; \mathring{H}^{m-3}_{\sigma})$ , which completes the proof that u is an  $H^m$ -solution in the sense of Definition 3.1. Uniqueness follows immediately from (4.12) in Theorem 4.4.

Finally, we will show that if  $\xi \in (W^{m+4,\infty})^K$ , then  $u \in C([0,T_*]; \mathring{H}_{\sigma}^m)$ . For given  $N \in \mathbb{N}$ , let  $u_0^N = P_{\leq 2^N} u_0 \in \mathring{C}_{\sigma}$ . By the above proof of local well-posedness, since  $\xi \in W_{\sigma}^{m+4,\infty}$  and  $|u_0^N|_{H^{m_*}} \leq |u_0|_{H^{m_*}}$  for all  $N \in \mathbb{N}$ , there exists a sequence of  $H^{m+2}$ -solutions  $\{u^N\}_{N=1}^{\infty} \subset L^{\infty}([0,T_*]; \mathring{H}_{\sigma}^{m+2}) \cap C^{p-\text{var}}([0,T_*]; \mathring{H}_{\sigma}^{m+1})$  of (RE) on interval  $[0,T_*]$  with initial conditions  $\{u_0^N\}_{N=1}^{\infty} \subset \mathring{H}_{\sigma}^{m+2}$ . Moreover, for all  $N \in \mathbb{N}$ ,

$$\sup_{t \le T_*} |u_t^N|_{H^{m_*}} \le \frac{e^{C_1(1+\varpi_{\mathbf{Z}}(0,T_*))}}{|u_0^N|_{H^{m_*}}^{-1} - e^{C_1(1+\varpi_{\mathbf{Z}}(0,T_*))}T_*}$$
$$\le \frac{e^{C_1(1+\varpi_{\mathbf{Z}}(0,T_*))}}{|u_0|_{H^{m_*}}^{-1} - e^{C_1(1+\varpi_{\mathbf{Z}}(0,T_*))}T_*},$$
$$\sup_{t \le T_*} |u_t^N|_{H^m} \le \sqrt{2} \exp\left(C_2\left(\int_0^{T_*} |u_s^N|_{H^{m_*}} \,\mathrm{d}s + \varpi_{\mathbf{Z}}(0,T_*)\right)\right)\right) |u_0|_{H^m}.$$

By virtue of (4.13), for all  $N \in \mathbb{N}$ , we have

$$\sup_{t \le T_*} |u_t^N - u_t|_{H^m} \le \sqrt{q_N(T_*)} |u_0^N - u_0|_{H^m} + T_* q_N(T_*) |u_0^N - u_0|_{H^{m-1}} |u_0^N|_{H^{m+1}},$$

where

$$q_N(T_*) := \sqrt{2} \exp\left(C\left(\int_0^{T_*} (|u_s^N|_{H^m} + |u_s|_{H^m}) \, \mathrm{d}s + \varpi_{\mathbf{Z}}(0, T_*)\right)\right).$$

It follows that

$$2^{-N} |u_0^N|_{H^{m+1}} = 2^{-N} |P_{\leq 2^N} u_0|_{H^{m+1}} \le |u_0|_{H^m}$$
$$\lim_{N \to \infty} 2^N |u_0^N - u_0|_{H^{m-1}} \le \lim_{N \to \infty} \left( \sum_{|n| > 2^N} |n|^{2m} |\hat{u}_0(n)|^2 \right)^{\frac{1}{2}} = 0,$$

which implies that the sequence  $\{u^N\}_{N=1}^{\infty}$  converges to u in  $C([0, T_*]; \mathring{H}_{\sigma}^m)$ , and hence  $u \in C([0, T_*]; \mathring{H}_{\sigma}^m)$ .  $\Box$ 

## 6. Proof of the remaining results

6.1. Recovery of the pressure and harmonic part (Proof of Proposition 3.3)

**Proof of Proposition 3.3.** Define  $\Xi : \Delta_T \to H^{m-3}$  by

$$\Xi_{st} = \pounds_{\xi_k}^* u_s Z_{st}^k + \pounds_{\xi_k}^* P \pounds_{\xi_l}^* u_s \mathbb{Z}_{st}^{lk}.$$

Using that P = I - Q - H (see Section 2.1.1), we find that for all  $(s, t) \in \Delta_T$ , we have

$$u_{st}^{P,\natural} = \delta u_{st} + \int_{s}^{t} u_r \cdot \nabla u_r \, \mathrm{d}r - \int_{s}^{t} Q(u_r \cdot \nabla u_r) \, \mathrm{d}r - \Xi_{st} + Q\Xi_{st} + H\Xi_{st}.$$

We will apply the sewing lemma (i.e., Lemma 2.2) to construct the rough integral

$$I_t = \int_0^t \pounds_{\xi_k}^* u_r \mathrm{d} Z_r^k, \quad I_0 = 0,$$

from the local expansion  $\Xi$ . For all  $(s, \theta, t) \in \Delta_T^2$ ,

$$\delta \Xi_{s\theta t} = -\pounds_{\xi_k}^* u_{s\theta}^{P,\sharp} Z_{\theta t}^k - \pounds_{\xi_k}^* P \pounds_{\xi_l}^* \delta u_{s\theta} \mathbb{Z}_{\theta t}^{lk},$$

where  $u^{P,\sharp}$  is as defined in Theorem 4.1. By Theorem 4.1, for all  $(s,t) \in \Delta_T$  with  $C\varpi_{\mathbf{Z}}(s,t) + C\int_{s}^{t} |u_{r}|_{H^{m}}^{2} \mathrm{d}r \leq 1$ , it holds that

$$\begin{aligned} |\Xi_{st}|_{H^{m-3}} &\leq C \sup_{r \leq T} |u_r|_{H^{m-1}} \varpi_{\mathbf{Z}}(s,t)^{\frac{1}{p}} \\ |\delta \Xi_{s\theta t}|_{H^{m-3}} &\leq C \left( |u^{P,\sharp}|_{\frac{p}{2}-\operatorname{var},[s,t],H^{m-2}}^{\frac{p}{2}} \varpi_{\mathbf{Z}}(s,t)^{\frac{1}{p}} + |u|_{p-\operatorname{var},[s,t],H^{m-1}}^{p} \varpi_{\mathbf{Z}}(s,t)^{\frac{2}{p}} \right). \end{aligned}$$

Thus, applying Lemma 2.2, we find that there is a paths  $I \in C^{p-\text{var}}([0,T]; H^{m-3})$ ,  $I^{Q} = QI \in C^{p-\operatorname{var}}([0,T]; \nabla \mathring{H}^{m-2}), \text{ and } h = HI \in C^{p-\operatorname{var}}([0,T]; \mathbb{R}^{d}) \text{ such that } u^{\natural} := \delta I - \Xi \in C^{\frac{p}{2}-\operatorname{var}}_{2,\operatorname{loc}}([0,T]; H^{m-3}), u^{Q,\natural} := \delta I^{Q} - Q\Xi \in C^{\frac{p}{2}-\operatorname{var}}_{2,\operatorname{loc}}([0,T]; \nabla \mathring{H}^{m-2}) \text{ and } u^{H,\natural} := \delta h - H\Xi \in C^{\frac{p}{2}-\operatorname{var}}_{2,\operatorname{loc}}([0,T]; \mathbb{R}^{d}).$ For all  $t \in [0, T]$ , we have

$$\int_{0}^{t} |Q(u_{r} \cdot \nabla u_{r})|_{H^{m-3}} \, \mathrm{d}r \le \int_{0}^{t} |u_{r} \cdot \nabla u_{r}|_{H^{m-3}} \, \mathrm{d}r \le \int_{0}^{t} |\nabla u_{r}|_{L^{\infty}} |u_{r}|_{H^{m-3}} \, \mathrm{d}r.$$

We define  $q \in C^{p-\operatorname{var}}([0,T]; \mathring{H}^{m-2})$  by

$$\nabla q_t = -\int_0^t Q(u_r \cdot \nabla u_r) \,\mathrm{d}r + I_t^Q.$$

Therefore, we have that for all  $(s, t) \in \Delta_T$ ,

$$u_{st}^{P,\natural} + u_{st}^{\natural} + u_{st}^{Q,\natural} + u_{st}^{H,\natural} = \delta u_{st} + \int_{s}^{t} u_r \cdot \nabla u_r \,\mathrm{d}r - \delta I_{st} + \nabla \delta q_{st} + \delta h_{st},$$

from which we deduce (3.4) since the right-hand-side is a path of  $\frac{p}{3}$ -variation, and hence zero.  $\Box$ 

# 6.2. Maximally extended solution (Proof of Corollary 3.8)

We will give a constructive proof of the maximally extended solution as it will be used in subsequent proofs.

**Remark 6.1.** Both Definition 3.1 and Theorem 3.7 can be extended in the obvious manner to account for an arbitrary initial time  $t_0 \ge 0$ . More precisely, for all  $u_0 \in \dot{H}_{\sigma}^m$  and  $\xi \in (W_{\sigma}^{m+2})^k$ , there exists a unique  $H^m$ -solution  $u \in C_w([t_0, T_*]; \dot{H}_{\sigma}^m) \cap C^{p-\text{var}}([t_0, T_*]; \dot{H}_{\sigma}^{m-1})$  on an interval  $[t_0, T_*]$  with initial condition  $u|_{t=t_0} = u_0$  for any time  $T_*$  satisfying

$$e^{C_1(1+\varpi_{\mathbf{Z}}(t_0,T_*))}(T_*-t_0) < |u_0|_{H^{m_*}}^{-1}$$

and

$$\sup_{t \in [t_0, T_*]} |u_t|_{H^{m_*}} \leq \frac{e^{C_1(1 + \varpi_{\mathbf{Z}}(t_0, T_*))}}{|u_0|_{H^{m_*}}^{-1} - e^{C_1(1 + \varpi_{\mathbf{Z}}(t_0, T_*))}(T_* - t_0)}$$

Moreover, if  $m > m_*$ ,

$$\sup_{t \in [t_0, T_*]} |u_t|_{H^m} \le \sqrt{2} \exp\left(C_2\left(\int_{t_0}^{T_*} |\nabla u_s|_{L^{\infty}} \,\mathrm{d}s + \varpi_{\mathbf{Z}}(t_0, T_*)\right)\right) |u_0|_{H^m}.$$

**Proof of Corollary 3.8.** Let R > 1 and  $\overline{C} > C_1(p, d, |\xi|_{W^{m_*+2,\infty}})$  be arbitrarily given, where  $C_1$  is as given in (3.10) in Theorem 3.7. Define  $f : \mathbb{R}^2_+ \to [0, \infty)$  by

$$f(s,\delta) = e^{\bar{C}(1+\varpi_{\mathbf{Z}}(s,s+\delta))}\delta.$$

Note that for all  $s \in \mathbb{R}_+$ , f(s, 0) = 0 and  $\lim_{\delta \uparrow \infty} f(s, \delta) = \infty$ . Since  $\varpi_{\mathbf{Z}}$  is non-decreasing in its second argument, f is strictly increasing in  $\delta$ , and hence there exists an inverse of f in its second argument  $\delta_* : \mathbb{R}^2_+ \to \mathbb{R}_+$ ; that is, for all  $(s, M) \in \mathbb{R}^2_+$ ,

$$f(s,\delta_*(s,M)) = e^{\overline{C}(1+\varpi_{\mathbf{Z}}(s,\delta_*(s,M)))}\delta_*(s,M) = M.$$

Let  $T_0 = 0$  and  $u_{T_0}^0 = u_0$ . Let  $T_1 = T_0 + \delta_*(T_0, (R + |u_{T_0}^0|_{H^{m_*}})^{-1}))$  so that

$$e^{C_1(1+\varpi_{\mathbf{Z}}(T_0,T_1))}(T_1-T_0) \le e^{\bar{C}(1+\varpi_{\mathbf{Z}}(T_0,T_1))}(T_1-T_0) \le (R+|u_{T_0}^0|_{H^{m_*}})^{-1} < |u_{T_0}^0|_{H^{m_*}}^{-1}.$$

By Theorem 3.7, there exists an  $H^m$ -solution  $u^1$  of (RE) on the interval  $[0, T_1]$ . Let  $T_2 = T_1 + \delta_*(T_1, (R + |u_{T_1}^1|_{H^{m_*}})^{-1}))$ . Since

$$e^{C_1(1+\varpi_{\mathbf{Z}}(T_1,T_2))}(T_2-T_1) \le e^{\bar{C}(1+\varpi_{\mathbf{Z}}(T_1,T_2))}(T_1-T_0) = (R+|u_{T_1}^1|_{H^{m_*}})^{-1} < |u_{T_1}^1|_{H^{m_*}}^{-1},$$

by Remark 6.1, there exists an  $H^m$ -solution  $\tilde{u}^2$  of (RE) on the interval  $[T_1, T_2]$ . Let  $u^2 = u^1$  on  $[0, T_1]$  and  $u^2 = \tilde{u}^2$  on  $[T_1, T_2]$  so that  $u^2$  is a  $H^m$ -solution of (RE) on the interval  $[0, T_2]$ . Proceeding by induction, we define

$$T_{l+1} = T_l + \delta_*(T_l, (R + |u_{T_l}^l|_{H^{m_*}})^{-1}), \ l \in \{3, \dots, \},$$

and appeal to Remark 6.1 to obtain an  $H^m$ -solution  $u^n$  of (RE) on the interval  $[0, T_l]$  for all  $l \in \mathbb{N}$ . Let  $T_{\max} = \sup_{n \in \mathbb{N}_0} T_l$  and  $u^{\max} = \lim_{l \to \infty} u^l$ . It follows that  $u^{\max}$  is the unique  $H^m$ -solution on the open  $[0, T_{\max})$  by virtue of (4.12) in Theorem 4.4.

Assume that  $T_{\max} < \infty$ . Suppose, by contradiction, that  $\limsup_{t \uparrow T_{\max}} |u_t|_{H^{m_*}} < \infty$ . In particular, there exists M > 0 such that  $|u_t|_{H^{m_*}} < M$  for all  $t < T_{\max}$ . For arbitrarily chosen  $\epsilon > 0$ , there exists a  $L = L(\epsilon) > 0$  such that for all l > L,  $T_{l+1} - T_l < \epsilon$ , which implies

$$e^{\bar{C}(1+\varpi_{\mathbf{Z}}(0,T_{\max}))}(R+M)^{-1} < e^{\bar{C}(1+\varpi_{\mathbf{Z}}(T_{l},T_{l+1}))}(R+|u_{T_{l}}|_{H^{m}})^{-1} < \epsilon.$$

Choosing  $\epsilon > 0$  smaller than the left-hand-side, we obtain a contradiction.

Suppose, by contradiction, there exists  $T \in (T_{\max}, \infty]$  and an  $H^m$ -solution  $\bar{u}$  of (RE) on the interval  $[0, \bar{T})$ . By virtue of uniqueness (i.e., (4.12) in Theorem 4.4), we have  $u \equiv \bar{u}$  on  $[0, T_{\max})$ , which implies

$$\limsup_{t\uparrow T_{\max}} |u_t|_{H^{m_*}} = |\bar{u}_{T_{\max}}|_{H^{m_*}} < \infty \quad \Rightarrow \quad T_{\max} = \infty.$$

This leads to a contradiction, which precludes the existence of an extension of u.  $\Box$ 

### 6.3. BKM blow-up criterion (Proof of Theorem 3.9)

**Proof of Theorem 3.9.** The strategy of the proof is to first construct an approximation sequence of  $H^{m+2}$ -solutions that converges to u on any interval  $[0,T] \subset [0,T_{\max})$ . We will then use the solution estimate given in (4.4) in Theorem 4.2 and then pass to the limit in both sides of the solution bound. The approximation sequence is constructed separately

for  $m = m_*$  and  $m > m_*$ . If  $m = m_*$ , we only approximate the initial condition, assume  $(W^{m_*+4,\infty}_{\sigma})^K$ , and use (4.13) in Theorem 4.4 to establish convergence in  $C([0,T]; \mathring{H}^{m_*}_{\sigma})$ , which is needed to continue the approximating sequence up to  $T_{\max}$ . If  $m > m_*$ , we can avoid the assumption  $\xi \in (W^{m+4,\infty})^K$  for all m by smoothing the initial condition and  $\xi$  and relying on compactness to obtain convergence in  $C([0,T]; \mathring{H}^{m_*}_{\sigma})$ , which allows us to continue the approximating sequence up to  $T_{\max}$ .

**Case**  $m = m_*$ . Let  $\{T_n\}_{n=0}^{\infty}$  be the sequence of times specified in the proof of Corollary 3.8 converging to  $T_{\max}$ . In the proof of Theorem 3.7, we showed that the sequence  $\{u^n\}_{n=1}^{\infty}$  of  $H^{m+2}$ -solutions on the interval  $[0, T_1]$  corresponding to the initial conditions  $\{P_{\leq 2^n}u_0\}_{n=1}^{\infty} \in C_{\sigma}^{\infty}$  converges to u in  $C([0, T_1]; \mathring{H}_{\sigma}^{m*})$ . In particular, there exists an  $N_2 \in \mathbb{N}$  be such that for all  $n \geq N_2$ ,  $|u_{T_1}^n|_{H^{m*}} < |u_{T_1}|_{H^{m*}} + 2^{-1}R$ . Thus, by Remark 6.1 and (4.12) (i.e., uniqueness), we can extend the solutions  $\{u^n\}_{n=N_2}^{\infty}$  to the interval  $[0, T_2]$  such that for all  $n \geq N_2$ ,

$$\sup_{t \in [T_1, T_2]} |u_t^n|_{H^{m_*}} \le \frac{e^{C_1(1 + \varpi_{\mathbf{Z}}(T_1, T_2))}}{|u_{T_1}^n|_{H^{m_*}}^{-1} - e^{C_1(1 + \varpi_{\mathbf{Z}}(T_1, T_2))}(T_2 - T_1)} \le \frac{e^{C_1(1 + \varpi_{\mathbf{Z}}(T_1, T_2))}}{(2^{-1}R + |u_{T_1}|_{H^{m_*}})^{-1} - (R + |u_{T_1}|_{H^{m_*}})^{-1}}$$

Repeating the argument at the end of the proof of Theorem 3.7 (i.e., applying (4.13)) on the interval  $[0, T_2]$ , we get that the sequence  $\{u^n\}_{n=N_2}^{\infty}$  of  $H^{m+2}$ -solutions on the interval  $[0, T_2]$  converges to u in  $C([0, T_2]; \mathring{H}_{\sigma}^{m_*})$ . Proceeding inductively, for all  $l \in \mathbb{N}$ , we find an  $N_l \in \mathbb{N}$  such that the sequence  $\{u^n\}_{n=N_l}^{\infty}$  converges to u in  $C([0, T_l]; \mathring{H}_{\sigma}^{m_*})$ .

**Case**  $m > m_*$ . There exists a sequence  $\{(u_0^n, \xi^n)\}_{n=1}^{\infty} \in \mathring{C}_{\sigma}^{\infty} \times (C_{\sigma}^{\infty})^K$  that converges to  $(u_0, \xi)$  in  $H_{\sigma}^m \times W^{m+1,\infty}$  and such that for all  $n \in \mathbb{N}$ ,  $|u_0^n|_{H^m} \leq |u_0|_{H^m}, |\xi^n|_{W^{m+2,\infty}} \leq |\xi|_{W^{m+2,\infty}}$ . Let  $\{T_n\}_{n=0}^{\infty}$  be the sequence of times specified in the proof of Corollary 3.8 converging to  $T_{\max}$ . By Theorem 3.7, there exists a sequence of  $H^{m+2}$ -solutions  $\{u^n\}_{n=1}^{\infty}$  on the interval  $[0, T_1]$  corresponding to the data  $\{(u_0^n, \xi^n)\}_{n=1}^{\infty}$  such that for all  $n \in \mathbb{N}$ ,

$$\sup_{t \in [0,T_1]} |u_t^n|_{H^{m_*}} \le \frac{e^{C_1(1+\varpi_{\mathbf{Z}}(0,T_1))}}{|u_0|_{H^{m_*}}^{-1} - (R+|u_0|_{H^{m_*}})^{-1}}$$
$$\sup_{t \in [0,T_1]} |u_t^n|_{H^m} \le \sqrt{2} \exp\left(C_2\left(\int_0^{T_1} |u_s^n|_{H^{m_*}} \,\mathrm{d}s + \varpi_{\mathbf{Z}}(0,T_1)\right)\right) |u_0|_{H^m}.$$

Following the proof of Theorem 3.7 (i.e., applying Theorem 4.1, compactness, and Arzelá-Ascoli, and then passing to the limit) and using uniqueness, we find that the full sequence  $\{u^n\}_{n=1}^{\infty}$  converges to u in  $C([0, T_1]; \mathring{H}_{\sigma}^{m-\epsilon})$  for any  $\epsilon > 0$  and in the weak-star topology of  $L^{\infty}([0, T_1]; \mathring{H}_{\sigma}^m)$ . In particular, there exists an  $N_2 \in \mathbb{N}$  be such that for all  $n \geq N_2$ ,  $|u_{T_1}^n|_{H^{m_*}} < |u_{T_1}|_{H^{m_*}} + 2^{-1}R$ . By Remark 6.1 and (4.12) (i.e., uniqueness), we may continue the solutions to the interval  $[0, T_2]$  such that for all  $n \geq N_2$ 

$$\sup_{t \in [T_1, T_2]} |u_t^n|_{H^{m_*}} \le \frac{e^{C_1(1 + \varpi_{\mathbf{Z}}(T_1, T_2))}}{(2^{-1}R + |u_{T_1}|_{H^{m_*}})^{-1} - (R + |u_{T_1}|_{H^{m_*}})^{-1}}$$
$$\sup_{t \in [0, T_2]} |u_t^n|_{H^m} \le \sqrt{2} \exp\left(C_2\left(\int_0^{T_2} |u_s^n|_{H^{m_*}} \,\mathrm{d}s + \varpi_{\mathbf{Z}}(0, T_2)\right)\right) |u_0|_{H^m}.$$

Again, following the proof of Theorem 3.7 and using uniqueness, we get that the full sequence  $\{u^n\}_{n=N_2}^{\infty}$  converges to u in  $C([0,T_2]; \mathring{H}_{\sigma}^{m-\epsilon})$  for any  $\epsilon > 0$  and in the weak-star topology of  $L^{\infty}([0,T_2]; \mathring{H}_{\sigma}^m)$ . Proceeding inductively, for all  $l \in \mathbb{N}$ , we find an  $N_l \in \mathbb{N}$  such that the sequence  $\{u^n\}_{n=N_l}^{\infty}$  converges to u in  $C([0,T_l]; \mathring{H}_{\sigma}^{m-\epsilon})$  for any  $\epsilon > 0$  and in the weak-star topology of  $L^{\infty}([0,T_l]; \mathring{H}_{\sigma}^m)$ .

We will now show (3.13). Let  $T < T_{\max} = \sup_{l \in \mathbb{N}} T_l$  be arbitrarily given and  $N(T) \in \mathbb{N}$  be such that the sequences constructed above  $\{u^n\}_{n=N(T)}^{\infty}$  of converges to u in  $C([0,T]; \mathring{H}_{\sigma}^{m_*})$  and in the weak-star topology of  $L^{\infty}([0,T]; \mathring{H}_{\sigma}^{m})$  if  $m > m_*$ . It follows that  $\{\omega^n = \mathbf{d}\flat u^n\}_{n=N(T)}^{\infty}$  converges to  $\omega$  in  $C([0,T]; L^{\infty})$ .<sup>3</sup> By (4.4) in Theorem (4.4), there exist constants  $C_1 = C_1(d,m)$  and  $C_2 = C_2(p,d,m,|\xi|_{W^{m+2,\infty}})$  such that for all  $n \geq N(T)$ ,

$$\sup_{t \le T} |u_t^n|_{H^m} \le C_1(1+|u_0|_{H^m}) \exp\left(C_2(1+\varpi_{\mathbf{Z}}(0,T)) \exp\left(C_2 \int_0^T |\omega_s^n|_{L^{\infty}} \,\mathrm{d}s\right)\right).$$

Using the lower semi-continuity of weak-star convergence if  $m > m_*$ , we pass to the limit as  $n \to \infty$  on both sides of the inequality to obtain (3.13).

If  $T_{\max} < \infty$ , then  $\limsup_{t \uparrow T_{\max}} |u_t|_{H^m} = \infty$ , which yields  $\limsup_{t \uparrow T_{\max}} \int_0^t |\omega_s|_{L^{\infty}} = \infty$  by (3.13). Conversely, if  $\limsup_{t \uparrow T_{\max}} \int_0^t |\omega_s|_{L^{\infty}} ds = \infty$ , then

$$\infty = \limsup_{t \uparrow T_{\max}} \int_{0}^{t} |\omega_{s}|_{L^{\infty}} ds \le \limsup_{t \uparrow T_{\max}} \int_{0}^{t} |u_{s}|_{H^{m}} ds$$

which implies  $\limsup_{t \uparrow T_{\max}} |u_t|_{H^m} = \infty$ .  $\Box$ 

## 6.4. Global well-posedness in two-dimensions (Proof of Theorem 3.10)

**Proof of Corollary 3.10.** Consider the sequence  $\{u^n\}_{n=1}^{\infty}$  of  $H^{\infty}$ -solutions on  $[0, T_{\max}^n)$  corresponding to the data  $\{(u_0^n, \mathbf{Z}^n, \xi^n)\}_{n=1}^{\infty} \in \mathring{C}_{\sigma}^{\infty} \times C_g^{1-\operatorname{var}} \times (C_{\sigma}^{\infty})^K$  from the proof of Theorem 3.7. For given  $n \in \mathbb{N}$ , let  $\tilde{\omega}^n = \operatorname{curl} u^n$ , so that (see Remark 3.5),

<sup>&</sup>lt;sup>3</sup> We actually only need convergence of  $\{u^n\}$  in  $C([0,T]; \mathring{H}^{m_*-\epsilon}_{\sigma})$  for any  $\epsilon > 0$  to conclude this. However, in order to construct the approximating sequence up to the maximal time  $T_{\max}$ , we needed the convergence in  $C([0,T]; \mathring{H}^{m_*}_{\sigma})$ .

$$\begin{cases} \mathrm{d}\tilde{\omega}^n + u^n \cdot \nabla \tilde{\omega}^n \mathrm{d}t + \xi_k^n \cdot \nabla \tilde{\omega}^n \mathrm{d}z_t^{n,k} = 0 \quad \text{on } [0, T_{\max}^n) \times \mathbb{T}^d, \\ \tilde{\omega}^n = \tilde{\omega}_0^n \quad \text{on } \{0\} \times \mathbb{T}^d. \end{cases}$$

It follows that for all  $n \in \mathbb{N}$ ,  $t \in [0, \infty)$ , and  $p \in [2, \infty]$ ,

$$|\omega_t^n|_{L^p} = |\omega_0^n|_{L^p} \le |\omega_0|_{L^p}.$$
(6.1)

Moreover,  $T_{\max}^n = \infty$  since (see, e.g., [5, Chapter 3] or [4, Chapter 7]) for all  $n \in \mathbb{N}$ ,

$$|u_t^n|_{H^m} \le C_1(1+|u_0^n|_{H^m}) \exp\left(C_2(1+\int_0^t |\dot{z}_s^n| \mathrm{d}s) \exp\left(C_2t|\omega_0^n|_{L^{\infty}}\right)\right)$$

Owing to (4.4) in Theorem 4.2, there exist constants  $C_1 = C_1(m)$  and  $C_2 = C_2(p, m, |\xi|_{W^{m+2,\infty}})$  such that for all  $n \in \mathbb{N}$  and  $t \ge 0$ ,

$$|u_t^n|_{H^m} \le C_1(1+|u_0|_{H^m}) \exp\left(C_2(1+\varpi_{\mathbf{Z}}(0,t))\exp\left(C_2t|\omega_0|_{L^{\infty}}\right)\right).$$
(6.2)

Proceeding as in the proof of Theorem 3.7 and using uniqueness, we find that  $\{u^n\}_{n=1}^{\infty}$  converges to u in  $C([0,\infty); \mathring{H}_{\sigma}^{m-\epsilon})$  for any  $\epsilon > 0$  and in the weak-star topology of  $L^{\infty}([0,\infty); \mathring{H}_{\sigma}^m)$ . We may then pass to the limit (6.2) using lower semicontinuity of weak-star convergence to obtain (3.15). Since there always exists an  $\epsilon > 0$  such that  $m - \epsilon > \frac{d}{2} + 1$ ,  $\{\tilde{\omega}^n\}_{n=1}^{\infty}$  converges to  $\omega$  in  $C([0,T]; L^{\infty})$ , and thus passing to the limit in (6.1) yields (3.14).  $\Box$ 

#### 6.5. Continuous dependence on data (Proof of Corollary 3.11)

**Proof of Corollary 3.11.** Let R > 1 be such that for all  $n \in \mathbb{N}$ ,

$$|u_0^n|_{H^m} \le R, \quad |\xi^n|_{W^{m+2,\infty}} + |\xi|_{W^{m+2,\infty}} \le R, \quad \varpi_{\mathbf{Z}^n}(s,t) \le R + \varpi_{\mathbf{Z}}(s,t), \; \forall (s,t) \in \Delta_{[0,\infty)}.$$

We will establish the convergence of  $\{u^n\}$  for the cases d = 2 and  $m > m_*$  separately. **Case** d = 2. Owing to (3.15), there exists a constant  $\overline{C} = \overline{C}(p, m, R)$  such that for all and  $n \in \mathbb{N}$  and  $t \ge 0$ ,

$$|u_t^n|_{H^m} \le C_1(1+R) \exp\left(\bar{C}(1+\varpi_{\mathbf{Z}}(0,t))\exp\left(\bar{C}|\tilde{\omega}_0|_{L^{\infty}}t\right)\right).$$

Proceeding as in the proof of Theorem 3.7, we find that  $\{u^n\}_{n=N_1}^{\infty}$  converges to u in  $C([0,\infty); \mathring{H}_{\sigma}^{m-\epsilon})$  for any  $\epsilon > 0$  and in the weak-star topology of  $L^{\infty}([0,\infty); \mathring{H}_{\sigma}^{m})$ .

**Case**  $m > m_*$ . Let  $C_1(|\xi|_{W^{m_*+2,\infty}}) = C_1(p,d,m,|\xi|_{W^{m_*+2,\infty}})$  denote the constant appearing in (3.10). Let  $\overline{C} = RC_1(p,d,m,R)$  and note that for all  $(s,t) \in \Delta_{[0,\infty)}$ ,

$$C_1(|\xi_n|_{W^{m_*+2,\infty}})(1+\varpi_{\mathbf{Z}^n}(s,t)) \vee C_1(|\xi|_{W^{m_*+2,\infty}})(1+\varpi_{\mathbf{Z}}(s,t)) \le \bar{C}(1+\varpi_{\mathbf{Z}}(s,t)).$$

Let  $\{T_n\}_{n=0}^{\infty}$  be the sequence of times specified in the proof of Corollary 3.8 with R and  $\bar{C}$  as just specified. We then proceed as in the proof of Theorem 3.9 to show that for all  $T < T_{\max}$ , there exists an  $N(T) \in \mathbb{N}$  such that the sequence  $\{u^n\}_{n=N(T)}^{\infty}$  of  $H^m$ -solutions converges to u in  $C([0,T]; \mathring{H}_{\sigma}^{m-\epsilon})$  for any  $\epsilon > 0$  and in the weak-star topology of  $L^{\infty}([0,T]; \mathring{H}_{\sigma}^m)$ .

We will now turn our attention to showing the convergence of the pressure and harmonic constant. Let  $T < T_{\max}$  and N(T) = 1 if d = 2 and N(T) as specified if  $m > m_*$ . By Proposition 3.3, for all  $n \ge N(T)$ , there exists  $q^n \in C^{p-\text{var}}([0,T]; \nabla \mathring{H}^{m-2})$  and  $h^n \in C^{p-\text{var}}([0,T]; \mathbb{R}^d)$  such that

$$u_t^n - u_0^n + \int_0^t u_s^n \cdot \nabla u_s^n \, \mathrm{d}s - \int_0^t \pounds_{\xi_k^n}^* u_s^n \, \mathrm{d}\mathbf{Z}_s^{n,k} = -\nabla q_t^n - h_t^n,$$

where, upon adopting the notation in the proof of Proposition 3.3, for all  $(s,t) \in \Delta_T$ , we have

$$\nabla \delta q_{st}^n := \int_{s}^{t} Q(u_r^n \cdot \nabla u_r^n) \mathrm{d}r + Q \Xi_{st}^n + u^{n,Q,\natural}, \quad \delta h_{st}^n = H \Xi_{st}^n + u^{n,H,\natural},$$

and

$$\Xi_{st}^n = \pounds_{\xi_k^n}^* u_s^n Z_{st}^{n,k} + \pounds_{\xi_k^n}^* P \pounds_{\xi_l^n}^* u_s^n \mathbb{Z}_{st}^{n,lk}.$$

Thus, for all  $(s,t) \in \Delta_T$ 

$$\begin{aligned} \nabla \delta q_{st}^n|_{H^{m-3}} &\leq (t-s) \sup_{s \leq r \leq t} |\nabla u_r^n|_{L^{\infty}} |u_r^n|_{H^{m-3}} \,\mathrm{d}r + |\Xi_{st}^n|_{H^{m-3}} + |u^{n,Q,\natural}|_{H^{m-3}} \\ |\delta h_{st}^n| &\leq |\Xi_{st}^n|_{H^{m-3}} + |u^{n,Q,\natural}|_{H^{m-3}}. \end{aligned}$$

There exists a constant C = C(p, d, m, R) such that

$$|\Xi_{st}^{n}|_{H^{m-3}} \le C \sup_{r \le T} |u_{r}^{n}|_{H^{m-1}} \varpi_{\mathbf{Z}^{n}}(s,t)^{\frac{1}{p}}.$$

Theorem 4.1 implies that there exists a constant C = C(p, d, m, R) such that for all  $(s,t) \in \Delta_T$  with  $C\varpi_{\mathbf{Z}^n}(s,t) + C \int_s^t |u_r^n|_{H^m}^2 dr \leq 1$ , it holds that

$$\begin{split} |\delta\Xi_{s\theta t}^{n}|_{H^{m-3}} &\leq C \left( |u^{n,P,\sharp}|_{\frac{p}{2}-\operatorname{var},[s,t],H^{m-2}}^{\frac{p}{2}} \varpi_{\mathbf{Z}^{n}}(s,t)^{\frac{1}{p}} + |u^{n}|_{p-\operatorname{var},[s,t],H^{m-1}}^{p} \varpi_{\mathbf{Z}^{n}}(s,t)^{\frac{2}{p}} \right) \\ |u^{n,P,\sharp}|_{\frac{p}{2}-\operatorname{var};[s,t];H^{m-2}}^{\frac{p}{2}} &\leq C \left( \int_{s}^{t} |u_{r}^{n}|_{H^{m}}^{2} \,\mathrm{d}r + \sup_{s \leq r \leq t} |u_{r}^{n}|_{H^{m}}^{n} \varpi_{\mathbf{Z}^{n}}(s,t)^{\frac{2}{p}} \right) \end{split}$$

The bound (2.13) in the sewing lemma then implies that

$$u^{n,Q,\natural} := \delta I^{n,Q} - Q\Xi^n$$
 and  $u^{n,H,\natural} := \delta h^n - H\Xi^n$ ,

are bounded independent of n in  $C_{2,\text{loc}}^{\frac{p}{3}-\text{var}}([0,T]; \nabla \mathring{H}^{m-2})$  and  $C_{2,\text{loc}}^{\frac{p}{3}-\text{var}}([0,T]; \mathbb{R}^d)$ , respectively. Therefore,  $\{(\nabla q^n, h^n)\}_{n=N(T)}^{\infty}$  is bounded in  $C^{p-\text{var}}([0,T]; \nabla \mathring{H}^{m-2}) \times C^{p-\text{var}}([0,T]; \mathbb{R}^d)$ . Following the proof of Theorem 3.7 and using uniqueness of u from which uniqueness of (q,h) follows, we deduce that  $\{(\nabla q^n, h^n)\}_{n=N(T)}^{\infty}$  converges to (q,h) in  $C([0,T]; \nabla \mathring{H}^{m-2-\epsilon}) \times C([0,T]; \mathbb{R}^d)$  for any  $\epsilon > 0$ .  $\Box$ 

#### Appendix A. Unbounded rough drivers

In this section, we present some elements of the theory of unbounded rough drivers [9] and the associated remainder estimates. We use Theorem A.5 in Section 4 to derive a priori estimates of the remainder and solution (i.e., Theorems 4.1, 4.2, and 4.4).

**Definition A.1** (Scale). We say a sequence of Banach spaces  $(E_n, |\cdot|_n)_{n=0}^3 = (E_n)$  is a scale, if  $E_{n+1}$  is continuously embedded into  $E_n$  for all  $n \in \{0, 1, 2, 3\}$ . Denote by  $E_{-n}$  the strong topological dual of  $E_n$ .

**Definition A.2** (Unbounded rough driver). For a given interval [0,T], let  $A^i : \Delta_T \to \mathcal{L}(E_{-n}; E_{-(n+i)})$  for  $n \in \{0, 1, 2\}$  and  $i \in \{1, 2\}$ . For a given  $p \in [2, 3)$ , a pair of 2-index maps  $\mathbf{A} = (A^1, A^2)$  is called an unbounded *p*-rough driver on the interval [0,T] with respect to the scale  $(E_n)$  if there exists a regular control  $\varpi_{\mathbf{A}}$  on [0,T] such that for every  $(s,t) \in \Delta_T$ ,

$$|A_{st}^{1}|_{\mathcal{L}(E_{-n};E_{-(n+1)})}^{p} \leq \varpi_{\mathbf{A}}(s,t) \text{ for } n \in \{0,2\},$$

$$|A_{st}^{2}|_{\mathcal{L}(E_{-n};E_{-(n+2)})}^{\frac{p}{2}} \leq \varpi_{\mathbf{A}}(s,t) \text{ for } n \in \{0,1\},$$
(A.1)

and, in addition, Chen's relations hold:

$$\delta A_{s\theta t}^1 = 0 \quad \text{and} \quad \delta A_{s\theta t}^2 = -A_{\theta t}^1 A_{s\theta}^1, \qquad \forall (s, \theta, t) \in \Delta_T^2.$$
(A.2)

**Definition A.3** (Smoothing). We say a family of operators  $(J^{\eta})_{\eta \in (0,1)}$  is a smoothing on a given scale  $(E_n)$  if the following conditions are satisfied:

$$|J^{\eta} - I|_{\mathcal{L}(E_m; E_n)} \lesssim \eta^{m-n} \quad \text{for}(n, m) \in \{(0, 1), (0, 2), (1, 2)\} |J^{\eta}|_{\mathcal{L}(E_n; E_m)} \lesssim \eta^{-(m-n)} \quad \text{for}(n, m) \in \{(1, 1), (1, 2), (2, 2), (1, 3), (2, 3)\}.$$

**Definition A.4** (Solution of unbounded rough driver equation). Let  $\mathbf{A} = (A^1, A^2)$  be a continuous unbounded *p*-rough driver on [0, T] with respect to a scale  $(E_n)$ . Let  $\mu \in$ 

 $C^{1-\text{var}}([0,T]; E_{-1})$ . Assume that  $(E_n)$  admits a smoothing. A bounded path  $f : [0,T] \to E_{-0}$  is called a solution of

$$df + \mu(dt) + \mathbf{A}(dt)f = 0 \tag{A.3}$$

on the interval [0, T], provided  $f^{\natural} : \Delta_T \to E_{-3}$  defined for every  $(s, t) \in \Delta_T$  and  $\phi \in E_3$  by

$$\langle f_{st}^{\natural}, \phi \rangle = \langle \delta f_{st}, \phi \rangle + \langle \delta \mu_{st}, \phi \rangle + \langle f_s, (A_{st}^{1,*} + A_{st}^{2,*})\phi \rangle$$

satisfies  $f^{\natural} \in C_{2,\text{loc}}^{\frac{p}{3}-\text{var}}([0,T]; E_{-3}).$ 

Define the map  $f^{\sharp}: \Delta_T \to E_{-3}$  for every  $(s, t) \in \Delta_T$  and  $\phi \in E_3$  by

$$\langle f_{st}^{\sharp}, \phi \rangle = \langle \delta f_{st}, \phi \rangle + \langle f_s, A_{st}^{1,*} \phi \rangle = -\langle \delta \mu_{st}, \phi \rangle - \langle f_s, A_{st}^{2,*} \phi \rangle + \langle f_{st}^{\natural}, \phi \rangle.$$

The first expression for  $f_{st}^{\sharp}$  consists of terms that are less regular in time and more regular in space than the second expression for  $f_{st}^{\sharp}$ . The following theorem is proved using interpolation, and hence properties of the smoothing operators, and the sewing lemma (i.e., Lemma 2.2). Its proof can be found, for example, in [39, Proposition 3.1].

**Theorem A.5** (Unbounded rough driver estimates). Let u be a solution of (A.3) and assume that there exists a regular control  $\varpi_{\mu}$  on the interval [0,T] such that for all  $(s,t) \in \Delta_T$ ,

$$|\delta\mu_{st}|_{E_{-1}} \le \varpi_{\mu}(s, t). \tag{A.4}$$

Then there exist positive constants C = C(p) and L = L(p) such that for all  $(s,t) \in \Delta_T$ with  $\varpi_{\mathbf{A}}(s,t) \leq L$  it holds that

$$|f^{\natural}|_{\frac{p}{3}-\operatorname{var};[s,t];E_{-3}}^{\frac{p}{2}} \leq C\left(\sup_{s\leq r\leq t}|f_{r}|_{E_{-0}}\varpi_{\mathbf{A}}(s,t)^{\frac{3}{p}} + \varpi_{\mu}(s,t)\varpi_{\mathbf{A}}(s,t)^{\frac{1}{p}}\right)$$

Furthermore, for all  $(s,t) \in \Delta_T$  with  $\varpi_{\mathbf{A}}(s,t) + \varpi_{\mu}(s,t) \leq L$  it holds that

$$\begin{split} |f^{\sharp}|_{\frac{p}{2}-\mathrm{var};[s,t];E_{-2}}^{\frac{p}{2}} &\leq C\left(\varpi_{\mu}(s,t) + \sup_{s \leq r \leq t} |f_{r}|_{E_{-0}} \varpi_{\mathbf{A}}(s,t)^{\frac{2}{p}})\right), \\ |f|_{p-\mathrm{var},[s,t];E_{-1}}^{p} &\leq C\left(\varpi_{\mu}(s,t) + \sup_{s \leq r \leq t} |f_{r}|_{E_{-0}} (\varpi_{\mu}(s,t)^{\frac{1}{p}} + \varpi_{\mathbf{A}}(s,t)^{\frac{1}{p}})\right). \end{split}$$

## Appendix B. Rough Gronwall's lemma

In this section, we state a rough version of Gronwall's lemma. The proof can be found, for example, in [23].

**Lemma B.1** (Rough Gronwall's lemma). Let L and p denote positive constants. Let  $\kappa \in L^1(I)$  and  $\varpi$  be a regular control on the interval [0,T]. Let  $\phi : \Delta_T \to \mathbb{R}_+$  be such that  $\phi(s,t) \leq \phi(0,T)$  for all  $(s,t) \in \Delta_T$ . Assume that  $G : [0,T] \to \mathbb{R}_+$  is such that for every  $(s,t) \in \Delta_T$  with  $\varpi(s,t) \leq L$ ,

$$\delta G_{st} \le \phi(s,t) + \int_{s}^{t} \kappa_r G_r \, \mathrm{d}r + \varpi(s,t)^{\frac{1}{p}} \sup_{r \le t} G_t.$$

Then there exists a positive constant  $\beta = \beta(L, p)$  such that

$$\sup_{0 \le t \le T} G_t \le 2 \exp\left(\beta \left(\int_0^T \kappa_r \,\mathrm{d}r + \varpi(0,T)\right)\right) \left(G_0 + \phi(0,T)\right).$$

### References

- Vladimir I. Arnold, Boris A. Khesin, Differential geometry of diffeomorphism groups, in: Topological Methods in Hydrodynamics, 1998, pp. 195–257.
- [2] Ralph Abraham, Jerrold E. Marsden, Tudor Ratiu, Manifolds, Tensor Analysis, and Applications, vol. 75, Springer Science & Business Media, 2012.
- [3] Judith Berner, Ulrich Achatz, Lauriane Batte, Lisa Bengtsson, Alvaro de la Cámara, Hannah M. Christensen, Matteo Colangeli, Danielle R.B. Coleman, Daan Crommelin, Stamen I. Dolaptchiev, et al., Stochastic parameterization: toward a new view of weather and climate models, Bull. Am. Meteorol. Soc. 98 (3) (2017) 565–588.
- [4] Hajer Bahouri, Jean-Yves Chemin, Raphaël Danchin, Fourier Analysis and Nonlinear Partial Differential Equations, vol. 343, Springer Science & Business Media, 2011.
- [5] Andrea L. Bertozzi, Vorticity and Incompressible Flow, Cambridge Texts in Applied Mathematics, Cambridge University Press, 2001.
- [6] Luigi C. Berselli, Franco Flandoli, On a stochastic approach to eddy viscosity models for turbulent flows, in: Advances in Mathematical Fluid Mechanics, Springer, 2009, pp. 55–81.
- [7] Dominic Breit, Eduard Feireisl, Martina Hofmanová, Stochastically forced compressible fluid flows, in: Stochastically Forced Compressible Fluid Flows, De Gruyter, 2018.
- [8] Zdzisław Brzeźniak, Franco Flandoli, Mario Maurelli, Existence and uniqueness for stochastic 2D Euler flows with bounded vorticity, Arch. Ration. Mech. Anal. 221 (1) (2016) 107–142.
- [9] Ismael Bailleul, Massimiliano Gubinelli, Unbounded rough drivers, Ann. Fac. Sci. Toulouse Math.
   (6) 26 (4) (2017) 795–830.
- [10] James Thomas Beale, Tosio Kato, Andrew Majda, Remarks on the breakdown of smooth solutions for the 3-D Euler equations, Commun. Math. Phys. 94 (1) (1984) 61–66.
- [11] Zdzisław Brzeźniak, Utpal Manna, Debopriya Mukherjee, Wong–Zakai approximation for the stochastic Landau–Lifshitz–Gilbert equations, J. Differ. Equ. 267 (2) (2019) 776–825.
- [12] Årpád Bényi, Tadahiro Oh, The Sobolev inequality on the Torus revisited, Publ. Math. (Debr.) 83 (3) (2013) 359.
- [13] Ismael Bailleul, Sebastian Riedel, Michael Scheutzow, Random dynamical systems, rough paths and rough flows, J. Differ. Equ. 262 (12) (2017) 5792–5823.
- [14] Colin Cotter, Dan Crisan, Darryl D. Holm, Wei Pan, Igor Shevchenko, Numerically modeling stochastic Lie transport in fluid dynamics, Multiscale Model. Simul. 17 (1) (2019) 192–232.
- [15] Colin Cotter, Dan Crisan, Darryl Holm, Wei Pan, Igor Shevchenko, Data assimilation for a quasigeostrophic model with circulation-preserving stochastic transport noise, J. Stat. Phys. (2020) 1–36.
- [16] Colin Cotter, Dan Crisan, Darryl D. Holm, Wei Pan, Igor Shevchenko, A particle filter for stochastic advection by Lie transport: a case study for the damped and forced incompressible two-dimensional Euler equation, SIAM/ASA J. Uncertain. Quantificat. 8 (4) (2020) 1446–1492.
- [17] Dan Crisan, Franco Flandoli, Darryl D. Holm, Solution properties of a 3D stochastic Euler fluid equation, J. Nonlinear Sci. 29 (3) (2019) 813–870.

- [18] Dan Crisan, Darryl D. Holm, James-Michael Leahy, Torstein Nilssen, Variational principles for fluid dynamics on rough paths, Adv. Math. 404 (2022) 108409.
- [19] Michele Coghi, Torstein Nilssen, Rough nonlocal diffusions, Stoch. Process. Appl. 141 (2021) 1–56.
- [20] Peter Constantin, Analysis of Hydrodynamic Models, SIAM, 2017.
- [21] Alexander M. Davie, Differential equations driven by rough paths: an approach via discrete approximation, Appl. Math. Res. Express 2008 (2008).
- [22] Joscha Diehl, Peter K. Friz, Wilhelm Stannat, Stochastic partial differential equations: a rough paths view on weak solutions via Feynman–Kac, Ann. Fac. Sci. Toulouse Math. (6) 26 (4) (2017) 911–947.
- [23] Aurélien Deya, Massimiliano Gubinelli, Martina Hofmanová, Samy Tindel, A priori estimates for rough PDEs with application to rough conservation laws, J. Funct. Anal. 276 (12) (2019) 3577–3645.
- [24] Theodore D. Drivas, Darryl D. Holm, Circulation and energy theorem preserving stochastic fluids, Proc. R. Soc. Edinb., Sect. A, Math. 150 (6) (2020) 2776–2814.
- [25] Theodore D. Drivas, Darryl D. Holm, James-Michael Leahy, Lagrangian averaged stochastic advection by Lie transport for fluids, J. Stat. Phys. 179 (5) (2020) 1304–1342.
- [26] Aythami Bethencourt de Leon, Darryl D. Holm, Erwin Luesink, So Takao, Implications of Kunita– Ito–Wentzell formula for k-forms in stochastic fluid dynamics, J. Nonlinear Sci. (2020) 1–34.
- [27] Jonathan Demaeyer, Stéphane Vannitsem, Stochastic parameterization of subgrid-scale processes: a review of recent physically based approaches, in: Advances in Nonlinear Geosciences, 2018, pp. 55–85.
- [28] David G. Ebin, Jerrold Marsden, Groups of diffeomorphisms and the motion of an incompressible fluid, Ann. Math. (1970) 102–163.
- [29] Peter K. Friz, Martin Hairer, A Course on Rough Paths, Springer, 2020.
- [30] Franco Flandoli, Martina Hofmanová, Dejun Luo, Torstein Nilssen, Global well-posedness of the 3D Navier–Stokes equations perturbed by a deterministic vector field, arXiv preprint, arXiv:2004.07528, 2020.
- [31] P.K. Friz, N.B. Victoir, Multidimensional Stochastic Processes as Rough Paths: Theory and Applications, vol. 120, Cambridge University Press, 2010.
- [32] François Gay-Balmaz, Darryl D. Holm, Stochastic geometric models with non-stationary spatial correlations in Lagrangian fluid flows, J. Nonlinear Sci. 28 (3) (2018) 873–904.
- [33] Emanuela Gussetti, Antoine Hocquet, A pathwise stochastic Landau-Lifshitz-Gilbert equation with application to large deviations, arXiv preprint, arXiv:2103.00926, 2021.
- [34] Andris Gerasimovičs, Antoine Hocquet, Torstein Nilssen, Non-autonomous rough semilinear PDEs and the multiplicative sewing lemma, J. Funct. Anal. 281 (10) (2021) 109200.
- [35] Nathan Glatt-Holtz, Vladimír Šverák, Vlad Vicol, On inviscid limits for the stochastic Navier–Stokes equations and related models, Arch. Ration. Mech. Anal. 217 (2) (2015) 619–649.
- [36] Nathan E. Glatt-Holtz, Vlad C. Vicol, Local and global existence of smooth solutions for the stochastic Euler equations with multiplicative noise, Ann. Probab. 42 (1) (2014) 80–145.
- [37] Massimo Germano, Ugo Piomelli, Parviz Moin, William H. Cabot, A dynamic subgrid-scale eddy viscosity model, Phys. Fluids A, Fluid Dyn. 3 (7) (1991) 1760–1765.
- [38] István Gyöngy, Pablo Raúl Stinga, Rate of convergence of Wong–Zakai approximations for stochastic partial differential equations, in: Seminar on Stochastic Analysis, Random Fields and Applications VII, Springer, 2013, pp. 95–130.
- [39] Antoine Hocquet, Martina Hofmanová, An energy method for rough partial differential equations, J. Differ. Equ. 265 (4) (2018) 1407–1466.
- [40] Martina Hofmanová, James-Michael Leahy, Torstein Nilssen, On the Navier–Stokes equation perturbed by rough transport noise, J. Evol. Equ. 19 (1) (2019) 203–247.
- [41] Martina Hofmanová, James-Michael Leahy, Torstein Nilssen, On a rough perturbation of the Navier– Stokes system and its vorticity formulation, Ann. Appl. Probab. 31 (2) (2021) 736–777.
- [42] Darryl D. Holm, Jerrold E. Marsden, Tudor S. Ratiu, The Euler–Poincaré equations and semidirect products with applications to continuum theories, Adv. Math. 137 (1) (1998) 1–81.
- [43] A. Hocquet, T. Nilssen, An Itô formula for rough partial differential equations. Application to the maximum principle, Potential Anal. 54 (2021) 331–386.
- [44] Antoine Hocquet, Torstein Nilssen, Wilhelm Stannat, Generalized Burgers equation with rough transport noise, Stoch. Process. Appl. 130 (4) (2020) 2159–2184.
- [45] Antoine Hocquet, Quasilinear rough partial differential equations with transport noise, J. Differ. Equ. 276 (2021) 43–95.
- [46] Darryl D. Holm, Variational principles for stochastic fluid dynamics, Proc. R. Soc. A, Math. Phys. Eng. Sci. 471 (2176) (2015) 20140963.

- [47] Peter Imkeller, Jin-Song Von Storch, Stochastic Climate Models, vol. 49, Birkhäuser, 2012.
- [48] Oana Lang, Dan Crisan, Well-posedness for a stochastic 2d Euler equation with transport noise, Stoch. Partial Differ. Equ., Anal. Computat. (2022) 1–48.
- [49] T.J. Lyons, M. Caruana, T. Lévy, Differential equations driven by rough paths, in: Lectures from the 34th Summer School on Probability Theory Held in Saint-Flour, July 6–24, 2004, in: Lecture Notes in Mathematics, vol. 1908, Springer, Berlin, 2007, With an introduction concerning the Summer School by Jean Picard.
- [50] Paul J. Mason, Large-eddy simulation: a critical review of the technique, Q. J. R. Meteorol. Soc. 120 (515) (1994) 1–26.
- [51] Etienne Mémin, Fluid flow dynamics under location uncertainty, Geophys. Astrophys. Fluid Dyn. 108 (2) (2014) 119–146.
- [52] Remigijus Mikulevicius, B. Rozovskii, On equations of stochastic fluid mechanics, in: Stochastics in Finite and Infinite Dimensions, Springer, 2001, pp. 285–302.
- [53] Paul J. Mason, David J. Thomson, Stochastic backscatter in large-eddy simulations of boundary layers, J. Fluid Mech. 242 (1992) 51–78.
- [54] Andrew J. Majda, Ilya Timofeyev, Eric Vanden Eijnden, A mathematical framework for stochastic climate models, Commun. Pure Appl. Math. 54 (8) (2001) 891–974.
- [55] R. Mikulevicius, G. Valiukevicius, On stochastic Euler equation in ℝ<sup>d</sup>, Electron. J. Probab. 5 (6) (2000) 20.
- [56] Ugo Piomelli, Jeffrey Robert Chasnov, Large-eddy simulations: theory and applications, in: Turbulence and Transition Modelling, Springer, 1996, pp. 269–336.
- [57] Ugo Piomelli, William H. Cabot, Parviz Moin, Sangsan Lee, Subgrid-scale backscatter in turbulent and transitional flows, Phys. Fluids A, Fluid Dyn. 3 (7) (1991) 1766–1771.
- [58] Ugo Piomelli, Large-eddy simulation: achievements and challenges, Prog. Aerosp. Sci. 35 (4) (1999) 335–362.
- [59] Stephen B. Pope, Turbulent Flows, IOP Publishing, 2001.
- [60] Valentin Resseguier, Long Li, Gabriel Jouan, Pierre Dérian, Etienne Mémin, B. Chapron, New trends in ensemble forecast strategy: uncertainty quantification for coarse-grid computational fluid dynamics, Arch. Comput. Methods Eng. (2020) 1–47.
- [61] Valentin Resseguier, Wei Pan, Baylor Fox-Kemper, Data-driven versus self-similar parameterizations for stochastic advection by Lie transport and location uncertainty, Nonlinear Process. Geophys. 27 (2) (2020) 209–234.
- [62] Oliver D. Street, Dan Crisan, Semi-martingale driven variational principles, Proc. R. Soc. A, Math. Phys. Eng. Sci. 477 (2247) (2021) 20200957.
- [63] Ulrich Schumann, Stochastic backscatter of turbulence energy and scalar variance by random subgrid-scale fluxes, Proc. R. Soc. Lond. Ser. A, Math. Phys. Sci. 451 (1941) (1995) 293–318.
- [64] Glenn Shutts, A kinetic energy backscatter algorithm for use in ensemble prediction systems, Q. J. R. Meteorol. Soc. 131 (612) (2005) 3079–3102.
- [65] William C. Skamarock, Joseph B. Klemp, Jimy Dudhia, David O. Gill, Zhiquan Liu, Judith Berner, Wei Wang, Jordan G. Powers, Michael G. Duda, Dale M. Barker, et al., A description of the advanced research WRF model version 4, National Center for Atmospheric Research, Boulder, CO, USA, 2019, p. 145.
- [66] Joseph Smagorinsky, General circulation experiments with the primitive equations: I. The basic experiment, Mon. Weather Rev. 91 (3) (1963) 99–164.
- [67] Chenyue Xie, Jianchun Wang, Weinan E, Modeling subgrid-scale forces by spatial artificial neural networks in large eddy simulation of turbulence, Phys. Rev. Fluids 5 (May 2020) 054606.